

Form I.R.

निशमन का प्रमाण-पत्र

	<b>A</b>
	CERTIFICATE OF INCORPORATION
	ता. की सं. 1999 No. <u>11-120599</u> of Date
	में स्तद्द्वारा प्रमाणित करता हूं कि आज
कम्पनी क	द्वितिषय (mse का सं. 1) के अधीन निगमित की गई है और कम्पनी परिसीमित है। certify that
i nereo	LIMITED
is this	day incorporated under the Companies Act, 1956 (No. 1 of 1956) and that the ny is limited.
South	भेरे प्रस्तानार से आज ता.  BENNAT  Given under my hand at this NEWSTYNUM
	day ofOne thousand nine hundred and
	( V. C. DAVEY)

## Certificate of Incorporation Consequent upon conversion to Public Limited Company



Registrar of companies, Mumbai Everest, 100 Marine Drive, Mumbai, Maharashtra, India, 400002

Corporate Identity Number: U45209MH1999PLC120599

Fresh Certificate of Incorporation Consequent upon Conversion from Private Company to Public Company

IN THE MATTER OF CONCORD ENVIRO SYSTEMS PRIVATE LIMITED

I hereby certify that CONCORD ENVIRO SYSTEMS PRIVATE LIMITED which was originally incorporated on First day of July One thousand nine hundred ninety-nine under the Companies Act, 1956 as CONCORD ENVIRO SYSTEMS PRIVATE LIMITED and upon an intimation made for conversion into Public Limited Company under Section 18 of the Companies Act, 2013; and approval of Central Government signified in writing having been accorded thereto by the RoC - Mumbai vide SRN F05070388 dated 09.06.2022 the name of the said company is this day changed to CONCORD ENVIRO SYSTEMS LIMITED.

Given under my hand at Mumbai this Ninth day of June Two thousand twenty-two.

DS MINISTRY OF THE APPROVACE OF THE APPR

ALPESH D MANIYA

Registrar of Companies

RoC - Mumbai

Mailing Address as per record available in Registrar of Companies office:

CONCORD ENVIRO SYSTEMS LIMITED

101, HDIL TOWERS,, ANANT KANEKAR MARG, BANDRA(EAST),,
MUMBAI, Mumbai City, Maharashtra, India, 400051



## MEMORANDUM OF ASSOCIATION

OF

## **CONCORD ENVIRO SYSTEMS LIMITED**

- The name of the Company is CONCORD ENVIRO SYSTEMS LIMITED.\*
- II. The Registered Office of the Company will be situated in the State of Maharashtra.
- III. The Objects for which the Company is established as the following:-
- (A) THE MAIN OBJECTS OF THE COMPANY TO BE PURSUED BY THE COMPANY ON ITS INCORPORATION ARE:-
- To takeover running business of "Universal Blenders," a registered partnership firm with its all assets, all liabilities, trade marks, goodwill, licences, permits etc. and after complete takeover, the pannership firm would be dissolved.
- 2. To deal in, import, export, sell, manufacture, fabricate, design, assemble, construct, erect, commission, hire all types of Pollution Control Systems, Environmental Control Systems, Water Treatments System alongwith their plants, machineries, accessories, Instruments incorporating or by themselves all associated equipment for Pretreatment such as precipitators, lime softeners, pressure tanks and filters, ion exchange softeners, ion exchange demineralisers, dealkalisers, metal recovery systems, condensate polishers, waste neutralization systems, electro- dialysers, diffusion dialysers, reverse osmosis plants, degassers and detractors or any similar equipments alongwith its spare parts, appliances, fixtures, attachments, implements, devices, materials, substances, minerals things and other requisites and facilities necessary or useful.

CONCORD ENVIRO SYSTEMS LIMITED

NVIRO

MONCORD ENVIRO SYSTEMS PVT. LTD.

PURECTOR

Clause I amended by special resolution passed at the Extra-ordinary General Meeting held on 25<sup>th</sup> May 2022

SING 아닌티티

#### (B) OBJECTS INCIDENTAL OR ANCILLARY TO THE ATTAIMENT OF THE MAIN OBJECTS:

- 3. To enter into Collaboration, Joint Venture, technology transfer or any similar agreements with any Company / firm or an individual whether Indian or Foreign having appropriate know-how / Technology in the pollution control/ environment and water treatment fields.
- 4. To manufacture, design, assemble, fabricate, service, construct, erect, commission, act as dealers, importers, and exporters in all kinds of pollution/ environment control and water treatment equipment alongwith all its instrumentation, controllers etc.
- 5. To deal in, import, export, market, process and manufacture compounds, chemicals, derivatives, by - products required for pollution/ environment control and water treatment systems.
- 6. To train and recruit personnel for Pollution/ Environment Control and Water Treatment process, controlling effluents, pollutants and operate such equipment.
- 7. To apply for tender, purchase or otherwise acquire any contracts, subcontracts, licences and concessions in connection with the main business and to undertake, execute carry out, dispose of or otherwise turn to account the same.
- 8. To buy land and erect and construct factory, service station, training centre, Spares Centre to enhance company's main objective in environment, pollution control and Water Treatment business and to takeover any concern, company, proprietary concern carrying on similar business or be takeover by any concern, company or partnership firm carrying on similar business.
- 9. To apply for, purchase acquire any patents, brevets d'inventionns, license, concessions and the like, conferring any exclusive or non- exclusive or limited right to use any secret information as to any invention which may seem capable of being use for any of the purpose of the company and to use,

CONCORD ENVIRO SYSTEMS PVT. LTD.

SONCORD ENVIRO SYSTEMS LIMITED Authorised Signatory exercise, develop or grant licenses in respect of or otherwise turn to account the property rights of information so acquired.

- 10. To borrow, to raise money or receive money in deposit either with or without security at interest or otherwise in such manner as the Company may think fit and in particular by the issue of debentures or debenture stock, perpetual or otherwise including debentures or debenture stock convertible into shares of this Company or perpetual annuities and in securities of such money so borrowed, raised, or received to mortgage, pledge or charge the whole or any part of property, asset or revenue of the Company present or future by special assignment or otherwise to transfer or convey the same absolutely or in trust and give the lender power of sale or other powers as may seem expedient and to purchase, redeem or pay off any such securities provided the Company shall not carry on the business of banking as defined in the Banking Regulation Act, 1949, subject to the provisions of section 58-A and directives of R.B.I.
- 11. To open bank accounts, borrow from banks or financial institutions, to draw, accept and make and to endorse, discount and negotiate promissory notes, hundies, bills of exchange, bills of lading and other negotiable or transferable instruments, in connection with the Company's business.
- 12. To Purchase land and to construct, maintain, alter, take on lease or in exchange hire, let otherwise acquire or sell any immovable and movable property for the purpose of the Company and resale, alienate, assign or relinquish any property.
- 13. To enter into partnership or into any arrangement for arrangement for sharing profit, union of interest, co-operation, joint venture or collaboration (Local, Foreign or NRI). reciprocal concession or otherwise with any person or partnership firm or association or company carrying or about to carry on or engage in any business or transaction, which the Company is authorized to carry on or to amalgamate with another company or companies having similar business, objects, to appoint trustees, recruit general and technical staff.
- 14. To lend money, guarantee contracts of, or otherwise acquire shares or securities of any such company and to sell, hold, reissue, with or without

CONCORD ENVIRO SYSTEMS LIMITED

CONCORD ENVIRO SYSTEMS PVT

RECTOR

guarantee or otherwise deal with the same, provided the Company shall not carry on the business of banking as defined in the Banking Regulation Act, 1949, and also to invest Company's surplus funds in any shares, securities, properties (movable or immovable) or as decided by the Board of Directors.

- 15. To establish, provide, maintain and conduct or otherwise subsidies research laboratories and experimental workshops for research and experiment and to undertake and carry on experiments and tests of all kinds and to promote studies, research and invention by providing, subsidinsing, endowing or assisting, workshops, libraries and conducting lectures and conference and providing the remunerations to professors or teachers award of scholarships, prizes, grants and bursaries to students or otherwise and generally to encourage, promote and reward studies, researches, investigation, experiments, tests and invention of any kind that may be considered likely to assist in any of the business which the Company is authorized to carry on, also to provide or construct houses for staff, directors and their dependents.
- 16. To grant bonuses, allowances, gratuities, pension to and provide for the welfare of employees and ex-employees, directors and ex-directors of the Company or the dependents of such persons and to support or to subscribe to any charitable institutions, associations, to contribute to any fund including religious, national, international, social and political and also for the staff and directors, contribute to P.P. superannuation and welfare funds.
- 17. To create depreciation fund, reserve fund, insurance fund or any special or any special or any other fund, whether for depreciation or repairing, improving, extending or maintaining any of the property or welfare of the Company or for redemption of debentures and redeemable preference shares or for any other purposes whatsoever conducive to the interest of the Company.
- 18. To transfer accumulated profit or premium received on sale of shares or debentures and the profit or money realized from the sale of forfeited shares to reserves and distribute them as fully or partly paid up bonus shares among the shareholders.

ACMOORD ENVIRO SYSTEMS PVT. LTD.

CONCORD ENVIRO SYSTEMS LIMITED

Authorised Signatory

DIRECTOR

- 19. To accept donations, gifts and to deposit, present or otherwise dispose of either voluntarily or for value property of the Company deemed to be national, public or local interest to any national trust, charitable institution, body, museum, corporation or authority any trustees for or on behalf of any above bodies.
- 20. To distribute any of the property of the Company amongst the members in specie or in kind, subject to the provisions of the Act, in the event of winding-up.
- 21. To open a branch in India or abroad or to promote or procure the Company authorized to do business as a company with limited liability in any trade. To pay all costs, charges and expenses of the promotion and establishment of the Company.
- 22. To acquire, merge, amalgamate and undertake the whole or any part of business, property and liability of any firm or persons or company carrying on any business which the Company is authorised to carry on or being taken over or become subsidiary of other company.
- 23. TO adopt such means of known the business of the Company as may seem expedient and in particular by advertising in the press, by circulars, by post or otherwise and hold exhibition of art or interest, by the publication of books and periodicals and by granting prizes, rewards and donations in connection with the Company's Business.
- 24. To pay for any property or right acquired by the Company either in cash or in fully or partly paid —up shares or by securities which the Company has power to issue or pailly in one mode and partly in another and generally on such terms as the Company may determine and further to sell, issue, share, debentures, bonds or any security of the Company to NRIS> foreign nationals, if so resolved.
- 25. To sell or dispose of, exchange, mortgage, transfer of any property or undertaking or easements of the Company or any part thereof in such manner and for such consideration as the Company may think fit.

CONCORD ENVIRO SYSTEMS LIMITED

ENVIRO

CONCORD ENVIRO SYSTEMS PVT. LTD.

DIRECTOR

26. To do the above things and all such other things as are main, incidental or other objects or may be thought conducive to the attainment of the above object or any of them and to do them as principals, agents, contractors, trustees, guarantors or otherwise and by or through trustees, agents or otherwise and either alone or in conjunction with others.

## (C) OTHER OBJECTS

- 27. To carry business as exporters, importers, dealers, manufactures in all types of desalination plants and products, electrical good, software including computers, videos, T.V.s satellite, wireless good (including relaying and broadcasting services )photographic articles, gramophones products, tape recorders, music instruments, maritime instruments, teleprinters, typewriters, calculators, thermometers, medical/ surgical equipments, weighing, measuring and precision instruments and communication including telecommunication goods and allied services.
- 28. To carry on business as marine, civil, mechanical, electrical, chemical and consulting engineers; surveyors, architect machinists, fabricators, erectors, millwrights, electroplates, metallurgists, foundry experts and galvanizers.
- 29. To carry on business of warehousing and developing, properties, flats, shopping centers, industrial estates, business units, theme parks, gyms, plantations, acquafarms, civil works, water works and to sell, lease, mortgage, donate, gift or dispose them of in any manner the Company may deem fit.
- 30. To carry on business as manufacturers, exporters, dealers, commission agents in automobile parts, cars, vehicles, furniture, stationery, all types of goods and items made of fabric, plastic, nylon, bakelite, yarn, leather, paper, steel, foam, bamboo, glass, rubber, chemicals, stone gold and silver.
- 31. To carry on business as manufactures, exporters and dealers in textiles, garments, hosiery, silk, jute, cotton, leather goods, perfumes, cosmetics, detergents, toiletries imitation jewellery, utility product and general merchandise and job equipment.
- 32. To print, import, export and deal in technical, educational, health and scientific books and publication, running of printing presses, libraries of books POWCOND INVINO IS THEMS PARE LIVE

CONCORD ENVIRO SYSTEMS LIMITED

HOYDERE

ANNOOR ENVIRO SYSTEMS PVT.

publication, journals and film cassettes and to set up training schools, technical institutions and colleges.

- 33. To print, import, export and deal in technical, educational, health and scientific books and publications, running of printing presses, libraries of books, publications, journals and firm cassettes and to set up training schools, technical institutions and colleges.
- 34. To deal in and carry on travel, transport, tour business (road, rail, water and air), ply vehicles, aircraft, to own, deal in, manage, maintain, repair, make, operate ships, barges, launches for the passengers and cargo, vehicles, two wheelers, three wheelers, tractors and their accessories.
- 35. To carry on business of hot leering, catering, lodging, boarding, housekeeping flight kitchen and mobile kitchen, money changes, foreign exchange dealers and any other business permitted by RBI.
- 36. To promote, sponsor, undertake and carry out rural development, farming, irrigation assists any activity of groMh of national economy water refining, air pollution control and produce such equipment, and promote academic and research pursuits, to recruit candidates for the employment, impart training in rural area with reference in agricultural and allied subjects.
- 37. To process, produce, cure, can, cut, size, pack, export and deal in rural products, agricultural produce, food, vegetables, oils, aqua products (including fish and pearls)cattle feed and soft-drinks, beverages, spirits, poultry products, bakery and confectionery products, marine produce, liquors, sugars, salts, provisions, fei1ilizers, fertilizers, fruits, flowers, plantations, timbers, plant biotechnology, forest products, pesticides and acquire, construct and operate connected refineries, mills and units.
- 38. To carry on business of publicity, advertisement, exhibition, film distribution, film production and any activity or business allied with film, to import, export and to deal in pictures, video cassettes and taps; financing of film production, distribution, hiring or subletting theaters, studies for drams, art shows and film exhibition photograph and supply of cinematographic, theatrical and

CONCORD ENVIRO SYSTEMS LIMITED

Authorised Signatory

CONCORD ENVIRO SYSTEMS PVT. LTD.

DIRECTOR

photographic material and service to act as agents for training, retaining, supplying artistes, directors, story writers and technicians.

- 39. To carry on business of interior decoration, architecture, designing, planning, sculpture, modern art, dealers in and suppliers/ manufactures, prospectors of construcțion materials including cements, tiles, bricks, lime, charcoal, marble and granite, decoration articles, mining products and colours.
- 40. To buy, sell, offer franchise, promote technical know- how, commercial data, managerial information and collaboration in chemicals, dyes, pharmaceutical, medicines, agriculture, engineering, power, architecture, construction, electronics, luggage, footwear, cinematography, water treatment desalination, electroplating, industrial development, foam, shipping, cables, wires, oil, textile, furniture, leather, transport, education, space science, dairy products and to deal in or to conduct research and development in and deal in all these or deal in any of these products or services.
- 41. To mobilize savings, carry on business as on investment company or leasing company and/ to undertake and to transact all kinds of trusts, agency and leasing business, to carry on business as financiers and underwriters, dealers, lenders, hire purchasers of all types of properties, goods and machinery.
- 42. To carry on business to deal in shares, stock, bonds or debentures securities (of any Government or public authority of Company) or bullion and to form, promote, subside and assist companies, partnership firms.
- 43. To carry on business of importing, exporting, manufacturing and dealing in jewellary, precious/semi precious stones, goldsmith's services, hardware, omnibuses, industrial and agricultural implements.
- 44. To carry on business of manufacturers, importers, exporters and developers of all kinds inorganic and organic and chemicals, acids, alklaies, solvents, waxes, lacquers, resins, varnishes, drugs, pharmaceutical, paints pigments, oil fuel, coaltar, dyes, foot wear and bye- products thereof.

CONCORD ENVIRO SYSTEMS LIMITED

WITH THE REAL

CONCORD ENVIRO SYSTEMS PVT. LTD.

- 45. To carry on business of imponers, expons and developers of security papers' cards printing apparatus and render courier service and disseminate repon, infofmation plans and statistics.
- 46. To produce professional service on export and international trade, indenting, finance, management, investment, marketing, surveying and secretarialservices.
- 47.To carry on business as producers, manufacturers processors, conveilers, exponers, leaders, retailers, agenls, buyers or sellers of all types of gases, fuels, fluids required fOf Or used in industries, agriculture, clinics, hospital, refrigeration, aviation, transport vehicles, space rockets and crafts, communication objects and media power plants domestic or public lighting, cooling or cooking purposes, lighters, plants producing water, chemicals or fuels, pesticides, defence or warfare establishment holliculture, forest or plant pfotection and growth and other allied purposes and to and to service, repair, manufacture, market or deal in, machinery, plants, spares, cylinders, containers, gadgets, application and accessories required for working on, using or producing any of such gases and products and lab equipment.
- 48. To carry on business of dealers in and manufacturers of machinery, tools, dies, fixtures, packing materials, trading equipment, mining equipment, carlons, laps, rubbers, gums labels and security/ defence products and general merchandise.
- IV. The liability of the members is limited.
- V. \*The Authorised share capital of the Company is Rs. 42,50,00,000/- (Rupees Forty Two Crores Fifty Lakhs only) comprising of 4,00,00,000 (Four Crores) equity shares of Rs. 5/- (Rupees Five) each ranking pari passu with the existing equity shares in all respects and 2,25,000 (Two lakh twenty five thousand) preference shares of Rs.1000/- (Rupees One Thousand) each, with the power to the Board to decide on the extent of variation in such rights and to classify and re-classify, from time to time, such shares into any class of shares."

\*Clause V amended by special resolution passed at the Extra-ordinary general meeting held on 25th May 2022

CONCORD ENVIRO SYSTEMS LIMITED

Authorised Signatory

CONCORD ENVIRO SYSTEMS PVT. LTD

DIRECTOR

We the several persons, whose names and addresses are subscribed below are desirous of being formed into a company in pursuance to this Memorandum of Association and We respectively agree to take the number of shares in the capital of the Company set opposite our respective names

Signature, Names, Addresses,	Number of Shares	Signatures, Names,
Occupations and Description	taken by each	Addresses, Occupation
of Subscribers	Subscriber	and Description of Witness
1. Bimal Kumar Goel	10 (Ten) Equity	
S/o Late Shri Lachman	Shares of	
Dass Goel 303-B, Vandana,	Rs. 100/- each	
Mhada Complex, Versova,		
Andheri (W)	Δ.	
Mumbai- 400 058		543
Business		
S/d-		
	10 (Ten)	
2. Kamlesh Kumar Goel	Equity Shares of	Jawahar Bathjia
S/o Late Sh. Lachman Dass	Rs. 100/- each	S/o Thakurdas
Goel		24 Veena Beena
302, Dheeraj Dhan		Shopping Centre,
St. Alexius Road,		Bandra, Mumbai- 50
Bandra (W),		
Mumbai- 400 050		Company Secretary
Business		
S/d-		S/d-
Total	20 share	

MUMBAI, Dated 23rd June 1999.

CONCORD ENVIRO SYSTEMS PVT. LTD.

CONCORD ENVIRO SYSTEMS LIMITED



# CONCORD SHIPPING (PVT.) LTD. 101, HDIL Towers, First Floor, Anant Kanekar Marg, Bandra (East), Mumbal - 400 051.

: +91 22 6704 9000 Fax : +91 22 6704 9010

E-mail: concord@rochemindia.com

June 23,2022

1

The Municipal Commissioner, Municipal Corporation of Greater Mumbai Mumbai

Sub: NOC for Shop & Establishment Registration

Dear Sir,

This is to confirm that we are the lawful owner of the office premises situated at 101, HDIL Towers, Anant Kanekar Marg, Bandra (East) Mumbai-400051. We further state that we have no objection to M/s. Concord Enviro Systems Ltd for apply to you for registration of their office under Shop & Establishment Act in our premises

This letter has been issued on request of M/s.Concord Enviro Systems Ltd

Thanking you,

Yours faithfully,

For Concord Shipping Pvt Ltd.

Director

CONCORD ENVIRO SYSTEMS LIMITED



महाराष्ट्र MAHARASHTRA

① 2021 ①

YW 935679

प्रधान मुद्रांक कार्यालय, मुंबई प.मु.वि.क. ८०००९ १ - 3 SEP 2021 सक्षम अधिकारी

श्री. कि क. गवर्ड

# This stamp paper found an integral pand of the stame & licence argument enembed on & all day of deptember 1021 at number between & Consend Shipping put ud and concered environ & Justice put ud.

CONCORD ENVIRO SYSTEMS LIMITED



#### LEAVE & LICENCE AGREEMENT

This Leave & License Agreement (the "Agreement") made at Mumbai on 22<sup>nd</sup> September, 2021 between

CONCORD SHIPPING PRIVATE LIMITED a Company incorporated under the Companies Act 1956 and having its Registered Office at 101, HDIL Towers Anant Kanekar Marg Bandra (East) Mumbai- hereinafter referred as LICENSOR (which expression shall unless repugnant to the context or meaning thereof mean and include their successors and permitted assigns) of the FIRST PART

#### AND

CONCORD ENVIRO SYSTEMS PRIVATE LIMITED a Company incorporated under the Companies Act 1956 and having its Registered Office at 101, HDIL Towers Anant Kanekar Marg Bandra (East) Mumbai- hereinafter referred as LICENSEE (which expression shall unless repugnant to the context or meaning thereof mean and include their successors and permitted assigns) of the SECOND PART

The Licensor and Licensee are hereinafter individually referred to as "Part" and collectively referred to as the "Parties".

- A Whereas, the Licensor is the absolute owner of the Office Premises situated at 101 & 103 HDIL Towers, Anant Kanekar, Bandra (East), Mumbai- 400051 and measuring 810.12 Sq. Mts (Carpet Area) more particularly described in the Schedule herein under written (hereinafter referred to as "Licensed Premises") intended to be Licensed hereby and has a clean and unrestricted right and title thereto and is legally permitted and competent to enter into this License on the terms and conditions contained hereto.

  Village Bandra (East) CTS No. 608 (Part)
- B. the Licensee is a Private Limited company engaged inter alia in the business of Manufacturing of Water Purification/ Filtration and desalination Plants (hereinafter to be referred as the "said Business")
- C. the Licensor represents that he had purchased the said office premises vide an agreement dated 26<sup>th</sup> March 2009 and 14<sup>th</sup> December, 2011 (hereinafter to be referred as "said Agreement for Sale").
- D. the Licensee has represented to the Licensor.
- i)The use of the Licensed premises to be allowed hereinafter stated will not tantamount to a tenancy and will not create any tenancy or other similar right or interest and the Licensee undertakes to assures not to make any such claims in any proceeding or otherwise
- ii) At no point of time Licensee or any one on behalf of the Licensee contend that this agreement confers any right, title or any nature or other similar right or interest to the said licensed premises or any part thereof.
- iii) The Licensee shall not claim protection of the Maharashtra Rent Control Act 1999 (Mah.18 of 2000) or any modification or re-enactment thereof or any law giving any protection to a tenant or a Licensee and that notwithstanding any change in law, the rights and liabilities of the parties set out herein will remain unaffected.
- iv) the parties hereto are desirous of recording the agreed terms and conditions in form of agreement on which the Leave and License has been agreed to be granted

CONCORD ENVIRO SYSTEMS LIMITED

Authorised Signal

# NOW THEREFORE THIS LICENSE AGREEMENT WITNESSETH AND IT IS HEREBY AGRRED BY AND BETWEEN THE PARTIES AS FOLLOWS:

The recitals hereinabove contained shall constitute an integral part of this agreement.

#### ARTICLE 1

- 1. The Licensor hereby grants to the Licensee and the Licensee hereby accepts a license of the part of said office premises measuring total 810.12 Sq. Mts Carpet Area (hereinafter referred to as said part of office/ Licensed Premises) more particularly described in the schedule herein under for running the business activities, for a period of 11 months commencing from 22.09.2021 and expiring by the efflux of time on 21.08.2022 (both days inclusive) (hereinafter to be referred as "Term").
- 2. The date of commencement of this License will be from 22.09.2021.
- 3. The Licensor and Licensee hereby agree to extend this agreement mutually after the expiry of this term and the same should be executed and signed by both the parties.

#### **ARTICLE 2**

#### LICENSE FEE AND TAXES

- 3.1 The Licensee shall keep with the Licensor an amount of Rs.50,000/- (Rupees fifty thousand only) (herein referred to as the Security Deposit) as a refundable Security Deposit without interest for due performance and observance of the original Leave & license Agreement. The said Additional Security Deposit is refundable by the Licensor simultaneously against the return of the vacant possession of the said premises on expiry of this Agreement or earlier on termination of this Agreement.
- 3.2 The Licensee agrees to pay rent of Rs.5000/- (Rupees five thousand only) per month.
- 3.3 The licensor shall also be liable to pay Property Tax and any increase in Property Tax levied by any authority payable in respect of the said part of office premises.
- 3.4 The Society & Non-Occupancy charges, if any, in respect of the licensed Premises shall be borne by the Licensee for the remaining term and under no circumstances shall the Licensor be held liable for the same. The Licensee indemnifies for any loss or cost incurred or claims made on the licensor on the Licensee's failure to do so.

## ARTICLE 3

## **PURPOSE**

3.1 The Licensee shall use the said part of the Office Premises solely for the purpose of carrying on its business activities.

#### **ARTICLE 4**

## LICENSOR'S AND LICENSEE'S COVENANTS

- 4.1 In addition to the payment of the Licensee fee, the Licensee shall also be liable to pay during the Term, the Electricity charges for the electricity consumed in accordance with the electricity bills received from electricity Supply Company. The Licensee will also pay the water charges as per water bill received from B.M.C.
- 4.2 The Licensee hereby covenants with the Licensor that it will obtain all necessary approval/ licenses and sanctions from the concerned authorities for carrying on its Business and comply with all the conditions of such licenses/ approvals/ sanctions and take appropriate

ONCO STORY ON TO STORY ON THE S

CONCORD ENVIRO SYSTEMS LIMITED

Insurance policy and third party insurance at its own for its fixtures, goods and articles lying in the said office premises.

- 4.3 The licensee shall keep, maintain and use the said office premises in good order and condition.
- 4.4 The Licensee shall permit the Licensor and/or their authorized representatives to enter upon the said office premises with prior written notice to the Licensee for the purpose of inspecting the service and maintenance of the said premises and to carry out repairs at reasonable time as and when necessary.
- 4.5 The Licensee shall not carry on any business or activity which could be construed as illegal, defamatory, immoral or obscene or might contravene any law, order, by law, rules or regulations of any government, central or state or Municipal or otherwise or which might constitute any nuisance or annoyance to the Licensor.
- 4.6 It is agreed by and between the parties hereto that the Licensors have merely given license to the Licensee only to use and occupy the said Licensed Premises strictly on the terms and conditions of this agreement. The Licensor shall not be responsible for a statutory power or water cut/ electricity cut imposed by the authorities and any irregular supply of the same.
- 4.7 That the agreement does not create nor it is intended to create any tenancy or sub-tenancy and the Licensee shall not claim any right to tenancy and the Licensee's status under this agreement is and shall be that of mere "Licensee" under section 24 together with chapter VIII of the Maharashtra Rent Control Act, 1999 as amended from time to time. The Licensee will not sub-let, under-let, assign, give on paying guest basis, or otherwise part with the possession of the said Licensed Premises during the period of the agreement. However, sister concerns the Licensee shall have the liberty to permit its own affiliates, subsidiary/ies or Parent Companies to use the Licensed premises. Further Licensee has an option to renew the subject Leave & License Agreement on same terms & conditions for further period of five years. The licensee shall observe all laws, rules and regulations in relation to their employees/ staff etc. whether existing or which may be imposed hereafter by any authorities from time to time and the licensee shall observe the same strictly in accordance with law. If any accident or mishap takes place on the said licensed premises during the period of license, then the licensee shall be liable for all the consequences arising there from including the liability if any, for all the damages and compensation that may be payable to workers/ employees for the Licensee and the Licensee shall always indemnify ad keep indemnified, the licensors against all such liabilities that may arise on account of the same or any claims that maybe preferred by the workers/ employees or others connected with the licensee's activity.
- 4.8 The Licensor shall not be responsible or liable for any theft, loss, damage or destruction of any property of the Licensee or of any other persons lying in the said Licensed Premises or for any damage to the licensee or their members, agents, employees, servants, visitors, invitees and other persons entering the said Licensed Premises for the time being from any cause whatsoever.

#### **ARTICLE 5**

#### **INDEMITY**

- 5.1 The Licensor hereby agrees to indemnify and keep the licensee indemnified against any claim or loss or damage, which the licensee may sustain or suffer or costs charges and expenses the licensee may incur or for which the licensee may become or be held liable or responsible of any nature whatsoever arising out of any act, deed, matter or thing done or not done or committed or any negligence or default or breach of promise or contract or violation on the part of the Licensor or their representatives in the course of this agreement or renewal, if any, or otherwise if and only if attributable to the neglect of the licensor.
- 5.2 The Licensee hereby agrees to indemnify and keep the licensor indemnified against any claim or loss or damage, which the licensor may sustain or suffer or costs charges and expenses the licensor may incur for which the licensor may become or

CONCORD ENVIRO SYSTEMS LIMITED

be held liable or responsible. If any of the licensee's customer or anyone else including any public authorities should hold him responsible or liable for payment or any loss or damage or costs, charges or expenses or proceedings of any nature whatsoever arising out of any act, deed, matter, or thing done or not done or committed or any negligence or default or breach of promise or contract or violation on the part of the Licensee or their representatives in the course of selling their products or otherwise if and only if attributable to the neglect of the Licensee.

#### ARTICLE 6

## **TERMINATION & CONSEQUENSES**

- 3.1 It is hereby agreed by and between the parties that Licensee shall, be at a liberty to terminate the License at any time during the Term by serving upon the Licensor a written License Termination Notice of three months in advance.
  - 3.2 The Licensor during the Term of the license, shall not have any right to terminate the License save and except as provided in Article herein below.
  - 3.3 In case the Licensee fails to pay the rent as agreed between the parties then the Licensor shall serve the notice in writing of 30 days to the Licensee for payment of the due amounts. However, of the Licensee still fails to discharge the liability on the expiry of the notice, the Licensor at its discretion, may terminate the License.
  - 3.4 However in the event of either of the parties committing a breach of any of the terms of this agreement and failing, within 30 days, to remedy or make good such breach on receipt of notice in writing from other party, the party giving the notice shall be entitled to for with terminate this agreement without prejudice to any other rights or remedies which that party may have under this agreement or under law. In such event the consequences stipulated in Article hereinafter appearing, shall apply, provided however that in the event of any party committing the same breach more than once then the aforesaid notice period of thirty days for remedying the breach shall stand reduced to seven days.
  - 3.5 The Licensee shall not carry on any business or activity which could be construed as illegal, defamatory, immoral or obscene or might contravene any law, order, by law, rules or regulations of any government, central or state or Municipal or otherwise or which might constitute any nuisance or annoyance to the Licensor.
  - 3.6 Upon expiry of the term or upon earlier termination by the Licensee, during the term, for whatsoever reason:
    - a. The licensee shall remove or cause to removed itself, its agent and all its employees and all others and their respective belongings, chattels, articles and thing from the said office premises and shall hand over to the Licensor vacant quite peaceful and exclusive possession of the said office premises together with the fitting and fixture in good condition as on date provided however that such movable furniture, removable fixture, fitting and installation brought in the said office premises by the licensee and save and except the permanent fitting and fixture, flooring, ceilings etc. which the licensee has carried out, maybe removed by the Licensee from the said office premises without causing any damage to the said office premises or part thereof or any fittings or any part thereof.
    - b. On expiry of the Term or earlier termination of the agreement by the licensee where the licensee has offered to handover physical and vacant possession of the said office premises.

CONCORD ENVIRO SYSTEMS LIMITED

ARTICLE 7

#### NO LEASE, TENANCY ETC

- **7.1.** The use of the said office premises allowed as per this agreement will not tantamount to a tenancy and will not create any tenancy or similar right or interest and the Licensee undertakes and assures not to make any such claims in any proceedings or otherwise.
- 7.2. At no point of time will the Licensee or any one on behalf of the licensee contend that this agreement confers any right title or interest of any nature or other similar right or interest to the said office premises or any part thereof.
- 7.3. The Licensee is not entitled to nor will they claim protection under the Maharashtra Rent Control Act, 1999 (MAH 18 of 2000) or any statutory modification or re-enactment thereof of any law giving any protection to an occupant or a licensee.

#### ARTICLE 8

#### NOTICE

8.1. That any notice required to be served upon the parties shall be served at the address mentioned in the title of this agreement or at the substituted address as any of the parties has given to the others in writing and shall be sufficiently served upon if posted by Registered A/D post or through a recognized Courier Service Company.

#### **ARTICLE 9**

#### NON-WAIVER

9.1. No failure on the part of Licensor/ Licensee to exercise and no delay on the part of Licensor/ Licensee in exercising any right herein under, shall operate as waiver thereof, nor shall any single or partial exercise of any such right precluded any other or future exercise thereof, or the exercise of any other right. The remedies herein are cumulative and not exclusive of any remedies provided by law.

## ARTICLE 10

#### FORCE MAJEURE

- 1 It is hereby expressly agreed by and between the parties hereto that if at any time during the said Term the said Licensed Premises or any part thereof shall be destroyed, damaged by fire, tempest, food, earthquake, civil commotions, riots, terrorist attack, governmental action, war etc. or by any other force majeure cause or if the same shall become unfit for occupation and use for the purpose for which the said licensed Premises have been given the following consequences shall follow:
  - a. In the event if the said Licensed Premises are partly destroyed and can be repaired within a period of 30 days without the Licensee being required vacate the premises then the Licensee shall be bound to pay the proportionate license fees and the premises shall be restored by the licensor at his cost.
  - b. In the event if the said Licensed Premises cannot be restored for the period beyond 30 days then notwithstanding anything contained in this agreement, the licensee shall have an option to terminate this agreement forthwith without any notice period. However, in the event if the Licensee opts not to terminate the agreement and occupy the said Licensed Premises after the same are restored then the Licensee shall be liable to pay the Licensee Fee for the period required to repair the said Licensed Premises.
  - c. In the event if the Licensee exercise the option to occupy the said Licensed
    Premises after the same is restored the Licensee shall be liable to pay the
    Licensee Fee for the period required to restore the said Licensee Premise to its
    original state, irrespective of whether the Licensee could use the said Licensed
    Premises or not.

    CONCORD ENVIRO SYSTEMS LIMITED

#### ARTICLE 11

#### PARTIAL INVALIDITY

11.1. If at any time, any provision of this agreement shall become or be held illegal, invalid or unenforceable. In any respect under any law, then the legality, validity or enforceability of the remaining provision shall not in any way be thereby affected or impaired. Any invalid or unenforceable provision of this agreement shall be replaced with a provision which is valid and enforceable and most nearly reflects the original intent of the invalid or unenforceable provision.

#### **ARTICLE 12**

#### **SUPERSESSION**

12.1. This agreement constitutes the entire agreement between the Licensor and the Licensee and supersedes all prior understanding and writing between the parties.

#### **ARTICLE 13**

#### OPTION TO SELL

13.1. It is agreed between the parties that anytime during the term, in case the Licensor is desirous of selling the said office premises to any outside, the Licensor will ensure that all terms and conditions of the Leave and License shall be binding on intending purchaser.

#### **ARTICLE 14**

#### COMPLETE AGREEMENT

14.1. Both the parties to this agreement acknowledge that this agreement is the complete and exclusive statement of the agreement between the Parties. This agreement shall not be modified, amended or in any other way changed without the written consent of both the parties hereto. It is further agreed that this agreement shall supersede all previous agreements, writing, correspondence or communications between the partied hereto.

#### **ARTICLE 15**

#### STAMP DUTY & REGISTERATION

15.1. The stamp duty, penalty, excess stamp-duty, if applicable that is required to be paid in respect of this agreement shall be paid by both Parties equally the Licensee shall give full cooperation to the Licensor for registering this agreement and attend the office of the Registrar whenever directed by the Licensor, the registration fee in respect of this agreement will be paid by both the parties equally.

## **ARTICLE 16**

Any dispute relating to this License Agreement shall be subject to jurisdiction of Courts at Mumbai.

OSYSTEMO CONCOOL

CONCORD ENVIRO SYSTEMS LIMITED

#### **SCHEDULE**

#### DESCRIPTION OF SAID PREMISES

100% of the office premises of the total office premises admeasuring 810.12 Sq. Mts. Situated at 101 & 103 HDIL Towers, Anant Kanekar Marg, Bandra (East), Mumbai-400051. Village Bandra (East) CTS No. 608 (Part)

IN WITNESS WHEREOF the parties hereto have executed this Agreement at Mumbai of this 22<sup>nd</sup> day of September, 2021.

SIGNED BY

For and on behalf of

M/S. CONCORD SHIPPING PRIVATE LIMITED

(Authorized Signatory) (LICENSOR)

In presence of

For and on behalf of

(Authorized Signatory)

M/S. CONCORD ENVIRO SYSTEMS PRIVATE LIMITED

In presence of

(LICENSEE)

Novak

CONCORD ENVIRO SYSTEMS LIMITED

	Creation history	
Prepared by:	Reviewed By:	Approved by:
Arati Shalak	NA	Prayas Goel & Prerak Goe
Document name :	Date created :	Document No :
Misconduct Policy	September 1, 09	CES11
Version No :	Status :	
1.3	Approved	

Version history				
Version	Date	Approver for Change	Author	Description
1.2	September 1, 09	Prayas Goel & Prerak Goel	Arati Shalak	New policy introduced
1.3	January 1, 15	Prayas Goel & Prerak Goel	Arati Shalak	Sexual harassment clause no. revised

Access list					
Document Type	List of Users	Access Type	Type of Media		
Word	All CES employees	Read	Electronic		
Word	Internal HR team	Read, Edit	Electronic		

## Handling guidelines Classification: CONFIDENTIAL

- Do not forward or copy data in part or full without explicit permission of Arati Shalak
- Data access is limited to CES employees and HR

## **Security Notice**

The information contained within this document is highly CONFIDENTIAL. Unauthorized disclosure is prohibited. Failure to observe Rochem Separation Systems Pvt. Ltd. policy regarding proprietary information can result in disciplinary action, including dismissal, as well as result in a violation of CES proprietary rights and subject you and/or third parties to legal liability.

CONCORD ENVIRO SYSTEMS LIMITED

- c. Professional negligence such as but not limited to failure to install antivirus on own machines and protect information, failure to abide by safety guidelines in factory or consistent non-performance.
- d. Abusive or offensive behavior directed towards other employees, client or visitor to CES
- e. Unauthorized use of the property, facilities or services of CES or its clients
- f. Failure to abide by company policies in part or full
- g. Failure to comply, in part or full, with the terms of employment
- h. Violating the principles and / or specifics of CES Code of Business Ethics, in part or full.
- i. Failing a zero tolerance parameter where applicable
- j. Misrepresentation of information at the time of joining

Failure to improve from a level 3 issue, will lead to a Level 4 Misconduct. CES will have the sole discretion to decide whether level three misconduct has been satisfactorily rectified or not.

#### 4. Level Four Misconduct:

Examples of Level Four Misconduct that may lead to dismissal with notice:

- a. Repetition of any offence following a Final Written Warning
- b. Any action above, which is deemed to be of a more serious nature than would warrant only a final written warning
- c. Misrepresentation of information at the time of joining.
- d. Failure to abide by company policies in part or full
- e. Failure to comply, in part or full, with the terms of employment
- Violating the principles and / or specifics of CES Code of Business Ethics, in part or full.

Failure to improve from a level 4 issue, will lead to a Level 5 Misconduct. CES will have the sole discretion to decide whether level four misconduct has been satisfactorily rectified or not.

#### 5. Level Five Misconduct:

Examples of Level Five Misconduct that may lead to summary dismissal:

- a. Fraud such as but not limited to fraudulent benefit claims, invoices, time and expenses, concealment of important information
- b. Theft of property belonging to CES or other employees
- c. Violence directed towards other employees, client or visitor to CES
- d. Willful damage to property belonging to CES or its employees
- e. Being under the influence of, possessing, or supplying of alcohol whilst on the premises of CES
- f. Under the influence of, possessing, supplying, producing and discussing of illegal drugs whilst on the premises of CES
- g. Being charged with a criminal offence by the Police or convicted by a court of law for a criminal offence and / or non-disclosure of the same
- h. Misrepresentation of information at the time of joining
- i. Serious act of insubordination
- Undertaking secondary or additional employment without due permission from approving authorities
- k. Deliberate tampering with or unauthorized use of computer hardware or software, e.g. using unauthorized passwords
- Serious negligence that causes unacceptable loss, damage or injury to CES or other employee
- m. Serious harassment of other employees, client or visitor to CES
- n. Serious infringement of health and safety rules
- o. Breach of rules relating to confidentiality
- p. Failure to comply, in part or full, with the terms of employment
- q. Serious breach of any CES policy
- r. Serious violation of principles and / or specifics of CES Code of Business Ethics

Provided however that in respect of certain acts of misconduct which appear in more than one level of misconduct, CES will have the sole discretion to assign the degree of severity to such act of misconduct as level B. The Disciplinary Process - Based on levels of Misconduct

#### 1. Verbal Warnings

First Verbal warnings may be given for minor misconduct or first offence, in a meeting with the relevant employee, and may include; advice, coaching of coarseling. It is important that the

Anthorised Signatory

CORD ENVIRO SYSTEMS LIMITED

employee understands what improvements need to be made, the timelines involved and that failure to improve may lead to disciplinary proceedings. Supervisor can issue the same with the consent of the Reporting Manager. Ideally the Supervisor, on mail should document the same with a copy to the Employee.

1.1 First Verbal warning

Should be documented by way of a file note and include circumstances surrounding the warning and the date the warning was given and should make clear:

a. The timelines within which this improvement will be required

b. The assistance which will be provided by CES to enable the employee to achieve such an improvement

1.2 Second Verbal Warning

Should be documented and include circumstances surrounding the warning and the date the warning was given, and should make clear:

a. The timelines within which this improvement will be required

b. The assistance which will be provided by CES to enable the employee to achieve such an improvement

#### 2. Written warnings

For all written warnings, Supervisor should document the case and mail it to the Supervisor, HOD the HRR. Supervisor, HOD and HRR should examine the facts of the case along with immediate manager of the Supervisor; and based on the findings a written warning should be

2.1 First Written Warning

This will be issued for more serious incidents, or if there is no improvement after a Verbal Warning has been issued. The warning will be documented by the Reporting Supervisor with the help of HRR. Post examination of the facts of the case, a copy will be given to the employee and a copy attached to their personnel file.

The warning will specify;

- a. The nature of the breach of discipline
- b. The likely consequence of a further breach

The required improvement

c. The required improvementd. The timelines within which this improvement will be required

e. The assistance which will be provided by CES to enable the employee to achieve such an improvement

The stage of the disciplinary process (i.e. first written warning)

The specific timelines during which the warning will remain active on the individual's personnel file, i.e., a minimum of 6 months.

2.2 Final Written warning

This will be given if there is a repetition of a breach of discipline, or if there is insufficient improvement in conduct after a First Written Warning has been issued, or in the case of a first instance of more serious misconduct. The warning will be documented by the Reporting Manager with the help of HRR. Post examination of the facts of the case, a copy will be given to the employee and a copy attached to their personnel file.

The warning will specify;

- a. The nature of the breach of discipline
- b. The likely consequence of a further breach

c. The required improvement

The timelines within which this improvement will be required

The assistance which will be provided by CES to enable the employee to achieve such an improvement

The stage of the disciplinary process (i.e. final written warning) f.

The specific period during which the warning will remain active on the individual's personnel file i.e., minimum of 6 months.

#### 3. Dismissals

3.1 Dismissal with notice

This will be given if there is a repetition of a breach of discipline or if there is insufficient improvement in conduct or performance after a Final Written Warning has been issued. Decisions to dismiss should be made by a Manager or above after discussion with a senior member of HR.

CONCORD ENVIRO SYSTEMS LIMITED

3.2 Summary Dismissal

This is likely to be the appropriate action on the occasion of gross misconduct. Decisions to dismiss should be made by a senior manager or above after discussion with a senior member of HR.

3.3 Investigation

Some offences may warrant not only potential dismissal, but also an investigation before the decision is made. For such situations, a 5 stage disciplinary process needs to be followed, which is as follows;

- 3.3.1 Investigation
- 3.3.2 Disciplinary Hearing
- 3.3.3 Adjournment
- 3.3.4 Decision
- 3.3.5 Appeal

The presence of an HRR is mandatory in any / all disciplinary proceedings. The line manager cannot take any action without the presence of the HRR. In situations where the HRR is an interested or influenced party of the outcome of disciplinary proceedings, an alternate HRR will be assigned to be present during / carry out the disciplinary proceedings.

In some circumstances it may be necessary to suspend the employee prior to the investigation. This would be appropriate in cases of potential gross misconduct or where the individual's Supervisor and HRR agree that it is appropriate. When this decision is taken, there should be no inference of blame. It should be made clear that suspension prior to and / or during an investigation is not considered disciplinary action. The Suspending Supervisor or HRR should not ordinarily be involved in the rest of the disciplinary process, for need of being non-judgmental through the process.

#### Procedure for Suspension

- Suspending Supervisor and HRR to be present at suspension meeting with the employee. The Suspending Supervisor should inform the employee of the disciplinary matter to be investigated.
- 2. A letter explaining the suspension terms together with the reason for the investigation and a copy of this policy should be issued at the suspension meeting. The suspension should be on full pay until the conclusion of the disciplinary process and any queries during the suspension period should be directed to the employee's HRR.
- 3. The employee will be escorted from the office and may have their entry pass surrendered depending on the circumstances of the suspension.

#### 3.3.1 Investigation

- 1. When a disciplinary matter arises, the appropriate HRR, should, in conjunction with the HOD, appoint an independent Investigation Officer and Note Taker (if required) in order to establish the facts promptly before recollections fade.
- The Investigation Officer and Note Taker (if required) must not have previously been involved in the matter. They should complete the investigation as soon as is reasonably practicable after the matter arises. The disciplinary panel chairperson should be appointed at the same time as the Investigation Officer, in preparation for a potential Disciplinary Hearing.

## Procedure for Investigation

- Inform employee in writing that a disciplinary matter is to be investigated and that employee has a responsibility to co-operate with the investigation.
- 2. Send employee a copy of the Disciplinary Procedure.
- 3. Interview the employee about the matter.
- Interview all relevant witnesses and complainant. Ensure all witnesses and the complainant are made aware that statements may be made available to the employee being investigated
- 5. Gather all the facts and background.
- 6. Record all meetings and findings in writing.
- Ensure all witness statements and complainant statements are reviewed by interviewees
- 8. Inform the employee at appropriate intervals about the progress of the investigation if there has been a delay in the investigation.

9. Give feedback on the findings to the appropriate HRR.

CONCORD ENVIRO SYSTEMS LIMITED

10. The Investigation Officer should s prepare a report after giving opportunity to the employee, witnesses and complainants to provide comments on the content before the full report is finalized

11. If not already appointed, the appropriate HRR, HOD should appoint a Disciplinary Panel

comprising 2 appropriate individuals (one from HR).

12. The appointed Disciplinary Panel should consider the evidence in the Investigation Officer's report and decide whether a Disciplinary Hearing will take place.

13. The Investigation Officer should not conduct the Hearing nor should anyone involved in the investigation or suspension. Their seniority will be dependent on the level of seriousness of the matter and the seniority of the individual.

#### 3.3.2 Disciplinary Hearing

- If after consideration of the Investigation Officer's report there is enough evidence to warrant a Disciplinary Hearing, the employee should be informed as soon as possible in writing of the decision. The investigation report and any other documented evidence should be included with the letter.
- Copies of the witness statements obtained during the investigation stage will usually be
  given to the employee if they are relevant to the allegations against them. There may
  be occasions where they cannot be disclosed or it is appropriate to provide only a
  summary for reasons of confidentiality.
- Minutes will be taken for each meeting throughout the investigation and disciplinary process.

#### Procedure for Disciplinary Hearing;

 The Disciplinary Panel (appointed above) will be convened and chaired by a senior person (Senior Manger / HOD) and will consist of; Chairperson (senior employees) or designee, HRR (not previously involved), Additional senior employee or designee (not previously involved) if required Additional attendees; Note Taker (if appropriate), Employee

A mutually convenient time and date should be arranged for the Hearing. Where possible the Hearing should take place within 7 working days of the completion of the

investigation, ensuring reasonable notice is given to the employee.

The Chairperson can at his/her discretion meet with witnesses and the complainant. The Chairperson and employee can at their discretion agree that the employee produces a written statement prior to the Hearing.

The purpose and outcome of the Disciplinary Hearing is;

To consider all the facts

To enable the employee to put forward an explanation for their actions

• To arrive at an appropriate action if necessary

#### 3.3.3 Adjournment

After the hearing it is important to adjourn the meeting to allow time to weigh up the evidence and seek additional advice in order to make a balanced judgment.

#### 3.3.4 Decision

When deciding whether a disciplinary action is appropriate and what form it should take, the disciplinary hearing should consider the following factors:

- a. The nature of the breach
- b. Any precedents
- c. The employee's general record

d. Employee's position

- e. Employee's length of service
- f. Explanation put forward by the employee
- g. Any special, exceptional, circumstances which might make it appropriate to adjust the severity of the penalty.

The employee should be invited back and given the decision and the reasons behind it. The decision should be put in writing and should include their right to appeal.

CONCORD ENVIRO SYSTEMS LIMITED

6

#### 3.3.5 Appeal

The opportunity to appeal against a disciplinary decision is essential to principle of natural justice. The employee has the right to appeal following a disciplinary action. Once received appeals should be dealt with as promptly as possible.

Appeals can be raised on a number of grounds including;

- a. The perceived unfairness of the decision
- b. The severity of the action
- c. New evidence coming to light
- d. Procedural irregularities

#### **Procedure for Appeals**

- An appeal must be lodged within 5 working days from receipt of the decision letter. It is recommended that the appeal will be held within 7 working days from receipt of the appeal request.
- 2. An Appeals Panel will be convened and chaired by a senior person and will consist of Chairperson, not previously involved or designee, HR Manager (not previously involved), Additional senior person or designee (not previously involved) if required, Additional attendees, Note Taker (if appropriate), Employee, Employee's Representative
- The employee should be informed in writing of the arrangements for the Appeal Hearing
- Where new evidence arises during the appeal, the employee should be given the opportunity to comment before any decision is taken.
- 5. The Appeals Panel has the power to cancel, reduce or vary a disciplinary penalty but they cannot increase it.
- 6. The decision of the Appeals Panel is final and binding on all concerned. There is no further right of appeal.
- 7. The employee will be informed of the results of the appeal as soon as is reasonably practicable and this will be confirmed in writing.

#### C. Sexual Harassment of Women

- CES shall abide by the provisions of The Sexual Harassment of Women at Workplace (Prevention, Prohibition and Redressal) Act, 2013 ('the Act') as enacted by the Parliament and applicable to the whole of India. A summary of the important provisions and the redressal mechanism under the Act is provided below.
- 2.1 Sexual Harassment of Women: This term in respect of female employees of CES shall have the meaning assigned to it under the Act. Section 2(n) of the Act defines Sexual Harassment to include any one or more of the following unwelcome acts or behaviour (whether directly or by implication) namely:
  - i. physical contact and advances; or
  - ii. a demand or request for sexual favours; or
  - iii. making sexually coloured remarks; or
  - iv. showing pornography; or
  - v. any other unwelcome physical, verbal or non-verbal conduct of sexual nature;
- 2.2 Workplace: In line with section 2(o) of the Act, workplace shall include any/all CES offices, administrative offices, research laboratories, clinics, any place visited by CES employees arising out of or during the course of employment including transportation provided by CES for undertaking the journey.

#### 3. Constitution of Internal Complaints Committee:

- 3.1 A Committee to be known as the "Internal Complaints Committee" shall be constituted at all administrative units or offices.
- 3.2 The Internal Complaints Committee shall consist of the following members to be nominated by CES, namely:
  - a. a Presiding Officer who shall be a woman employed at a senior level at the concerned CES office from amongst the employees. Provided that in case a senior level woman employee is not available, the Presiding Officer shall be nominated from other offices or administrative units of CES. Provided further that in case the other offices or administrative units of CES do not have a senior level woman employee, the Presiding Officer shall be nominated from any other workplace of CES or other department or organization.

b. Not less than two Members from amongst employees preferably committed to the cause of women or who have had experience in social work or have legal knowledge.

CONCORD ENVIRO SYSTEMS LIMITED

- c. One member from amongst non-governmental organizations or associations committed to the cause of women or a person familiar with the issues relating to sexual harassment. Provided that at least one-half of the total Members so nominated shall be women.
- 3.3 The Presiding Officer and every Member of the Internal Committee shall hold office for such period, not exceeding three years, from the date of their nomination as may be specified by CES.
- 3.4 The Member appointed from amongst the non-governmental organizations or associations shall be paid such fees or allowances for holding the proceedings of the Internal Committee, by CES, as may be prescribed.
- 3.5 Where the Presiding Officer or any Member of the Internal Committee:
  - contravenes the provisions of section 16 of the Act (prohibition of publication or making known contents of complaint and inquiry proceedings); or
  - has been convicted for an offence or an inquiry into an offence under any law for the time being in force is pending against him; or
  - has been found guilty in any disciplinary proceedings or a disciplinary proceeding is pending against him; or
  - has so abused her position as to render her continuance in office prejudicial to the public interest, such Presiding Officer or Member, as the case may be, shall be removed from the Committee and the vacancy so created or any casual vacancy shall be filled by fresh nomination in accordance with the provisions of this section.

#### 4. Complaint of Sexual Harassment:

- 4.1 Any aggrieved woman may make, in writing, a complaint of sexual harassment at work place to the Internal Committee within a period of three months from the date of incident and in case of a series of incidents, within a period of three months from the date of last incident.
  - Provided that where such complaint cannot be made in writing, the Presiding Officer or any Member of the Internal Committee, as the case may be, shall render all reasonable assistance to the man or woman for making the complaint in writing.
  - Provided further that the Internal Committee for the reasons to be recorded in writing, extend the time limit not exceeding three months, if it is satisfied that the circumstances were such which prevented the woman from filing a complaint within the said period.
- 4.2 Where the aggrieved woman is unable to make a complaint on account of her physical or mental incapacity or death or otherwise, her legal heir or such other person as may be prescribed may make a complaint.

#### 5. Conciliation

- 5.1 The Internal Committee may, before initiating an inquiry at the request of the aggrieved woman take steps to settle the matter between her and the respondent through conciliation. Provided that no monetary settlement shall be made as a basis of conciliation.
- 5.2 Where a settlement has been arrived at, the Internal Committee shall record the settlement so arrived and forward the same to CES or the District Officer to take action as specified in the recommendation.
- 5.3 The Internal Committee shall provide the copies of the settlement as recorded to the aggrieved woman and the respondent.
- 5.4 Where a settlement is arrived at, no further inquiry shall be conducted by the Internal Committee

## 6. Inquiry into Complaint

- 6.1 Subject to the above specifications, the Internal Committee shall, proceed to make inquiry into the complaint in accordance with the provisions of 3.3.1 (Investigation), 3.3.2 (Disciplinary Hearing) and 3.3.3 (Adjournment) hereinabove applicable to the respondent
- 6.2 Provided that where the aggrieved woman informs the Internal Committee that any term or condition of the settlement arrived at has not been complied with by the respondent, the Internal Committee shall proceed to make an inquiry into the complaint or, as the case may be, forward the complaint to the police.
- 6.3 Provided further that where both the parties are employees, the parties shall, during the course of inquiry, be given an opportunity of being heard and a copy of the findings shall be made available to both the parties enabling them to make representation against the findings before the Committee.

CONCORD ENVIRO SYSTEMS LIMITED

- 6.4 For the purpose of making an inquiry, the Internal Committee shall have the same powers as are vested in a civil court under the Code of Civil Procedure, 1908 when trying a suit in respect of the following matters, namely:—
  - summoning and enforcing the attendance of any person and examining him on oath;
  - requiring the discovery and production of documents; and
  - any other matter which may be prescribed.
- 6.5 The inquiry shall be completed within a period of ninety days.

7 Action During Pendency of Inquiry

- 7.1 During the pendency of an inquiry, on a written request made by the aggrieved woman, the Internal Committee may recommend to CES —
- transfer the aggrieved woman or the respondent to any other workplace; or
- · grant leave to the aggrieved woman up to a period of three months; or
- grant such other relief to the aggrieved woman as may be prescribed.
- 7.2 The leave granted to the aggrieved woman hereunder shall be in addition to the leave she would be otherwise entitled.
- 7.3 On the recommendation of the Internal Committee CES shall implement the recommendations made hereunder and send the report of such implementation to the Internal Committee.

8. Inquiry report

- 8.1 On the completion of an inquiry, the Internal Committee shall provide a report of its findings to CES, within a period of ten days from the date of completion of the inquiry and such report will be made available to the concerned parties.
- 8.2 Where the Internal Committee, arrives at the conclusion that the allegation against the respondent has not been proved, it shall recommend to CES that no action is required to be taken in the matter.
- 8.3 Where the Internal Committee, arrives at the conclusion that the allegation against the respondent has been proved, it shall recommend to CES—
- to take action for sexual harassment as a misconduct in accordance with the provisions of clause 3 (Dismissals) hereinabove;
- to deduct from the salary or wages of the respondent such sum as it may consider appropriate to be paid to the aggrieved woman or to her legal heirs, as it may determine, in accordance with the provisions of clause 10 (Determination of Compensation) hereinbelow
- 8.4 Provided that in case CES is unable to make such deduction from the salary of the respondent due to his being absent from duty or cessation of employment it may direct to the respondent to pay such sum to the aggreed party:
- 8.5 Provided further that in case the respondent fails to pay the stipulated sum, the Internal Committee may forward the order for recovery of the sum as an arrear of land revenue to the concerned District Officer.
- 8.6 CES shall act upon the recommendation within sixty days of its receipt by it.

9. Punishment for false or malicious complaint and false evidence

- 9.1 Where the Internal Committee, arrives at a conclusion that the allegation against the respondent is malicious or the aggrieved woman or any other person making the complaint has made the complaint knowing it to be false or the aggrieved woman or any other person making the complaint has produced any forged or misleading document, it may recommend to CES, to take action against the woman or the person who has made the complaint in accordance with the provisions of clause 2 (Written Warning) hereinabove:
- 9.2 Provided that a mere inability to substantiate a complaint or provide adequate proof need not attract action against the complainant under this section:
- 9.3 Provided further that the malicious intent on part of the complainant shall be established after an inquiry in accordance with the procedure prescribed, before any action is recommended.

9.4 Where the Internal Committee, arrives at a conclusion that during the inquiry any witness has given false evidence or produced any forged or misleading document, it may recommend to CES, to take action in accordance with the provisions of clause 2 (Written Warning) hereinabove.

CONCORD ENVIRO SYSTEMS LIMITED

10 Determination of Compensation:

- 10.1 For the purpose of determining the sums to be paid to the aggrieved woman, the Internal Committee shall have regard to
- the mental trauma, pain, suffering and emotional distress caused to the aggrieved woman;
- the loss in the career opportunity due to the incident of sexual harassment;
- medical expenses incurred by the victim for physical or psychiatric treatment;
- the income and financial status of the respondent;
- feasibility of such payment in lump sum or in installments.

# 11. Prohibition of publication or making known contents of complaint and inquiry proceedings

- 11.1 Notwithstanding anything contained in the Right to Information Act, 2005, the contents of the complaint made, the identity and addresses of the aggrieved woman, respondent and witnesses, any information relating to conciliation and inquiry proceedings, recommendations of the Internal Committee and the action taken by CES shall not be published, communicated or made known to the public, press and media in any manner:
- 11.2 Where any person entrusted with the duty to handle or deal with the complaint, inquiry or any recommendations or action to be taken under the above provision, contravenes the same, he shall be liable under clause 2 (Written Warning) hereinabove.

#### IV. Exception Approval

All exceptions will be based on specific business need and are required to have the approval of the respective HOD/ Leadership or designate thereof as well as the HR Lead or designate thereof.

## V. Supplementary documentation

#### A. Rationale

It is CES's intention that all performance, attendance and disciplinary issues should be dealt with as quickly as is equitable in the circumstances and that the principles of natural justice should be observed. All matters relating to disciplinary procedures will be treated confidentially.

As far as is possible CES will endeavor to comply with the time limits stated in this policy. However, it may not be possible to comply with these where the investigation requires more time or if the relevant stakeholders are not readily available.

CES reserves the right to commence the disciplinary process at the level considered appropriate for the nature of the issue and the circumstances in which it arose.

## B. Key Terms and Definitions

- Habitual: Implies similar recurring instances within a short period of time. Unless and until
  specified otherwise, it is deemed to be 3 instances in a calendar month.
- Late Attendance: This applies in the case of mandatory shift start timings. Unless and until specified otherwise, it is deemed from the start time of a given shift.
- Harassment: Harassment may be defined as unwanted or offensive behavior by one or more employee(s) towards another. If found, harassment can result in summary dismissal.

In addition to the definition, examples of harassment include:

- a. Display of sexually offensive material, including downloaded material from the Internet by a female employee to a male employee
- Threat of dismissal, loss of promotion, etc, on grounds of disability, gender, race, religion, beliefs, or sexual orientation
- c. Conduct related to an employee's disability, gender, race, religion, beliefs, or sexual orientation which has the purpose or effect of unreasonably interfering with an employee's work performance or creates an intimidating, hostile or offensive working environment
- d. Threatened or actual violence
- e. Jokes or stories related to disability, sex, race, religion, beliefs, or sexual orientation that could be perceived as hostile
- f. Bullying (persistent behavior directed against an individual which is intimidating, offensive or malicious and which undermines the confidence and self-esteem of the recipient)
- g. Unwanted physical attention

CONCORD ENVIRO SYSTEMS LIMITED

These examples are not exclusive or exhaustive. Threat or actual violence of any kind is an obvious example of gross misconduct, which if found is likely to result in summary dismissal. Other actions, however, may also constitute gross misconduct depending on the circumstances and the seriousness of the case in question.

Whilst these guidelines do not cover harassment from client personnel, or other business contacts, employees are urged to report such incidents. Such incidents will again be taken extremely seriously and CES will make every reasonable endeavor to resolve or prevent a reoccurrence. An employee can raise a harassment complaint either informally or formally in line with the Grievance Procedure, depending on circumstances. CES shall ensure that adequate protection will be provided to employees who complain of harassment, who report harassment that they witnessed, or who support a complainant. Every harassment case will be dealt in line with Misconduct Policy.

CONCORD ENVIRO SYSTEMS LIMITED

11

#### ARTICLES OF ASSOCIATION OF CONCORD ENVIRO SYSTEMS LIMITED

(New set of Articles of Association adopted pursuant to a Special Resolution passed at the Extraordinary General Meeting held on August 26, 2024 in substitution of and to the exclusion of the previous Articles of Association)

The Articles of Association of the Company comprise two parts, Part A and Part B. Till the time of listing and trading of the Equity Shares of the Company on any recognised stock exchange in India pursuant to an initial public offering of the Equity Shares of the Company, Part A and Part B shall, unless the context otherwise requires, co-exist with each other. Notwithstanding anything contained herein, in the event of any conflict between the provisions of Part A and Part B of these Articles, the provisions of Parts B of these Articles shall prevail. Upon the commencement of listing and trading of the Equity Shares of the Company on any recognised stock exchange in India pursuant to an initial public offering of the Equity Shares of the Company, Part B shall automatically stand deleted, not have any force and be deemed to be removed from the Articles of Association and the provisions of the Part A shall automatically come in effect and be in force, without any further corporate or other action by the Company, directors, or its shareholders.

#### PART A

#### I - APPLICABILITY OF TABLE F

1. The regulations as contained in Tables F of Schedule I of Companies Act 2013 shall apply to the Company only so far as they are not inconsistent with any of the provisions contained in these Articles or modification thereof or are not expressly or by implication excluded from these Articles.

## II - DEFINITION AND INTERPRETATION

## 2. **Definitions and Interpretation**

## 2.1. **Definition.** In these Articles:

- (a) "Act" means the (Indian) Companies Act, 2013 the rules framed thereunder and any subsequent amendment or re-enactment thereof for the time being in force.
- (b) "Articles" or "these Articles" means the Articles of Association of the Company, as amended from time to time.
- (c) "Associate Company" shall have the meaning ascribed to it in Section 2(6) of the Companies Act, 2013.
- (d) "Board" or the "Board of Directors" means the board of directors of the Company as constituted from time to time.
- (e) "**Business**" means the business of providing environmental engineering solutions with focus on wastewater treatment and recycling, desalination and industrial solid waste management.

- (f) "Business Day" means a day, not being a Saturday or a Sunday or a public holiday, on which banks are open for business in Mumbai, India and Mauritius, in the context of a payment being made to or from a bank in a place other than Mumbai, India or Mauritius, in such other place;
- (g) "Company" means CONCORD ENVIRO SYSTEMS LIMITED.
- (h) "Director" means a director on the Board from time to time.
- (i) "Employees Stock Option Plan" or "ESOP" means any employee stock option plan as formulated by the Company and approved by the Board of the Company and applicable, inter alia, to the employees and to such other persons as are eligible, under applicable Law to receive such options.
- (j) "Equity Securities" means Equity Shares, preference shares, or any other security in the share capital of the Company, debentures or any note or debt security having or containing equity or profit participation features, or any option, warrant or other security or right which is directly or indirectly convertible into or exercisable or exchangeable for Equity Shares of the Company.
- (k) "**Equity Shares**" means the equity shares in the issued, subscribed and paid up equity share capital of the Company.
- (l) "**Financial Year**" means the financial year of the Company, commencing on April 1 every year and ending on March 31 of the following year, or such other financial year of the Company as the Company may from time to time legally designate as its financial year.
- (m) "Government" or "Governmental Authority" means any government, statutory authority, any department, agency or instrumentality of any government, any court, tribunal or arbitral tribunal, board and the governing body of any securities exchange or Stock Exchange(s).
- (n) "INR" means Indian Rupees, the currency and legal tender of the Republic of India, for the time being in force.
- (o) "Law" shall include all applicable statutes, enactments, acts of legislature, laws, ordinances, rules, bye-laws, regulations, guidelines, policies, directions, directives and orders of any Government, and applicable international treaties and regulations, in force at the relevant time.
- (p) "Memorandum" means the memorandum of association of the Company as originally framed or altered from time to time in accordance with the Agreement, the Subscription Agreement and the Act.
- (q) "Month" means English calendar month.

- (r) "Office" means the Registered Office for time being of the Company.
- (s) "**Person**" shall include an individual, proprietorship, Hindu undivided family, partnership, corporation, company, unincorporated organization or association, trust or other entity, whether incorporated or not.
- (t) "**Promoters**" shall mean Mr. Prayas Goel and Mr. Prerak Goel.
- (u) "**Proxy**" includes Attorney duly constituted under a Power of Attorney.
- (v) **Relative**" shall have the meaning ascribed to such term in Section 2 (77) of the Act.
- (w) "Related Party(ies)" shall have the meaning ascribed to such term as defined in Section 2 (76) of the Act.
- (x) "Rules" means any rule(s) made pursuant to section 469 of the Act or such other provisions pursuant to which the Central Government is empowered to make rules, and shall include such rules as may be enacted, re-enacted or amended from time to time.
- (y) "Seal" means the Common Seal of Company.
- (z) "**SEBI**" shall mean the Securities and Exchange Board of India, a body established under the provisions of the Securities and Exchange Board of India Act, 1992.
- (aa) "**SEBI Regulations**" means the Securities and Exchange Board of India (Issue of Capital and Disclosure Requirements) Regulations, 2018, as amended or reenacted from time to time, including any rules or circulars issued thereunder.
- (bb) "**Shareholder**" means the duly registered holder from time to time of the Equity Securities of the Company.
- (cc) "Stock Exchange" means the National Stock Exchange, the BSE Limited or such other stock exchange approved by the Board.
- (dd) "Subsidiaries" means the subsidiaries of the Company.
- (ee) "Tax" or "Taxation" means, to the extent applicable, all forms of taxation, duties, levies imposts and social security charges, present and future, whether direct or indirect including without limitation corporate income tax, wage withholding tax, value added tax, customs and excise duties, capital gains tax, dividend withholding tax, dividend distribution tax, land taxes, environmental taxes and duties and any other type of taxes or duties payable by virtue of any applicable national, regional or local Law or regulation and levied by any Governmental Authority; together with any interest, penalties, surcharges or

fines relating to them, due, payable, levied, imposed upon or claimed to be owed in any relevant jurisdiction.

## 2.2. **Interpretation**

- (a) "In writing and Written" includes printing, lithography and other modes of representing or reproducing words in a visible form.
- (b) Words importing the singular number include the plural number and vice versa.
- (c) Words importing the masculine gender include the feminine gender.
- (d) Reference to a Party shall include, such Party's legal heirs, executors, administrators, successors and permitted assigns and any Persons deriving title under it, as applicable.
- (e) References to the shareholding of any Shareholder in the Company shall (a) refer to the shareholding of such Shareholder computed on a Fully Diluted Basis, and (b) include the shareholding of such Shareholder's Affiliates holding any shares in the Company, if any.
- (f) The terms "hereto", "hereof," "herein," "hereby" and derivative or similar words refer to these Articles and not to any particular clause, article or section of the Articles.
- (g) The word "including" herein shall always mean "including, without limitation".
- (h) The phrases "ordinary course of business" and "ordinary course of business consistent with past practice" refer to the business and practice of the Company as conducted prior date of execution of the Agreement;
- (i) Whenever the Articles refer to a number of days, such number shall refer to calendar days, unless specifically provided otherwise.
- (j) Headings and captions are used for convenience only and shall not affect the interpretation of these Articles.
- (k) Unless the context otherwise requires, words and expression used in these presents shall bear the same meaning as in the Act, or any statutory modification thereof in force at the date when these regulation become binding on the Company.

## III - PUBLIC COMPANY

## 3. **Public Company**

3.1. The Company is a public company within the meaning of Section 2(71) of the Companies Act, 2013 and accordingly:

"public company" means a company which-

- a) is not a private company and
- b) has a minimum paid-up share capital as may be prescribed

Provided that a company which is a subsidiary of a company, not being a private company, shall be deemed to be public company for the purposes of this Act even where such subsidiary company continues to be a private company in its articles.

3.2. The Company may at any time convert itself into a private company within the meaning of and subject to the provisions of the Act and Rules made thereunder.

#### IV - SHARE CAPITAL

## 4. Share Capital

- 4.1. **Authorised Capital.** The Authorised Share Capital of the Company will be as set out under the Memorandum of Association. The Company shall have the power to alter its share capital to the extent permissible under the Act including to increase or reduce the capital, to divide the share in capital for the time being into several class and to attach thereto respectively such preferential, deferred, qualified or special rights, privileges or any condition as may be determined in such manner as may be permitted by the Act or by these Articles and to consolidate or sub-divide the shares and to vary, modify for abrogate such rights, privileges, or conditions in such manner as may for the time being be permitted by the Act or these Articles. The Board shall have power to classify or reclassify the shares into different types.
- 4.2. **Alteration of Capital.** The Company may from time to time change the structure of the capital, alter in any way the conditions of its memorandum relating to capital in accordance with the provisions of section 61 of the Companies Act, 2013 for all or any of the following purposes:
  - a. To increase its authorised share capital by such amount as it thinks expedient;
  - b. To consolidate and divide all or any of its share capital into shares of larger amount than its existing shares, provided that no consolidation and division which results in changes in the voting percentage of shareholders shall take effect unless it is approved by the Tribunal on an application made in the prescribed manner;
  - c. To convert all or any of its fully paid-up shares into stock, and reconvert that stock into fully paid up shares of any denomination;

- d. To sub-divide its shares or any of them into shares of smaller amount than is fixed by the Memorandum, so, however, that in the sub-division, the proportion between the amount paid and the amount, if any unpaid, on each reduced share shall be the same as it was in the case of the share from which the reduced share is derived; and
- e. To cancel any shares which at the date of the passing of the resolution, have not been taken or agreed to be taken by any persons and diminish the amount of its share capital by the amount of the shares so cancelled. Cancellation of shares in pursuance of this sub-clause shall not be deemed to be a reduction of the capital of the Company within the meaning of the Act.
- 4.3. **Kinds of Share Capital**. The Company may issue different kinds of Shares in accordance with these Articles and applicable provisions of the Act and Rules including shares with differential rights as to dividend, voting or otherwise in accordance with the Act and the Rules.
- 4.4. **Reduction.** The Company may (subject to the provisions of Sections 55 of the Act) from time to time, by Special Resolution, reduce in any manner and with, and subject to, any incident authorised and consent required by law:
  - (a) Its share capital;
  - (b) Any capital redemption reserve account; or
  - (c) Any share premium account.
- 4.5. **Variation of Members rights.** If at any time the share capital of the Company is divided into different classes of shares, the rights attached to any class (unless otherwise provided by the term of issue of the shares of that class) may, subject to the provisions of the Act and the applicable Rules, and whether or not the Company is being wound up, be varied with the consent in writing of the holders of not less than <sup>3</sup>/<sub>4</sub> (three-fourth) of the issued shares of that class or with the sanction of a Special Resolution passed at a separate meeting of the holders of the shares of that class.

# V - ISSUE AND ALLOTMENT OF SECURITIES

# 5. Further Issue of securities:

- (a) Where at any time, the Company proposes to increase its subscribed capital by the issue of further shares, such shares shall be offered—
  - (i) to persons who, at the date of the offer, are holders of Equity Shares of the Company in proportion, as nearly as circumstances admit, to the paid up share capital on those shares by sending a letter of offer subject to the following conditions, namely:
    - a. the offer shall be made by notice specifying the number of shares offered and limiting a time not being less than 15 (fifteen) days, or such lesser number of days as may be prescribed, and not exceeding 30 (thirty) days from the date of the offer within which the offer, if not accepted, shall be deemed to have been declined;

- b. the aforesaid offer be deemed to include a right exercisable by the Person concerned to renounce the shares offered to him or any of them in favour of any other Person; and the notice referred to in clause a. above shall contain a statement of this right;
- c. after the expiry of the time specified in the notice aforesaid, or on receipt of earlier intimation from the Person to whom such notice is given that he declines to accept the shares offered, the Board may dispose of them in such manner which is not disadvantageous to the Shareholders and the Company;
- (ii) to employees under a scheme of employees' stock option, subject to special resolution passed by the Company and subject to the Rules and such other conditions, as may be prescribed under Law; or
- (iii) to any Persons, if it is authorised by a special resolution, whether or not those Persons include the Persons referred to in clause (i) or clause (ii) above, either for cash or for a consideration other than cash, at such price as may be determined in accordance with Law, subject to the compliance with the applicable provisions of the Act and any other conditions as may be prescribed under Law.
- (b) The notice referred to in (a) (i) above shall be dispatched through registered post or speed post or through electronic mode to all the existing Shareholders at least 3 (three) days before the opening of the issue.
- (c) Nothing in this Article shall apply to the increase of the subscribed capital of a Company caused by the exercise of an option as a term attached to the Debentures issued or loan raised by the Company to convert such Debentures or loans into shares in the Company:
  - Provided that the terms of issue of such debentures or loan containing such an option have been approved before the issue of such debentures or the raising of loan by a special resolution passed by the Company in a general meeting.
- (d) Notwithstanding anything contained (c) above, where any debentures have been issued or loan has been obtained from any Government by the Company, and if that Government considers it necessary in the public interest so to do, it may, by order, direct that such debentures or loans or any part thereof shall be converted into shares in the Company on such terms and conditions as appear to the Government to be reasonable in the circumstances of the case even if terms of the issue of such debentures or the raising of such loans do not include a term for providing for an option for such conversion.
  - Provided that where the terms and conditions of such conversion are not acceptable to the Company, it may, within sixty days from the date of communication of such order, appeal to the competent authority which shall after hearing the company and the Government pass such order as it deems fit.
- (e) Where the Government has, by an order made under sub-clause (d), directed that any debenture or loan or any part thereof shall be converted into shares in the Company and where no appeal has been preferred to the competent authority under (d) above or where such appeal has been dismissed, the Memorandum of Association of the Company shall, where such order has the effect of increasing the authorized Share Capital of the Company, be altered and the authorized share capital of the Company shall stand increased by an amount equal to the amount of the value of shares which such debentures or loans or part thereof has been converted into.
- (f) The provisions contained in this Article shall be subject to the provisions of Section 42 and Section 62 of the Act, the Rules and the applicable provisions of the Act.

- 6. **Private Placement of Securities.** The Company may in accordance with provisions of the Act and the applicable Rules, issue further securities by way of a private placement of its securities.
- 7. **Power to Issue Redeemable Preference Shares.** Subject to and in accordance with the provisions of the Act and the Companies (Share Capital and Debentures) Rules, 2014, and subject to the provisions on which any Shares may have been issued, the Company shall have the power to issue preference Shares which are, or at the option of Company, liable to be redeemed within a period not exceeding 20 (twenty) years from the date of their issue, and the resolution authorising such issue shall prescribe the manner, terms and conditions of redemption.
- 8. Subject to the provisions of these Articles, the shares shall be under the control of the Directors who may allot or otherwise dispose of the same to such person on such term and conditions and at such time as Directors think fit.
- 9. **Bonus Shares**: The Company is hereby permitted to issue fully paid-up bonus shares to its members in the manner as set out in Section 63 of the Act and Companies (Share Capital and Debentures) Rules, 2014. Bonus shares may be issued in lieu of dividends.
- 10. **Buy- back of Shares:** The Company is hereby permitted to purchase its own Shares or other specified Securities in accordance with the provision of Sections 68, 69 and 70 of the Companies Act, 2013 and other applicable provisions and applicable Rules. If and to the extent permitted by Law, the Company shall also have the power to re-issue the shares or securities so bought back.

The Company may pay commission (whether absolute or conditional) to any person in connection with the subscription or procurement of subscription to its Securities, as may be determined by the Board.

## 11. Shares at the Disposal of the Board of Directors

(a) Subject to the provisions of Section 62 and other applicable provisions of the Act, and these Articles, the shares in the Capital of the Company for the time being (including any shares forming part of any increased Capital of the Company) shall be under the control of the Board who may issue, allot or otherwise dispose of the same or any of them to Persons in such proportion and on such terms and conditions and either at a premium or at par or at discount (subject to compliance with Section 53 of the Act) at such time as they may, from time to time, think fit, to give to any person or persons the option or right to call for any shares either at par or premium or at a discount subject to the provisions of the Act during such time and for such consideration as the Directors think fit, and may issue and allot Shares in the capital of the Company on payment in full or part of any property sold and transferred or for any services rendered to the Company in the conduct of its business and any shares which may be so allotted may be issued as fully paid up shares and if so issued, shall be deemed to be fully paid up shares. Provided that option or right to call shares shall not be given to any Person or Persons without the sanction of the Company in the Shareholders' Meeting.

# VI - CALLS ON SHARES / DEBENTURES

# 12. Calls on Shares/Debentures

- 12.1. The Board of Director may subject to the provisions of Section 49 of the Act from time to time to make such calls as they think fit upon the members in respect of all moneys unpaid on the shares held by them respectively and not by the conditions of allotment thereof made payable at fixed times and each members shall pay the amount of every call so made on him to the person and at the time and place appointed by the Directors. A call may be made payable by installments.
- 12.2. A call shall be deemed to have been made at the time when the resolution of the Board authorizing such call was passed.
- 12.3. Joint-holder of shares shall be jointly as well as severally liable for calls made on shares held by them.
- 12.4. No call shall exceed one fourth of the nominal amount of a share and be made payable within one month after the last preceding call was payable.
- 12.5. Not less than fourteen days' notice on any call shall be given specifying the time and place of payment and to whom such call be paid.
- 12.6. The Board may, if they think fit, receive from any member willing to advance the same, all or any part of the monies uncalled upon any shares held by him even if no part of the amount is called up, and upon all or any of the monies so advanced may (until the same would, but for such advance, become presently payable) pay interest at such rate (not exceeding 12% per annum without the sanction of the Company in the general meeting) as may be agreed upon between the member paying the sum in advance and the Board. Moneys paid in advance of calls shall not, in respect thereof, confer a right to vote or to dividend or to participate in profits. The Directors may at any time repay the amount so advanced upon giving to such member 3 (three) Months' notice in writing. Provided that option or right to call of Shares shall not be given to any person or persons without the sanction of the Company in the General Meeting.

# 13. Payment in anticipation of call may carry interest

- (a) The Board may, if it thinks fit (subject to the provisions of Section 50 of the Act) agree to and receive from any Shareholder willing to advance the same, the whole or any part of the money due upon the shares held by him beyond the sums actually called up, and upon the amount so paid or satisfied in advance or so much thereof as from time to time and at any time thereafter as exceeds the amount of the calls then made upon and due in respect of the shares in respect of which such advance has been made, the Company may pay interest, as the Shareholder paying such sum in advance and the Board agree upon in accordance with the provisions of the Act, provided that the money paid in advance of calls shall not confer a right to participate in profits or dividend. The Directors may at any time repay the amount so advanced.
- (b) No Shareholder shall be entitled to voting rights in respect of the money (ies) so paid by him until the same would but for such payment, become presently payable.
- (c) The provisions of these Articles shall mutatis mutandis apply to the calls on Debentures of the Company.

#### VII - SECURITIES CERTIFICATE

#### 14. Securities Certificate

- 14.1. The Company shall issue its securities in dematerialized form.
- 14.2. Unless the conditions of issue of any shares or debentures provide otherwise, the Company shall, within (i) 2 (two) Months after the allotment of any of its shares; (ii) within 1 (one) Month from the date of receipt of transfer request from the Depository Participant or, as the case may be, of the intimation of transmission, credit the shares to the holders demat accounts as the case may be in accordance with the provisions of the applicable laws for the time being in force.
- 14.3. In case the law permits issue physical issue of shares, every certificate shall be under the Seal of the Company and shall specify the share to which it relates and amount paid thereon, and shall bear signature of at least two directors of the Company, and the secretary of the company or any person authorized by the Board of Directors in this regard.
- (a) If any certificate be worn out, defaced, mutilated or torn or if there be no further space on the back thereof for endorsement of transfer, then upon production and surrender thereof to the Company, a new certificate may be issued in lieu thereof, and if any certificate is lost or destroyed then upon proof thereof to the satisfaction of the Company and on execution of such indemnity as the Company deems adequate, being given, a new certificate in lieu thereof shall be given to the party entitled to such lost or destroyed certificate, within a period of 30 days from the receipt of such lodgement. Every certificate under the Articles shall be issued without payment of fees if the Directors so decide, or on payment of such fees (not exceeding Rupees two for each certificate) as the Directors shall prescribe. Provided that, no fee shall be charged for issue of a new certificate in replacement of those which are old, defaced or worn out or where there is no further space on the back thereof for endorsement of transfer.

Provided that notwithstanding what is stated above the Directors shall comply with such rules or regulation or requirements of any Stock Exchange or the rules made under the Act or rules made under Securities Contracts (Regulation) Act, 1956 or any other Act, or rules applicable thereof In this behalf.

The provision of this Article shall mutatis mutandis apply to debentures of the company.

- 14.4. In respect of share or shares held jointly by several persons, the Company shall not be bound to issue more than one certificate and delivery and certificate for all securities to one of several joint shall be sufficient delivery to all such security holders.
- 14.5. The Company shall observe rules and conditions as may be prescribed under the Act and the Companies (Shares and Debentures) Rules, 2014 for renewal of security certificates or issue of duplicate security certificates.

# VIII - FORFEITURE OF SHARES

#### 15. Forfeiture of Shares

- 15.1. If a member fails to pay any call within the time allowed by the Board of Directors at the time of making a call or subsequently, the Board may issue final notice naming a further date (not being earlier than the expiry of fourteen days from the date of service of the notice) on or before which the payment required by the notice is to be made and if the calls remain unpaid till then, the shares may be subject to forfeiture without further notice. Such forfeiture shall include all dividends declared in respect of the forfeited shares and not actually paid before the forfeiture.
- 15.2. Forfeited shares may be restored to the holder thereof on payment of the call money in full, before the forfeited shares are sold to other member of members of the Company or otherwise disposed of.
- 15.3. Any shares so forfeited shall be deemed to be the property of the Company and may be sold, re-allotted or otherwise disposed of either to existing members or to any other person upon such terms and conditions and in such manner as the Board shall think fit.
- 15.4. A person whose shares have been forfeited shall cease to be a member in respect of the forfeited shares, but shall notwithstanding the forfeiture, be liable to pay to the Company all moneys, which at the time of forfeiture were presently payable by him to the Company in respect of the share.
- 15.5. The Company may receive the consideration, if any, given for the shares on any sale or disposal thereof and may execute a transfer of the share in favour of the person to whom the share is sold or disposed of. The transfer thereupon shall be registered as the holder share shares.
- 15.6. Neither a judgment nor a decree in favour of the Company for calls or other moneys due in respect of any Shares nor any part payment or satisfaction thereof nor the receipt by the Company of a portion of any money which shall from time to time be due from any Member in respect of any Shares either by way of principal or interest nor any indulgence granted by the Company in respect of payment of any such money shall preclude the forfeiture of such Shares as herein provided. There shall be no forfeiture of unclaimed dividends before the claim becomes barred by applicable Law.

#### IX - TRANSFER OF SHARES

- 16. **General:** A common form of transfer should be used.
  - 16.1. **Director's powers to refuse to register a transfer.** Subject to the provisions of Sections 58 and 59 of the Act, these Articles and other applicable provisions of the Act or any other Law for the time being in force, the Board may, refuse to register the transfer of, or the transmission by operation of law of the right to, any securities or interest of a Shareholder in the Company. The Company shall, within 30 (thirty) days from the date on which the instrument of transfer, or the intimation of such

transmission, as the case may be, was delivered to the Company, send a notice of refusal to the transferee and transferor or to the Person giving notice of such transmission, as the case may be, giving reasons for such refusal.

Provided that, registration of a transfer shall not be refused on the ground of the transferor being either alone or jointly with any other Person or Persons indebted to the Company on any account whatsoever except when the Company has a lien on the shares.

Subject to the applicable provisions of the Act and these Articles, the Directors shall have the absolute and uncontrolled discretion to refuse to register a Person entitled by transmission to any shares or his nominee as if he were the transferee named in any ordinary transfer presented for registration, and shall not be bound to give any reason for such refusal and in particular may also decline in respect of shares upon which the Company has a lien.

#### 16.2. **Instrument of Transfer**

(a) The instrument of transfer of any Share shall be in writing and all the provisions of the Act shall be duly complied with in respect of all transfer of Shares and registration thereof. The Company shall use the form of transfer, as prescribed under the Act, in all cases. In case of transfer of Shares, where the Company has not issued any certificates and where the Shares are held in dematerialised form, the provisions of the Depositories Act shall apply.

In accordance with Section 56 of the Act, the Rules and such other conditions as may be prescribed under Law, every instrument of transfer of shares held in physical form shall be in writing. In case of transfer of shares where the Company has not issued any certificates and where the shares are held in dematerialized form, the provisions of the Depositories Act shall apply. The Company shall cause to be kept a register and index of members / beneficial owners in accordance with all applicable provisions of the Companies Act, 2013 and the Depositories Act, 1996 with details of shares held in physical and dematerialised forms in any medium as may be permitted by law including in any form of electronic medium. The Company shall be entitled to keep in any State or Country outside India a branch register of beneficial owners / Register of Members, resident in that State or Country. The Register and Index of Beneficial Owners maintained by a Depository under the Depositories Act, shall be deemed to be the Register and Index (if applicable) of Shareholders and Securityholders for the purposes of these Articles.

- (b) (i) An application for the registration of a transfer of the shares in the Company may be made either by the transferor or the transferee within the time frame prescribed under the Act.
  - (ii) Where the application is made by the transferor and relates to partly paid shares, the transfer shall not be registered unless the Company gives notice of the application to the transferee in a prescribed manner and the transferee communicates no objection to the transfer within 2 (two) weeks from the receipt of the notice.
- (c) No fee shall be charged for registration of transfer, transmission, probate, succession certificate and letters of administration, certificate of death or marriage, power of attorney or similar other document.

#### X - TRANSMISSION OF SHARES/ DEBENTURE

#### 17. Transmission of Shares/Debentures

- 17.1. On the death of a member, the survivor or survivors where the member was a joint holder, and his nominee or nominees or the heir of the deceased or his legal representative shall be the persons recognized by the Company as having any title or interest in the shares/ debentures.
- 17.2. Nothing in these Articles, shall release the estate of a deceased joint holder from any liability in respect of any share/debenture which has been jointly held by him with other persons.
- 17.3. Any heir or successor being owner of the shares in consequence of the death of the member, upon such evidence being produced as may properly be required by the Board elect either (i) to register his or their names as the holders of the shares or (ii) to make arrangement to transfer of shares in the way as the deceased member could have made in accordance with the transfer clause of these Articles.
- 17.4. If the person or persons aforesaid shall elect to transfer the shares, he shall testify his election by executing a transfer of the share, in conformity with the transfer clause of these Articles.
- 17.5. A person becoming holder of share on such transmission, shall become member of the Company entitled to the same dividends and advantages as his predecessors would have otherwise enjoyed, but if he be a minor, lunatic or insolvent, he shall have no voting right.
- 17.6. The person so becoming entitled and willing to elect to be registered holder of the share shall deliver or send to the Company a notice in writing stating that he so elects.
- 17.7. The Board shall register his name in place of the deceased member.
- 17.8. If the persons aforesaid fails to elect either to be registered himself to transfer the share, the Board on having confirmation of the event of the death of the member shall serve a notice to person asking him to comply with the requirement within three months of the issue of the notice failing which Board may thereafter withhold payment of all dividends, bonuses or other money payable in respect of the share.
- 17.9. If nothing is done within three months, the Company shall issue a notice in the local newspaper asking the legal heirs or the legal representative of the deceased member whether to be registered as shareholders or to transfer the said shares in conformity with these Article three months thereafter, and if nothing is done within the said period, the shares shall be liable to forfeiture and the forfeiture clause of these Articles shall prevail. The forfeited shares may be re-issued to the member at the called up value of the shares and forfeited amount be transferred to the Capital Reserve Account or dealt with as the Board of Directors shall think fit.

- 17.10. The Company shall have a first and paramount lien on (i) every share, not being a fully paid share, for all moneys, whether presently payable or not, called or payable in respect of the share and (ii) on all shares, not being fully paid shares, standing in the name of any person, for all moneys, presently payable by him, provided that the Board of Directors may at any time declare any share to be wholly or in part exempt from the provisions of the clause.
- 17.11. The Company's lien, if any, on a share shall extend to all dividends payable thereon.
- 17.12. For the purpose of enforcing such lien as aforesaid the Directors may sell the Shares subject thereto in such manner as they shall think fit, but no sale shall be made:
  - (a) Unless a sum in respect of which the lien exists is presently payable; and
  - (b) Until the expiration of seven days after a notice in writing, stating and demanding payment of such part of the amount in respect of which the lien exists and as is presently payable, has been given to the registered holder for the time being of the Share or to the person entitled thereto by transmission, and default shall have been made by him in payment of the sum payable as aforesaid for seven days after such notice.

The residue, if any, shall be paid to the person entitled by transmission to the shares so sold at the date of the sale.

17.13. The fully paid up Shares shall be free from all lien and in the case of partly paid up Shares the Company's lien shall be restricted to moneys called or payable at a fixed time in respect of such Shares.

# 18. **COMPANY'S LIEN:**

#### i. On shares:

- (a) The Company shall have a first and paramount lien:
  - (i) on every share (not being a fully paid share), for all money (whether presently payable or not) called, or payable at a fixed time, in respect of that share;

Provided that the Board may, at any time, declare any shares wholly or in part to be exempt from the provisions of this Article.

- (b) Company's lien, if any, on such partly paid shares, shall extend to all Dividends payable and bonuses declared from time to time in respect of such shares.
- (c) Unless otherwise agreed, the registration of a transfer of shares shall operate as a waiver of the Company's lien, if any, on such shares. The fully Paid up shares shall be free from all lien and that in case of partly paid shares, the Company's lien shall be restricted to money called or payable at a fixed time in respect of such shares.
- (d) For the purpose of enforcing such lien, the Board may sell the shares, subject thereto in such manner as they shall think fit, and for that purpose may cause to be issued a duplicate certificate in respect of such shares and may authorise one of their Shareholders to execute and register the transfer thereof on behalf of and in the name of any purchaser. The

purchaser shall not be bound to see to the application of the purchase money, nor shall his title to the shares be affected by any irregularity or invalidity in the proceedings in reference to the sale.

Provided that no sale shall be made:

- (i) unless a sum in respect of which the lien exists is presently payable; or
- (ii) until the expiration of 14 days after a notice in writing stating and demanding payment of such part of the amount in respect of which the lien exists as is presently payable, has been given to the registered holder for the time being of the share or the person entitled thereto by reason of his death or insolvency.

The net proceeds of any such sale shall be received by the Company and applied in payment of such part of the amount in respect of which the lien exists as is presently payable. The residue, if any, shall (subject to a like lien for sums not presently payable as existed upon the shares before the sale) be paid to the Person entitled to the shares at the date of the sale.

(e) No Shareholder shall exercise any voting right in respect of any shares registered in his name on which any calls or other sums presently payable by him have not been paid, or in regard to which the Company has exercised any right of lien.

#### ii. On Debentures:

- (a) The Company shall have a first and paramount lien:
  - (i) on every Debenture (not being a fully paid Debenture), for all money (whether presently payable or not) called, or payable at a fixed time, in respect of that Debenture;

Provided that the Board may, at any time, declare any Debentures wholly or in part to be exempt from the provisions of this Article.

- (b) Company's lien, if any, on the Debentures, shall extend to all interest and premium payable in respect of such Debentures.
- (c) Unless otherwise agreed, the registration of a transfer of Debentures shall operate as a waiver of the Company's lien, if any, on such Debentures. The fully paid up Debentures shall be free from all lien and that in case of partly paid Debentures, the Company's lien shall be restricted to money called or payable at a fixed price in respect of such Debentures.
- (d) For the purpose of enforcing such lien, the Board may sell the Debentures, subject thereto in such manner as they shall think fit, and for that purpose may cause to be issued a duplicate certificate in respect of such Debentures and may authorize the debenture trustee acting as trustee for the holders of Debentures or one of the holder of Debentures to execute and register the transfer thereof on behalf of and in the name of any purchaser. The purchaser shall not be bound to see to the application of the purchase money, nor shall his title to the Debentures be affected by any irregularity or invalidity in the proceedings in reference to the sale.

Provided that no sale shall be made:

- (i) unless a sum in respect of which the lien exists is presently payable; or
- (ii) until the expiration of 14 days after a notice in writing stating and demanding payment of such part of the amount in respect of which the lien exists as is presently payable,

has been given to the registered holder for the time being of the Debenture or the Person entitled thereto by reason of his death or insolvency.

The net proceeds of any such sale shall be received by the Company and applied in payment of such part of the amount in respect of which the lien exists as is presently payable. The residue, if any, shall (subject to a like lien for sums not presently payable as existed upon the Debentures before the sale) be paid to the Person entitled to the Debentures at the date of the sale.

(e) No holder of Debentures shall exercise any voting right in respect of any Debentures registered in his name on which any calls or other sums presently payable by him have not been paid, or in regard to which the Company has exercised any right of lien.

#### XI - GENERAL MEETING

# 19. **General Meeting**

- 19.1. An annual general meeting of the Shareholders of the Company shall be held within six (6) months of the end of each financial year of the Company subject to any extension provided by the registrar of companies. Subject to the foregoing, the Board or the Parties may convene an extraordinary general meeting of the Shareholders of the Company whenever they deem appropriate.
- 19.2. At least Twenty-one (21) days prior written notice of every general meeting of members shall be given to all Shareholders whose names appear on the register of members of the Company. A general meeting may be called after giving shorter notice than that specified above if consent, in writing or by electronic mode, is accorded thereto—
  - (i) in the case of an annual general meeting, by not less than ninty-five per cent. of the members entitled to vote thereat; and
  - (ii) in the case of any other general meeting, by members of the company majority in number of members entitled to vote and who represent not less than ninety-five per cent. of such part of the paid-up share capital of the company as gives a right to vote at the meeting; or
- 19.3. The notice of each general meeting of Shareholders shall include an agenda setting out the place, date and time and business proposed to be transacted at the meeting, together with copies of all relevant papers connected therewith and/or proposed to be placed before or tabled at the general meeting and not business shall be transacted at such meeting unless the same has been stated in the notice convening the meeting.
- 19.4. The Chairman of a general meeting of the Company shall not have any second or casting vote. The Chairman of the Board shall be the Chairman for all general meetings.
- 19.5. Any shareholder of the Company may appoint another Person as his proxy (and in case of a corporate shareholder, an authorized representative) to attend a meeting and vote

thereat on such shareholder's behalf, provided that the power given to such proxy must be in writing.

19.6. Quorum. The quorum for a general Meeting of the Shareholders, shall be as provided under the Act.

#### XII - DIRECTORS

#### 20. **Directors**

- 20.1. Qualification Shares. A Director shall not be required to hold any qualification shares.
- 20.2. A director shall not be disqualified by reason of his holding any other office or place of profit under the company other than those mentioned in Section 188 of the Companies Act, 2013 in conjunction with his office of director, except that of auditor and may be appointed thereto upon such terms as to remuneration, tenure of office, as the case may, with the previous consent of the Company, accorded by a special resolution.
- 20.3. Subject to the provisions of Section 188 of the Act, the Directors shall not be disqualified from contracting with the Company either as vendors, purchasers or otherwise, nor shall any such contract or arrangement entered into by or on behalf of the Company with any company or partnership in which any director so contracting being member or so interested, be liable to account to the Company for any profit realized by any such contract or arrangement by reason of such director holding that office or of the fiduciary relation thereby established but the nature of his interest must be disclosed by him at the meeting of the Board of Directors at which the contract or arrangement is determined on, if the interest then exists, or in any other case at the first meeting of the Board of directors after the acquisition of the interest. Provided, nevertheless that such Director shall be entitled to present at the meeting and to vote as Director in respect of the contract or arrangement in which he is so interested as aforesaid and if he does so, vote shall be counted. No general notice and no renewal thereof shall be of effect unless either it is given at a meeting of a Board or the director concerned takes reasonable steps to secure that it is brought up and read at the first meeting of the Board after it is given.
- 20.4. The payment of sitting fees shall be governed by the policy of the Company in this regard as may be in force from time to time.
- 20.5. The Board of directors may fix such remuneration to be paid to a Director of the Company as they think fit, subject to section 188 of the Act, if applicable.
- 20.6. In addition to the remuneration payable to them in pursuance of these Articles, the Directors may be paid all traveling, hotel and other expenses property incurred by them connection of the business of the Company as the Board may determine.
- 20.7. No director shall retire at any general meeting of the Company and subject to the provisions of the Act and these Articles, a Director shall continue to act as such until

- death, resignation and/or removal, as the case may be, unless tenure of employment is fixed by the Board in writing.
- 20.8. The Board of directors shall have powers to appoint the Directors nominated by any lending financial institution. Such director shall not be required to retire by rotation until the arrangement with the said financial institution continues.
- 20.9. The Board shall also have power to fill a casual vacancy in the Board. Such casual vacancy shall be filled up by the Board of Directors at a meeting of the Board. Any person so appointed shall hold office only upto the date upto which the Director in whose place he is appointed would have held office if it had not been vacated as aforesaid, but he shall then be eligible for election.
- 20.10. **Managing Director.** The Board may subject to provisions of the Act appoint any person to be the Managing Director of the Company and may fix the remuneration to be paid to him and to determine to power exercisable and duties to be performed by such Managing Director. Subject to applicable law, an individual may be appointed as the Managing Director and Chairman of the Company at the same time.
- 20.11. **Company Secretary.** A company secretary may be appointed upon such terms and conditions as may be fixed by the Board.
- 20.12. **Additional Director**. An additional director may be appointed by the Board, upon such terms and conditions as may be fixed by the Board.

### XIII - PROCEEDINGS OF BOARD

## 21. **Proceedings of Board**

- 21.1. <u>Number of Board Meetings</u>. The Board shall meet at least four (4) times in every calendar year *provided that*, not more than 120 (one hundred and twenty) days shall intervene between 2 (two) consecutive Board meetings. Subject to applicable Law, a Board meeting may also be held by video conferencing and/or any other permitted means remote participation.
- 21.2. <u>Convening meetings of the Board</u>. Any Director may, and the secretary of Company, if so appointed, shall on the requisition of a Director, summon a meeting of the Board, in accordance with the notice and other requirements set out below in Article 21.3 (Notice for Board Meetings).
- 21.3. <u>Notice for Board Meetings</u>. A Board meeting may be called by a Director and 14 (fourteen) days' written notice of each meeting of the Board or a committee thereof shall be given to each of the Directors at the address notified from time to time by each of them, in writing to the Company, whether in India or abroad, *provided that* a meeting may be convened by a shorter notice with written consent (which may be signified by letter, facsimile or e-mail with receipt acknowledged) of all the Directors..

# 21.4. Contents of the Notice.

- (a) Every notice of a Board meeting of the Company and/ or the Subsidiaries shall set forth in full and sufficient detail each item of the business to be transacted thereat, and no item or business shall be transacted at such Board meeting, unless the same has been stated in full and in sufficient detail in the notice convening the meeting, except as otherwise consented to by all the Directors, or their respective alternate directors.
- (b) The draft resolutions and other documents for all matters to be considered at the Board meetings of the Company and/ or the Subsidiaries must be furnished to all the Directors along with the notice for such Board meeting.
- 21.5. <u>Decisions of the Board.</u>), A decision shall be said to have been made and/or a resolution passed at a meeting of the Board of the Company only if at a validly constituted meeting, such decisions are approved of by a majority of the Directors, present and voting at such Board meeting. In the event there is a vacancy on the Board and an individual has been designated to fill such vacancy, the first order of business shall be to fill such vacancy.
- 21.6. <u>Chairman.</u> Each meeting of the Board shall be chaired by the Chairman nominated by the Board. The Chairman shall not have a casting vote, to resolve a deadlock in voting. In the absence of the Chairman at any meeting, the Board shall elect one of their members to chair the meeting in question.
- 21.7. Resolution by circulation or written consent Except for those actions required by the Act to be determined at a meeting of the Board, all decisions of the Board may be taken by circular resolution. No resolution shall be deemed to have been duly passed by a Board of the Company and/ or the Subsidiaries, or a committee thereof by circulation or written consent, unless the resolution has been circulated in draft, together with the information required to make a fully-informed good faith decision with respect to such resolution, to all Directors, or to all members of the relevant committee, at their usual address (whether in India or abroad) or through electronic means, and has been approved, by majority of the Directors.
- 21.8. Removal/Resignation of Directors. The Company and the Shareholders shall not remove any nominee Director. Each Party entitled to nominate a Director shall have the right to require the removal of such nominee at any time and shall be entitled, to nominate another Person as the nominee Director in place of the Person removed. In the event of the resignation, retirement or vacation of office of the Director nominated by any Party, such Party shall be entitled to nominate another Person as Director in place of such Director and the Shareholders shall exercise their rights in such manner so as to cause the appointment of such Person as nominee Director to the Board of the Company.

- 22. Subject to section 179 of the Companies Act, 2013, the Board may from time to time at their discretion raise and borrow and may themselves lend and secure the payment of any sum or sums of money for the purpose of the Company.
- 23. The Board may raise or secure the repayment of such sum or sums in the manner and upon such terms and conditions in all respect as they deem fit and particular by creation of any mortgage or charge on the undertaking of the whole or any part of the property, or future, or uncalled capital of the Company or by the issue of bonds, redeemable debentures or denture-stock of the Company charged upon all or any part of the property of the Company both present and future including its uncalled capital for the time being.
- 24. All monies hundies, bills of exchange, notes belonging to the Company shall be deposited with one or more banks as the Board of Directors may decide, into an account to be opened in the name of the Company. Cheques on the banks shall be signed by one or more of the managing directors or working directors or by the members of the staff of the Company to whom the Board assign power of attorney or empower him/them by the Board resolution.
- 25. The Board of Directors are authorized to open accounts with in any bank in the name of the Company and operate upon those accounts and to invest the funds of the Company in Government or other securities of any kind whatsoever and to collect interest on the same from time to time and/or endorse, transfer or otherwise deal with securities of the company.
- 26. The Board of Director by proper resolutions shall have the power to do all or any of the acts mentioned in the Memorandum of Association subject to section 179 of the Act.
- 27. The Company shall comply with the provisions of the Act, or any statutory modification thereof for the time being in force and in particular in regard to the registration of particulars of mortgages and charges affecting the property of the Company or created by it, and to keeping a register of the Directors and to send to the Registrar of Companies an annual return, statutory returns and a summary of particulars relating thereto and notice of any increase of share capital, and copies of special resolutions.
- 28. Debentures, Debenture-Stock, bonds, or other securities may be made assignable free from any equities between the Company and the person to whom the same may be issued.
- 29. Subject to applicable Law and these Articles, any debenture, debenture stock, bonds, or other securities may be issued at discount, premium or otherwise and may be issued on condition that they shall be convertible into shares of any denomination and with any privileges and conditions as to redemption, surrender, drawing, allotment of shares, attending (but not voting) at the General Meeting, appointment of Directors and otherwise Debentures with the right to conversion into or allotment of shares shall be issued only with the consent of the company in the General Meeting by a Special Resolution.

## XV - DISQUALIFICATIONS OF DIRECTORS

- 30. The office of a director shall become vacant:
  - 30.1. If by notice in writing to the Company a Director resigns the office of director,

- 30.2. If a Director ceases to be a director by virtue of the Act or becomes disqualified in accordance with provisions of Section 164 of the Companies Act, 2013,
- 30.3. If a Director adjudged insolvent, found lunatic or of unsound mind or involved in such offence that tell upon him as committing an act moral turpitude in the eyes of law or does such act during discharge of his duties as director as tantamount to criminal breach of trust.

# XVI - ACCOUNTS AND AUDIT

- 31. The Board of Director shall cause true account to be kept of the sums of money received and expended by the Company, all sales and purchases of goods by the Company and the matters in respect of which such receipt and expenditure take place and of the assets, credit and liabilities of the Company.
- 32. The books of accounts shall be kept the registered office of the Company or at such other place or places as the Board of the directors of the Company in writing decide and shall be open to inspection by the directors during business hours.
- 33. The Company may keep such books of account or other relevant papers in electronic mode in the manner as prescribed under the Companies (Accounts) Rules, 2014, if the Board so decides.
- 34. The Board shall from time to time determine whether and to what extent and at what time and places and under what conditions or regulation the accounts, books and documents of the Company and any of them shall be open to the inspection by the members not being the Directors and no member (not being a Director) shall have any right of inspecting any account or books or document of the Company except as conferred by statute or authorized by the Board or by the Company in general meeting.
- 35. Once at least in every year the accounts of the Company shall be examined and the correctness of the Balance Sheet and Profit and Loss Account ascertained by one or more statutory auditor or auditors.
- 36. The Company shall, at the annual general meeting, appoint an individual or a firm as an auditor who shall hold office for such period as determined by the Board and the manner and procedure of selection of auditors shall be as prescribed under Section 139 of the Companies Act, 2013 and other applicable provisions of the Act and the Companies (Audit and Auditors) Rules, 2014, provided that the Company shall place the matter relating to such appointment for ratification by members at every annual general meetings. Eligibility, qualifications, disqualification, powers, duties and removal of auditors shall be as prescribed under the Act and Companies (Audit and Auditors) Rules, 2014 as applicable to a public company.
- 37. The Company may at a general meeting remove any such auditor or all or any of such auditors and appoint in his or their places any other person or persons who have been nominated for appointment by any member of the Company and whose nomination has been given to the

members of the Company not earlier than three months but at least fourteen days before the date of the meeting.

## XVII - DIVIDENDS AND RESERVES

- 38. Subject to applicable Law, the Board may from time to time recommend and pay to the member such interim divided as will appear it to be justified by the profit of the Company and the final divided including interim dividend have to be approved by the general body meeting held thereafter, which shall be payable to such members whose names appear in the members register on the date of declaration at the general body meeting.
- 39. The general body can't declare dividend more than recommended by the Board.
- 40. The Board recommend any dividend, but before that set aside out of the profit of the Company such sum as it think proper as a reserves which shall at the discretion of the Board be applicable for any purpose to which the profit of the Company may be properly applied, including provision for meeting contingencies or for equalizing dividends and pending such application, may, at the discretion, either be employed in the business of the company or to be invested in such investments (other than shares of the Company) as the Board may from time to time, think fit.
- 41. The Board may also carry toward any profit which it may think prudent not to divide, with or without setting aside any profit as reserve, but such decision must be supported by cogent reasons recorded in the Boards resolution and must be approved by the members in a following general meeting.
- 42. Subject to the rights of the persons, if any, entitled to shares with special rights as to dividends, all dividends shall be declared and paid in cash according to and in proportion to the amounts paid or credited as paid in respect whereof the dividend is to be paid but if and so long as there be any arrears in any of the calls of the shares in the Company, dividends may be declared according to the amount of the shares already paid and the amounts of dividends may be adjusted against arrears of such calls on the shares.
- 43. Except as otherwise provided in these Articles, all dividends shall be apportioned and paid proportionate to the amount paid or credited as paid on the shares as on the last day of the year but if any share is issued on terms providing that they shall rank for dividend as from a particular date such share shall rank for divided accordingly.
- 44. The Board may deduct from any divided payable to any member all sums of money, if any presently payable be him to the Company on account of calls or otherwise in relation to the share of the Company.
- 45. Every divided warrant may be sent by post to last registered address of the member entitled thereto and the receipt by the person whose name at the date of declaration of the dividend appears on the Register of Members as the owner of the share or in case of joint holder, the receipt by any one of such holders shall be a good discharge to the Company for all payments made in respect of such dividend.

## 46. Right to unpaid or unclaimed dividend

- (a) If the Company has declared a Dividend but which has not been paid or the Dividend warrant in respect thereof has not been posted or sent within 30 (thirty) days from the date of declaration, transfer the total amount of dividend, which remained unpaid or unclaimed within 7 (seven) days from the date of expiry of the said period of 30 (thirty) days to a special account to be opened by the Company in that behalf in any scheduled bank or private sector bank.
- (b) Any money so transferred to the unpaid Dividend account of the Company which remains unpaid or unclaimed for a period of 7 (seven) years from the date of such transfer, shall be transferred by the Company to the Fund established under sub-section (1) of Section 125 of the Act, viz. "Investors Education and Protection Fund".
- (c) No unpaid or unclaimed Dividend shall be forfeited by the Board before the claim becomes barred by Law and such forfeiture, if effected, shall be annulled in appropriate cases.

## XVIII - THE SEAL

- 47. The Directors shall provide for the safe custody of the seal.
- 48. The seal of the Company shall not be affixed to any instrument except by authority of a resolution of the Board or of a committee of the Board authorised by it in that behalf, and except in the presence of at least one Director and of the secretary or such other person as the Board may appoint for the purpose; and that one Director and the secretary or other person aforesaid shall sign every instrument to which the seal of the Company is so affixed in their presence.

# XIX - INDEMNITY

- 49. The Directors, manager, auditor, secretary and other officers for the time being of the Company and any trustees for the time being acting in relation to any affairs of the Company and the heirs, executors and administrators respectively shall be indemnified for the losses which any of them shall or may incur or sustain by reason of any act done or committed in or about the execution of their duty in their respective offices of trust (except such ) (if any) as they shall incur or sustain by or through their own willful neglect or default respectively) and no such officer or trustee shall be answerable for the acts, receipts, neglects or defaults of any other officer or trustee or for joining in any receipt, for the sake of conformity or for the solvency or honesty of any bankers or other person with whom any money or effects belonging to the Company may be lodged or deposited for the safe custody or for any insufficiency or deficiency or any security upon which any money of the Company shall be invested or for any other loss or damages due to any such cause as aforesaid or which may happen in or about the execution of his office or trust unless the same shall happen through the willful neglect or default of such office or trustee. This Article is subject to section 197 of the Act.
- 50. Every officer or agent for the time being of the Company shall be indemnified out of the assets of the Company against any liability incurred by him in defending any proceedings whether civil or criminal, in which judgment is given favour or in which he is acquitted.

## **XX - SECRECY CLAUSE**

- 51. Every Director, (except institutional/ex-officio Director) auditor, trustee, member of a committee, officer, servant, agent, accountant or other person employed in the business of the Company shall, if so required by the Directors before entering upon his duties, sign a declaration pledging himself to observe strict secrecy respecting all transactions and affairs of the Company and all matters relating thereto, and shall by such declaration pledge himself not to reveal any of the matters which may come to his knowledge in the discharge of his duties except when required to do so by the Directors or by law or by the person to whom such matters relate and except so far as may be necessary in order to comply with any of the provisions of these Articles.
- 52. No member shall be entitled to visit or inspect any works of the Company, without the permission of the Directors or to require discovery of or any information respecting any details of the Company's trading or any matter which is or may be in the nature of a trade secret, mystery of trade, secret process or any other matter which may relate to the conduct of the business of the Company and which, in opinion of the Director, it would be inexpedient in the interest of the Company to disclose.

#### PART B

#### XXI - APPLICABILITY OF TABLE F

1. The regulations as contained in Tables F of Schedule I to the Companies Act, 2013, shall not apply to the Company except in so far as the same are repeated, contained or expressly made applicable in these Articles.

#### XXII - DEFINITION AND INTERPRETATION

# 2. **Definitions and Interpretation**

#### 2.1. **Definition.** In these Articles:

- (a) "Act" means the (Indian) Companies Act, 2013 the rules framed thereunder and any subsequent amendment or re-enactment thereof for the time being in force.
- (b) "Action Plan" means the plan made and mutually agreed between the Investor and the Promoters set out in schedule 5 of the Agreement, which sets out the specific environmental, social, labour, health and safety or security and environmental measures to be undertaken by the Company and the Subsidiaries, to enable the Business of the Company and/ or the Subsidiaries to be equipped, operated and undertaken in compliance with the Performance Standards and which shall be revised in accordance with Article 41.8 on an annual basis.
- (c) "Addendum Agreement" means the subscription cum addendum agreement dated 21 March, 2016 executed by and between the Investor, the Company and the Promoters.
- (d) "Affiliate" in relation to a Person:
  - (i) being a corporate entity, means any entity or Person, which controls, is controlled by, or is under the common control of such Person;
  - (ii) being an individual, means any entity or Person, which is controlled by such individual or a Relative of such individual;
  - (iii) in the case of the Investor, without prejudice to the generality of the foregoing, shall include its respective limited partners and any fund or investment vehicle owned, managed, advised, controlled or promoted by the Investor (as the case maybe) or by its respective Affiliates, investment managers or investment advisors;

For the avoidance of doubt, a Competitor shall, under no circumstances be capable of being classified as an Affiliate of any Party/Parties to any of the Transaction Documents.

- (e) "Agreed ESOP" shall have the meaning ascribed to it in Article 42.
- (f) "Agreement" or "Shareholders Agreement" means the shareholders agreement dated 07 August, 2015 executed by and between the Investor, the Company and the Promoters, as amended from time to time.
- (g) "Applicable S&E Law" means all applicable Laws and Consents of applicable Governmental Authorities setting standards/measures concerning environmental, social, labor, health and safety or security risks, including of the type specified in the Action Plan or imposing liability for the breach thereof..
- (h) "Articles" or "these Articles" means the Articles of Association of the Company, as amended from time to time.
- (i) "Associate Company" shall have the meaning ascribed to it in Section 2(6) of the Companies Act, 2013.
- (j) "Board" or the "Board of Directors" means the board of directors of the Company as constituted from time to time.
- (k) "Business" means the business of providing environmental engineering solutions with focus on wastewater treatment and recycling, desalination and industrial solid waste management.
- (l) "Business Day" means a day, not being a Saturday or a Sunday or a public holiday, on which banks are open for business in Mumbai, India and Mauritius, in the context of a payment being made to or from a bank in a place other than Mumbai, India or Mauritius, in such other place;
- (m) "Business Plan" shall have the meaning ascribed to it in Article 41.11.
- (n) "Closing" shall have the meaning ascribed to it in the Share Subscription Agreement.
- (o) "Closing Date" shall have the meaning ascribed to it in the Share Subscription Agreement.
- (p) "Company" means CONCORD ENVIRO SYSTEMS LIMITED.
- (q) "Competitor" means any Person (resident in any part of the world), other than a Financial Investor, who derives at least 20% of its revenues on a stand-alone or consolidated basis from a business which is the same as the Business of the Company and/ or the Subsidiaries, and where such revenues are equal to or more than 75% of the consolidated revenues of the Company and the Subsidiaries as reflected in the financial statements for Financial Year immediately preceding the date of determination.

- (r) "Consent" means any permit, permission, license, approval, authorization, consent, clearance, waiver, no objection certificate or other authorization of whatever nature and by whatever name called, which is required to be granted by any Person, including any Governmental Authority.
- (s) "Control" or "Controlling" shall include ownership or control (whether directly or indirectly) of more than 50% (fifty per cent) of the total voting securities (calculated on a Fully Diluted Basis), whether by shareholding or contract or otherwise or control of, or the power to control, policy decisions and/or the composition of the board of directors of the other entity.
- (t) "Conversion Shares" means the Equity Shares issued upon conversion of (i) Series A CCPS in the manner described in Articles 13 to 17; or (ii) Series A1 CCPS in the manner described in Articles 18 to 22, or (iii) Series A2 CCPS in the manner described in Article 23.4.
- (u) "Corporate Action Event" means any share split, bonus issue, stock dividend, consolidation, recapitalization or recombination affecting the Equity Securities and any other transaction having the effect of any of the foregoing.
- (v) "**Deed of Adherence**" means the deed of adherence in the form set forth in schedule 1 of the Agreement.
- (w) "Director" means a director on the Board from time to time.
- (x) "**Dividend**" includes bonus.
- (y) "**Draft Business Plan**" shall have the meaning ascribed to it in Article 41.11.
- (z) "Employees Stock Option Plan" or "ESOP" means any employee stock option plan (including the Agreed ESOP) as formulated by the Company and approved by the Board of the Company and applicable, inter alia, to the employees, including those in the Key Management Team of the Company and its Subsidiaries, if any, and to such other persons as are eligible, under applicable Law to receive such options.
- (aa) "Encumbrances" means any mortgage, pledge, equitable interest, assignment by way of security, hypothecation, claim, security interest, title defect, title retention agreement, voting trust agreement, option, lien, charge, restriction or limitation of any nature whatsoever (including any restriction or limitation imposed by way of court orders, interim awards, injunctions or any similar order or ruling issued by a Governmental Authority or other judicial / quasijudicial authority, including restriction on use, voting rights, transfer, receipt of income or exercise of any other attribute of ownership, right of set-off, any arrangement) the purpose of, or which has the effect of, granting security, or any other security interest of any kind whatsoever, or any agreement, whether conditional or otherwise.

- (bb) "Equity Securities" means Equity Shares, preference shares, including, but not limited to, Series A CCPS, Series A1 CCPS, Series A2 CCPS, any other security in the share capital of the Company, debentures or any note or debt security having or containing equity or profit participation features, or any option, warrant or other security or right which is directly or indirectly convertible into or exercisable or exchangeable for Equity Shares of the Company.
- (cc) "**Equity Shares**" means the equity shares in the issued, subscribed and paid up equity share capital of the Company.
- (dd) "Exempted Issuances" means any of the following: (i) any Equity Securities issued to the Company's officers, employees, directors and other service providers pursuant to any ESOP or any other stock option plan approved by the Board of the Company, including the Investor Nominee Director, if any; (ii) any Equity Securities actually issued upon the exercise of options or upon the conversion or exchange of convertible securities, including the Series A CCPS, Series A1 CCPS, Series A2 CCPS or warrants, in each case provided such issuance is pursuant to the terms of such option, convertible security or warrant; (iii) any Equity Securities issued pursuant to a transaction described in Article 17.8(b); (iv) any Equity Securities issued in an IPO; (v) any Equity Securities issued pursuant to a capital restructuring such as a merger, amalgamation, acquisition, etc.; and (vi) any Equity Securities issued to the Investor pursuant to the anti-dilution protection provisions set forth in Article 33.
- (ee) "Exit Period" means a period commencing on expiry of 60 (Sixty) months from the Closing Date and expiring after expiry of 84 (eighty four) months from the Closing Date.
- (ff) "Fair Market Value" means the equity valuation of the requisite Equity Securities determined by a Valuer and/ or a Third Valuer in accordance with the provisions of Section XXXIV of these Articles.
- (gg) "Financial Investor" means any asset management companies, private equity/venture capital entities (whether incorporated as limited liability partnerships, trusts or companies or otherwise), mutual funds, proprietary funds, financial institutions, banks (nationalized or otherwise and domestic or international) and foreign institutional investors who are normally engaged in the business of investing for purely financial returns, provided that where such Financial Investor is controlled by a Competitor or where such Financial Investor is in control of a Competitor, shall not be considered a Financial Investor.
- (hh) "**Financial Year**" means the financial year of the Company, commencing on April 1 every year and ending on March 31 of the following year, or such other

financial year of the Company as the Company may from time to time legally designate as its financial year.

- (ii) "Fully Diluted Basis" means that the calculation of shareholding is to be made assuming that (i) all Equity Securities (whether or not by their terms then currently convertible, exercisable or exchangeable), and all outstanding commitments to issue Equity Shares, membership or ownership interests, at a future date whether or not due to the occurrence of an event or otherwise, have been so converted, exercised or exchanged, (ii) that all unallocated options reserved for issuance under any ESOP have been issued and exercised and (iii) all partly paid Equity Shares (if any) have been fully paid-up.
- (jj) "Government" or "Governmental Authority" means any government, statutory authority, any department, agency or instrumentality of any government, any court, tribunal or arbitral tribunal, board and the governing body of any securities exchange or Stock Exchange(s).
- (kk) "Identified Terms" means the following:
  - (i) Lock-in (Article 30.1);
  - (ii) Failure of the Promoters to purchase the Offered Securities within the Purchase Period in accordance with the provisions of Article 30.5 hereof;
  - (iii) Pre-emptive right of the Investor (Article 24);
  - (iv) Anti-Dilution (Article 33) except where such right of the Investor is incapable of being exercised as a result of applicable Law;
  - (v) No solicitation (Article 41.2);
  - (vi) Non-compete (Article 41.3);
  - (vii) Anti-corruption (Article 41.7);
  - (viii) Transactions with Prohibited Persons (Article 41.9);
  - (ix) OFAC (Article 41.12);
  - (x) Greenhouse Gas Audit (Article 41.13); and
  - (xi) Reserved Matters as stated in Article 38.
- (ll) "INR" means Indian Rupees, the currency and legal tender of the Republic of India, for the time being in force.

- (mm) "Investment Amount" means the aggregate of Subscription Amount and Purchase Amount and such other amount as may be invested by the Investor in the Company from time to time in accordance with the provisions of the Agreement.
- (nn) "Investor" shall mean AF Holdings.
- (oo) "Investor Nominee Director" shall have the meaning ascribed to such term in Article 39.1.
- (pp) "**Investor Shares**" means the aggregate of Subscription Shares and the Purchase Shares, and such other Equity Securities as may be subscribed to / purchased by the Investor in the Company from time to time.
- (qq) "**IPO**" means the initial public offering and listing of the Equity Shares of the Company on any Stock Exchange(s)
- (rr) "IWWTC Share Purchase Agreement" means the share purchase agreement executed simultaneously with the Agreement by the Company, India Waste Water Treatment Company and the Investor.
- (ss) "Key Management Team" means the management team of the Company consisting of the managing director, whole time director, chief executive officer, chief financial officer, company secretary, head of production and head of technology at any present or future plants of the Company and the Subsidiaries, as appointed from time to time, including any persons discharging the roles and powers substantially similar to the aforesaid persons, notwithstanding their designations or any other function head or departmental head and any other employee whose cost to the Company exceeds INR 40,00,000 (Indian Rupees Forty Lakhs only) per annum.
- "Law" shall include all applicable statutes, enactments, acts of legislature, laws, ordinances, rules, bye-laws, regulations, guidelines, policies, directions, directives and orders of any Government, and applicable international treaties and regulations, in force at the relevant time.
- (uu) "Listing Date" means the date of listing of the Equity Shares on a recognized stock exchange.
- (vv) "Liquidation Event" means the commencement of any proceedings for the liquidation, dissolution or winding up of the Company either through members' or creditors' voluntary winding-up process or a court directed winding-up process.
- (ww) "Liquidity Event" means any of the following events:
  - (i) any merger, amalgamation, consolidation, reconstitution, restructuring or similar transaction with or into another Person following which the

Company's shareholders immediately post such transaction (or series of related transactions) hold less than 50% of the outstanding voting power of the Company on a Fully Diluted Basis or the surviving or acquiring entity; or

- (ii) a transaction or series of related transactions in which a Person or group of affiliated Persons (other than an underwriter of the Company's Equity Shares), acquires Equity Securities from Shareholders representing more than 50% (fifty percent) of the outstanding voting power of the Company; or
- (iii) sale, transfer or other disposition of assets and properties (including tangible and intangible assets) of the Company and/ or RSS where such assets and properties constitute more than 50% (fifty percent) of the value of all assets and properties (including tangible or intangible assets) of the Company and/ or RSS;

provided, however, that a transaction shall not constitute a Liquidity Event if its sole purpose is to change the jurisdiction of the Company's incorporation or to create a holding company that will be owned in substantially the same proportions by the Persons who held the Company's Shares immediately prior to such transaction.

Notwithstanding the foregoing, the treatment of any particular transaction or series of related transactions as a Liquidity Event may be waived by the vote or written consent of the Investor.

- (xx) "Management Reports" means the management internal reports summarizing progress against the annual budget, including (i) actual versus forecast financial results, (ii) actual versus forecast capital expenditure, and (iii) progress against business development targets, as well as noting any significant operational issues.
- (yy) "Memorandum" means the memorandum of association of the Company as originally framed or altered from time to time in accordance with the Agreement, the Subscription Agreement and the Act.
- (zz) "Month" means English calendar month.
- (aaa) "Office" means the Registered Office for time being of the Company.
- (bbb) "Offer for Sale" shall include any secondary offering by the Investor pursuant to or as part of an IPO and shall include the meaning assigned to it in the SEBI Regulations.
- (ccc) "**Performance Standards**" means International Finance Corporation's Performance Standards on Social and Environmental Sustainability, dated

- January 1, 2012, copies of which have been delivered to and receipt of which have been acknowledged by the Company and the Promoters.
- (ddd) "**Person**" shall include an individual, proprietorship, Hindu undivided family, partnership, corporation, company, unincorporated organization or association, trust or other entity, whether incorporated or not.
- (eee) "**Preference Holder**" shall have the meaning ascribed to it in Article 16.1.
- (fff) "**Promoters**" shall mean Mr. Prayas Goel and Mr. Prerak Goel.
- (ggg) "Promoter Share Purchase Agreement" means the share purchase agreement executed concurrently with the execution of the Agreement pursuant to which the Investor shall purchase from the Promoters Equity Shares as set forth therein.
- (hhh) "Proxy" includes Attorney duly constituted under a Power of Attorney.
- (iii) "Purchase Amount" means the aggregate consideration paid by the Investor under the IWWTC Share Purchase Agreement and the Promoter Share Purchase Agreement for the acquisition of the Purchase Shares.
- "Purchase Shares" means 11,056 (eleven thousand and fifty six) fully paidup Equity Shares of face value of INR 100 (Rupees One Hundred only) each purchased by the Investor from India Waster Water Treatment Company, and 695 (six hundred and ninety five) Equity Shares of face value of INR 100 (Rupees One Hundred only) each purchased by the Investor from Promoters in accordance with the terms of the IWWTC Share Purchase Agreement and the Promoter Share Purchase Agreement, respectively.
- (kkk) "QIPO" means an IPO (a) pursuant to which the Equity Shares of the Company are listed on any of the Stock Exchanges, and (b) which is managed and underwritten by one or more reputable investment banking firms of recognised standing in the market where the Equity Shares are to be offered, as agreed by the Investor;
- (III) **Relative**" shall have the meaning ascribed to such term in Section 2 (77) of the Act.
- (mmm) "**Related Party(ies)**" shall have the meaning ascribed to such term as defined in Section 2 (76) of the Act.
- (nnn) "Rules" means any rule(s) made pursuant to section 469 of the Act or such other provisions pursuant to which the Central Government is empowered to make rules, and shall include such rules as may be enacted, re-enacted or amended from time to time.
- (000) "Seal" means the Common Seal of Company.

- (ppp) "**SEBI**" shall mean the Securities and Exchange Board of India, a body established under the provisions of the Securities and Exchange Board of India Act, 1992.
- (qqq) "**SEBI Regulations**" means the Securities and Exchange Board of India (Issue of Capital and Disclosure Requirements) Regulations, 2018, as amended or reenacted from time to time, including any rules or circulars issued thereunder.
- (rrr) "Second Addendum Agreement" means the subscription cum addendum agreement dated 26 October, 2017 executed by and between the Investor, the Company and the Promoters.
- (sss) "Series A CCPS" means the Series A fully and compulsorily convertible, non-cumulative, preference shares of the Company, each having a face value of INR 1000 (Rupees One Thousand only), and carrying such terms and conditions set out in Articles 13 to 17 hereunder, in the Subscription Agreement and the Articles.
- (ttt) "Series A1 CCPS" means the Series A1 fully and compulsorily convertible, non-cumulative, preference shares of the Company, each having a face value of INR 1000 (Rupees One Thousand only), and carrying such terms and conditions as set out in Articles 18 to 22 hereunder and in the Addendum Agreement.
- (uuu) "Series A2 CCPS" means the Series A2 fully and compulsorily convertible, non-cumulative, preference shares of the Company, each having a face value of INR 1000 (Rupees One Thousand only), and carrying such terms and conditions as set out in Article 23 and in the Second Addendum Agreement.
- (vvv) "Series A Original Issue Price" shall have the meaning as set out in the Agreement.
- (www) "Series A1 Original Issue Price" shall have the meaning as set out in the Addendum Agreement.
- (xxx) "Series A2 Original Issue Price" has the meaning as set out in the Second Addendum Agreement.
- (yyy) "**Shareholder**" means the duly registered holder from time to time of the Equity Securities of the Company.
- (zzz) "**Stock Exchange**" means the National Stock Exchange, the BSE Limited or such other stock exchange approved by the Board.
- (aaaa) "Subscription Amount" shall have the meaning as set out in the Agreement.

- (bbbb) "Subscription Agreement" or "SSA" means the share subscription agreement executed concurrently with the execution of the Agreement pursuant to which the Company shall issue and allot such number of Series A CCPS to the Investor as set forth therein.
- (cccc) "Subsidiaries" means RSS, CBT, Reva, CES-FZE, BWTT and CES Mexico and shall include any future subsidiaries of the Company.
- (dddd) "Tax" or "Taxation" means, to the extent applicable, all forms of taxation, duties, levies imposts and social security charges, present and future, whether direct or indirect including without limitation corporate income tax, wage withholding tax, value added tax, customs and excise duties, capital gains tax, dividend withholding tax, dividend distribution tax, land taxes, environmental taxes and duties and any other type of taxes or duties payable by virtue of any applicable national, regional or local Law or regulation and levied by any Governmental Authority; together with any interest, penalties, surcharges or fines relating to them, due, payable, levied, imposed upon or claimed to be owed in any relevant jurisdiction.
- (eeee) "**Third Party**" means any Person other than the Parties to the Agreement or any Affiliates of any Parties to this Agreement.
- (ffff) "**Third Valuer**" shall have the meaning ascribed to the term in Article 82 below.
- (gggg) "**Tranche I Investment**" shall have the meaning ascribed to it in the Share Subscription Agreement.
- (hhhh) "**Tranche II Closing Date**" shall have the meaning ascribed to it in the Addendum Agreement.
- (iiii) "**Tranche III Closing Date**" has the meaning ascribed to it in the Second Addendum Agreement.
- (jjjj) "Transaction Documents" means and includes the Agreement, the Subscription Agreement, the disclosure letter, the Addendum Agreement, the Second Addendum Agreement, the IWWTC Share Purchase Agreement, the Promoter Share Purchase Agreement and these Articles.
- (kkkk) "Transfer" (including the terms "Transferred by", "Transferring" and "Transferability") means to directly or indirectly transfer, sell, assign, exchange, gift, dispose of in any manner, or subject to any Encumbrance, whether or not voluntarily, and whether by operation of law or otherwise.
- (IIII) "Valuer" means any Big 4 Accounting Firm or any merchant banker ranked amongst the top 10 merchant bankers in India by Thomson Reuters immediately prior to the date of appointment, and appointed in accordance with the provisions of Section XXXIV of these Articles.

# 2.2. Interpretation

- (a) "In writing and Written" includes printing, lithography and other modes of representing or reproducing words in a visible form.
- (b) Words importing the singular number include the plural number and vice versa.
- (c) Words importing the masculine gender include the feminine gender.
- (d) Reference to a Party shall include, such Party's legal heirs, executors, administrators, successors and permitted assigns and any Persons deriving title under it, as applicable.
- (e) References to the shareholding of any Shareholder in the Company shall (a) refer to the shareholding of such Shareholder computed on a Fully Diluted Basis, and (b) include the shareholding of such Shareholder's Affiliates holding any shares in the Company, if any.
- (f) The terms "hereto", "hereof," "herein," "hereby" and derivative or similar words refer to these Articles and not to any particular clause, article or section of the Articles.
- (g) The word "including" herein shall always mean "including, without limitation".
- (h) The phrases "ordinary course of business" and "ordinary course of business consistent with past practice" refer to the business and practice of the Company as conducted prior date of execution of the Agreement;
- (i) Whenever the Articles refer to a number of days, such number shall refer to calendar days, unless specifically provided otherwise.
- (j) All accounting terms used herein and not expressly defined herein shall have the meanings given to them under Indian GAAP.
- (k) Headings and captions are used for convenience only and shall not affect the interpretation of these Articles.
- (l) Unless the context otherwise requires, words and expression used in these presents shall bear the same meaning as in the Act, or any statutory modification thereof in force at the date when these regulation become binding on the Company.

# XXIII - PUBLIC COMPANY

## 3. **Public Company**

- 3.1. The Company is a public company within the meaning of Section 2(71) of the Companies Act, 2013 and accordingly:
  - "public company" means a company which-
  - c) is not a private company and
  - d) has a minimum paid-up share capital as may be prescribed

Provided that a company which is a subsidiary of a company, not being a private company, shall be deemed to be public company for the purposes of this Act even where such subsidiary company continues to be a private company in its articles.

3.2. The Company may at any time convert itself into a private company within the meaning of and subject to the provisions of the Act and Rules made thereunder.

#### XXIV - SHARE CAPITAL

# 4. Share Capital

- 4.1. **Authorised Capital.** The Authorised Share Capital of the Company is Rs. 42,50,00,000 (Indian Rupees Forty Two Crores Fifty Lakhs) divided into 4,00,00,000 (Four Crores) Equity Shares of INR 5 (Indian Rupees Five) each and 2,25,000 (Two Lakhs Twenty Five Thousand) Preference shares of INR 1000 (Indian Rupees Thousand) each. Subject to Article 38 (Reserved Matters), the Company shall have the power to alter its share capital to the extent permissible under the Act including to increase or reduce the capital, to divide the share in capital for the time being into several class and to attach thereto respectively such preferential, deferred, qualified or special rights, privileges or any condition as may be determined in such manner as may be permitted by the Act or by these Articles and to consolidate or sub-divide the shares and to vary, modify for abrogate such rights, privileges, or conditions in such manner as may for the time being be permitted by the Act or these Articles. The Board shall have power to classify or reclassify the shares into different types.
- 4.2. **Alteration of Capital.** Subject to Article 38 (Reserved Matters), the Company may from time to time change the structure of the capital, alter in any way the conditions of its memorandum relating to capital in accordance with the provisions of section 61 of the Companies Act, 2013 for all or any of the following purposes:
  - f. To increase its authorised share capital by such amount as it thinks expedient;
  - g. To consolidate and divide all or any of its share capital into shares of larger amount than its existing shares, provided that no consolidation and division which results in changes in the voting percentage of shareholders shall take effect unless it is approved by the Tribunal on an application made in the prescribed manner;
  - h. To convert all or any of its fully paid-up shares into stock, and reconvert that stock into fully paid up shares of any denomination;

- i. To sub-divide its shares or any of them into shares of smaller amount than is fixed by the Memorandum, so, however, that in the sub-division, the proportion between the amount paid and the amount, if any unpaid, on each reduced share shall be the same as it was in the case of the share from which the reduced share is derived; and
- j. To cancel any shares which at the date of the passing of the resolution, have not been taken or agreed to be taken by any persons and diminish the amount of its share capital by the amount of the shares so cancelled. Cancellation of shares in pursuance of this sub-clause shall not be deemed to be a reduction of the capital of the Company within the meaning of the Act.
- 4.3. **Kinds of Share Capital**. The Company may issue different kinds of Shares in accordance with these Articles and applicable provisions of the Act and Rules including shares with differential rights as to dividend, voting or otherwise in accordance with the Act and the Rules.
- 4.4. **Reduction.** The Company may (subject to the provisions of Sections 55 of the Act) from time to time, by Special Resolution, reduce in any manner and with, and subject to, any incident authorised and consent required by law:
  - (a) Its share capital;
  - (b) Any capital redemption reserve account; or
  - (c) Any share premium account.
- 4.5. **Variation of Members rights.** If at any time the share capital of the Company is divided into different classes of shares, the rights attached to any class (unless otherwise provided by the term of issue of the shares of that class) may, subject to the provisions of the Act and the applicable Rules, and whether or not the Company is being wound up, be varied with the consent in writing of the holders of not less than <sup>3</sup>/<sub>4</sub> (three-fourth) of the issued shares of that class or with the sanction of a Special Resolution passed at a separate meeting of the holders of the shares of that class.

# XXV - ISSUE AND ALLOTMENT OF SECURITIES

## 5. Further Issue of securities:

- (g) Where at any time, the Company proposes to increase its subscribed capital by the issue of further shares, such shares shall be offered—
  - (iv) to persons who, at the date of the offer, are holders of Equity Shares of the Company in proportion, as nearly as circumstances admit, to the paid up share capital on those shares by sending a letter of offer subject to the following conditions, namely:
    - d. the offer shall be made by notice specifying the number of shares offered and limiting a time not being less than 15 (fifteen) days, or such lesser number of days as may be prescribed, and not exceeding 30 (thirty) days from the date of the offer within which the offer, if not accepted, shall be deemed to have been declined;

- e. the aforesaid offer be deemed to include a right exercisable by the Person concerned to renounce the shares offered to him or any of them in favour of any other Person; and the notice referred to in clause a. above shall contain a statement of this right;
- f. after the expiry of the time specified in the notice aforesaid, or on receipt of earlier intimation from the Person to whom such notice is given that he declines to accept the shares offered, the Board may dispose of them in such manner which is not disadvantageous to the Shareholders and the Company;
- (v) to employees under a scheme of employees' stock option, subject to special resolution passed by the Company and subject to the Rules and such other conditions, as may be prescribed under Law; or
- (vi) to any Persons, if it is authorised by a special resolution, whether or not those Persons include the Persons referred to in clause (i) or clause (ii) above, either for cash or for a consideration other than cash, at such price as may be determined in accordance with Law, subject to the compliance with the applicable provisions of the Act and any other conditions as may be prescribed under Law.
- (h) The notice referred to in (a) (i) above shall be dispatched through registered post or speed post or through electronic mode to all the existing Shareholders at least 3 (three) days before the opening of the issue.
- (i) Nothing in this Article shall apply to the increase of the subscribed capital of a Company caused by the exercise of an option as a term attached to the Debentures issued or loan raised by the Company to convert such Debentures or loans into shares in the Company:
  - Provided that the terms of issue of such debentures or loan containing such an option have been approved before the issue of such debentures or the raising of loan by a special resolution passed by the Company in a general meeting.
- (j) Notwithstanding anything contained (c) above, where any debentures have been issued or loan has been obtained from any Government by the Company, and if that Government considers it necessary in the public interest so to do, it may, by order, direct that such debentures or loans or any part thereof shall be converted into shares in the Company on such terms and conditions as appear to the Government to be reasonable in the circumstances of the case even if terms of the issue of such debentures or the raising of such loans do not include a term for providing for an option for such conversion.

Provided that where the terms and conditions of such conversion are not acceptable to the Company, it may, within sixty days from the date of communication of such order, appeal to the competent authority which shall after hearing the company and the Government pass such order as it deems fit.

- (k) Where the Government has, by an order made under sub-clause (d), directed that any debenture or loan or any part thereof shall be converted into shares in the Company and where no appeal has been preferred to the competent authority under (d) above or where such appeal has been dismissed, the Memorandum of Association of the Company shall, where such order has the effect of increasing the authorized Share Capital of the Company, be altered and the authorized share capital of the Company shall stand increased by an amount equal to the amount of the value of shares which such debentures or loans or part thereof has been converted into.
- (l) The provisions contained in this Article shall be subject to the provisions of Section 42 and Section 62 of the Act, the Rules and the applicable provisions of the Act.

- 6. **Private Placement of Securities.** The Company may in accordance with provisions of the Act and the applicable Rules, issue further securities by way of a private placement of its securities.
- 7. **Power to Issue Redeemable Preference Shares.** Subject to and in accordance with the provisions of the Act and the Companies (Share Capital and Debentures) Rules, 2014, and subject to the provisions on which any Shares may have been issued, the Company shall have the power to issue preference Shares which are, or at the option of Company, liable to be redeemed within a period not exceeding 20 (twenty) years from the date of their issue, and the resolution authorising such issue shall prescribe the manner, terms and conditions of redemption.
- 8. Subject to the provisions of these Articles and Article 38 (Reserved Matters) and Article 24 (Pre-emptive Rights), the shares shall be under the control of the Directors who may allot or otherwise dispose of the same to such person on such term and conditions and at such time as Directors think fit.
- 9. **Bonus Shares**: The Company is hereby permitted to issue fully paid-up bonus shares to its members in the manner as set out in Section 63 of the Act and Companies (Share Capital and Debentures) Rules, 2014. Bonus shares may be issued in lieu of dividends.
- 10. **Buy-back of shares:** The Company is hereby permitted to purchase its own Shares or other specified Securities in accordance with the provision of Sections 68, 69 and 70 of the Companies Act, 2013 and other applicable provisions and applicable Rules. If and to the extent permitted by Law, the Company shall also have the power to re-issue the shares or securities so bought back.
- 11. The Company may pay commission (whether absolute or conditional) to any person in connection with the subscription or procurement of subscription to its Securities, as may be determined by the Board.

## 12. Shares at the Disposal of the Board of Directors

(b) Subject to the provisions of Section 62 and other applicable provisions of the Act, and these Articles, the shares in the Capital of the Company for the time being (including any shares forming part of any increased Capital of the Company) shall be under the control of the Board who may issue, allot or otherwise dispose of the same or any of them to Persons in such proportion and on such terms and conditions and either at a premium or at par or at discount (subject to compliance with Section 53 of the Act) at such time as they may, from time to time, think fit, to give to any person or persons the option or right to call for any shares either at par or premium or at a discount subject to the provisions of the Act during such time and for such consideration as the Directors think fit, and may issue and allot Shares in the capital of the Company on payment in full or part of any property sold and transferred or for any services rendered to the Company in the conduct of its business and any shares which may be so allotted may be issued as fully paid up shares and if so issued, shall be deemed to be fully paid up shares. Provided that option or right to call shares shall not be given to any Person or Persons without the sanction of the Company in the Shareholders' Meeting.

#### TERMS OF SERIES A CCPS

#### 13. Status of CCPS

Unless specifically agreed to by the Preference Holders, the Series A CCPS shall rank senior to the Equity Shares and other preference shares at all times and in all events.

# 14. Voting Rights

The voting rights in respect of the Series A CCPS shall be governed by Article 37.7.

### 15. **Term**

In the event that, for any reason whatsoever, the conversion of the Series A CCPS does not occur in the manner set forth herein, each Series A CCPS shall have a term of 20 (twenty) years.

#### 16. **Dividends**

- 16.1. Subject to applicable Law, each holder of Series A CCPS (a "**Preference Holder**") shall be entitled to receive a dividend at the rate of 0.001% of the face value per annum on each Series A CCPS held by such holder, payable when, as and if declared by the Board of the Company. Except to any other Equity Security issued in future and which ranks higher in priority than the Series A CCPS, no dividend or distribution may be paid to, or set aside for any other Shareholder unless dividend hereunder is paid to the holders of Series A CCPS.
- 16.2. Further, subject to applicable Law, after the payment of the dividend specified above, any additional dividends shall be distributed to all Shareholders in proportion to the number of Equity Shares held by each Shareholder on a Fully Diluted Basis (which for the avoidance of doubt, includes the Investor assuming conversion of the Series A CCPS).
- 16.3. No dividend or distribution shall be paid on any share of any class or series of the Company if and to the extent that as a consequence of such dividend or distribution holders of Series A CCPS would be entitled to a dividend hereunder greater than the maximum amount permitted to be paid in respect of preference shares of an Indian company held by a non-resident under applicable Laws (including without limitation, the Foreign Exchange Management (Transfer or Issue of Security by a Person Resident Outside India), Regulations, 2000).

#### 17. Conversion of the Series A CCPS.

17.1. Conversion Right. Each Preference Holder shall have the right to require the Company to convert all or a part of such Series A CCPS held by them into such number of fully paid Equity Shares equal to the Series A Original Issue Price divided by the Series A Conversion Price (as defined below) then in effect (the conversion ratio for the Series A CCPS is referred to herein as the "Conversion Ratio"), in accordance with the terms of the Agreement and Articles (the "Conversion Right") at any time before the

- conversion of the Series A CCPS is carried out under Article 17.2 below. The Conversion Ratio shall initially be 1:1, until adjusted in accordance with Article 17.8 and Article 33.2 (*Broad Based Anti-Dilution Protection*).
- 17.2. <u>Mandatory Conversion</u>. The Company shall mandatorily convert each Series A CCPS into Equity Shares at the Conversion Ratio then in effect upon the earlier of (i) immediately prior to the filing of a red herring prospectus in relation to any IPO (approved by the Investor) or any proposed QIPO, or (ii) the issuance of a Conversion Notice in terms of Article 17.3 below, or (iii) transfer of Offered Securities by the Investor to a Competitor in terms of Article 30.5 below. Notwithstanding the foregoing, each Series A CCPS shall mandatorily convert into Equity Shares at the Conversion Ratio then in effect one day prior to the expiry of the twentieth anniversary of the date on which the Series A CCPS were first issued by the Company.
- 17.3. Exercise of Conversion Right and procedure for conversion. A Preference Holder may exercise the Conversion Right by (a) delivering a written notice (a "Conversion Notice") to the Company of its intention to so convert Series A CCPS held by such Preference Holder into Equity Shares and (b) surrendering the relevant share certificates representing such Series A CCPS at the office of the Company together with the Conversion Notice. If the Equity Shares are to be issued in dematerialized form, the Preference Holder shall also provide standard information required to allow the Company to issue such shares in dematerialized form. The Conversion Notice shall specify the number of Series A CCPS that such Preference Holder elects to convert (such Series A CCPS referred to as the "Relevant CCPS") and state therein the name or names of any nominee for such Preference Holder in which the certificate or certificates for Equity Shares are to be issued and whether or not such Equity Shares are to be issued in physical form or dematerialized form.
- 17.4. As soon as reasonably practicable, but in no event later than 7 (seven) Business Days from the date of the Conversion Notice, the Company shall take all necessary corporate actions and obtain all necessary Consents and issue the appropriate number of Equity Shares into which the Relevant CCPS are convertible at the Conversion Ratio then in effect. Not later than the 7<sup>th</sup> (seventh) Business Day from the date of the Conversion Notice, the Company shall deliver to such Preference Holder:
  - (a) duly stamped and executed share certificates with respect to the Conversion Shares issued on conversion of the Relevant CCPS;
  - (b) certified true copies of all filings necessary to effect and validate the issue of the Conversion Shares, including Form PAS-3;
  - (c) certified true copy of the register of members of the Company showing the Preference Holder as the registered owner of the Conversion Shares;
  - (d) duly acknowledged copies of all intimations and filings made by the Company with the RBI in respect of the conversion of the Series A CCPS and issue of Equity Shares in lieu thereof; and

- (e) in the event that the Preference Holder has requested in the Conversion Notice or otherwise in writing to the Company and has provided all required information to the Company to hold the converted Equity Shares in a dematerialized form, evidence that such Equity Shares have been deposited in the account of such Preference Holder or to the nominee or nominees of such Preference Holder previously identified to the Company in writing accompanied with a beneficiary position statement issued by the concerned depository.
- 17.5. Procedure for Mandatory Conversion. In the case of a mandatory conversion of Series A CCPS pursuant to Article 17.2, the Company shall take all necessary corporate and other actions and obtain all Consents on or prior to the date of conversion, and shall provide the documents/ information listed in Articles 17.4(a) to 17.4(e), to the Preference Holders on the date of conversion of the Series A CCPS.
- 17.6. No Fractional Shares. No fractional Conversion Shares shall be issued upon conversion of Series A CCPS. If the computation of the number of Conversion Shares to be issued, results in a fraction, then the number of Conversion Shares shall be rounded off to the nearest whole number.
- 17.7. Conversion Price. The "Series A Conversion Price" for the Series A CCPS shall initially be equal to the Series A Original Issue Price and, subject to applicable Law, shall be adjusted in accordance with Article 17.8 and Article 33.2 (*Broad Based Anti-Dilution Protection*). It is clarified that any adjustment of the Series A Conversion Price and Conversion Ratio shall not automatically result in conversion of the Series A CCPS. In the event of any adjustment to the Series A Conversion Price and the Conversion Ratio of the Series A CCPS, the Company, at its expense, shall promptly compute such adjustment and inform the holders of Series A CCPS of the details of such adjustment in writing. The Company shall upon the written request at any time of any Preference Holder furnish or cause to be furnished to such holder a certificate setting forth (A) such adjustment, (B) the Series A Conversion Price and Conversion Ratio at the time in effect, and (C) the number of Equity Shares and the amount, if any, that at the time would be received upon the conversion of Series A CCPS.
- 17.8. Conversion Price and Conversion Ratio Adjustments for Certain Dilutive Issuances, Splits and Combinations. The Series A Conversion Price and Conversion Ratio shall be subject to adjustment from time to time as follows:
  - (a) Adjustments for Dilutive Issuances. If at any time after the Closing Date, the Company issues to any Person any Equity Securities (other than pursuant to an Exempted Issuance), at a price per security that is lower than the Series A Conversion Price in effect immediately prior to such issuance (a "Dilutive Issuance"), then the holders of Series A CCPS shall be entitled to a broad-based weighted average anti-dilution protection in accordance with Article 33.2. In such an event, the Company and the other Shareholders shall be bound to cooperate with the holders of Series A CCPS and the Company such that, the Company forthwith takes all necessary steps to adjust the Series A Conversion Price and Conversion Ratio in accordance with Article 33.2.

Notwithstanding the foregoing, if the adjustment set forth in this Article 17.8(a) is not permitted to be made, in whole or in part, under applicable Law, the Parties agree to take all necessary acts to put the Preference Holders in the position that they would have been if such adjustment to the Series A Conversion Price had been made, including potentially the issuance of new Equity Shares to the Preference Holders, or an Affiliate or designated nominee of a Preference Holder, whereby the Preference Holders, or such Affiliate or designated nominee thereof, are not required to pay any additional amounts for the issuance of such new Equity Shares, if permitted by applicable Law.

- In the event the Company should at any time or from time to time after the (b) Closing Date fix a record date for the effectuation of a split or subdivision of the outstanding Equity Shares or the determination of holders of Equity Shares entitled to receive a dividend or other distribution payable in additional Equity Shares or other securities or rights convertible into, or entitling the holder thereof to receive directly or indirectly, additional Equity Shares (hereinafter referred to as "Equity Share Equivalents") without payment of any consideration by such holder for the additional Equity Shares or the Equity Share Equivalents (including the additional Equity Shares issuable upon conversion or exercise thereof), then, as of such record date (or the date of such dividend distribution, split or subdivision if no record date is fixed), the Series A Conversion Price and Conversion Ratio shall be appropriately decreased so that the number of Equity Shares issuable on conversion of each Series A CCPS shall be increased in proportion to such increase of the aggregate of Equity Shares outstanding and those issuable with respect to such Equity Share Equivalents.
- (c) If the number of Equity Shares outstanding at any time after the Closing Date is decreased by a combination or consolidation (reverse stock split) of the outstanding Equity Shares, then, following the record date of such combination or consolidation (reverse stock split), the Series A Conversion Price and Conversion Ratio shall be appropriately increased so that the number of Equity Shares issuable on conversion of each Series A CCPS shall be decreased in proportion to such decrease in outstanding Equity Shares.
- (d) Subject to the provisions of Clause 12.3 of the SSA, the Conversion Ratio of the Series A CCPS may be adjusted at the option of the Indemnified Party (as defined in the Share Subscription Agreement), on the basis of the Fair Market Value, such that the number of Conversion Shares increases to make good the Indemnity Amount (as defined hereinafter) payable to the Indemnified Party ("Indemnity Conversion Ratio"). Provided however that, in the event the Indemnifying Party pays the Indemnity Amount or part thereof, to the Indemnified Party, the Indemnity Conversion Ratio shall be readjusted such that the number of Conversion Shares is reduced to the extent of the Indemnity Amount (or part thereof), paid to the Indemnified Party. Provided further that, at the time of exit of the Investor in accordance with the terms of the Agreement and the Articles, if the value received by the Investor as a consequence of the exit ("Exit Value") and in respect of the Conversion Shares

(reckoned as a result of the adjustment of the Conversion Ratio) is greater than the Indemnity Amount, the Investor shall transfer to the Promoters, such number of Investor Shares representing the difference between the Exit Value of the Conversion Shares and the Indemnity Amount. Any Tax liability associated with any such adjustment shall be borne by the Investor.

- (e) The Conversion Ratio of the Series A CCPS shall also be adjusted at the time of allocation / issuance of any Equity Shares pursuant to the Agreed ESOP by the Company, the intention being that upon allocation / issuance of Equity Securities under the Agreed ESOP, the Investor shall not be diluted and the Conversion Shares to be issued at the time of conversion are increased to the extent of Equity Securities allocated / issued under the Agreed ESOP.
- (f) The Conversion Ratio of the Series A CCPS shall be adjusted in respect of amounts receivable by CES-FZE, in accordance with the provisions of clause 3.5.8 (f) of the Agreement.
- (g) The Conversion Ratio of the Series A CCPS may be adjusted at the option of the holder of the Series A CCPS upon the occurrence of a Liquidity Event in accordance with the formula and illustration thereof as stated below:

Conversion Ratio =

[(Total no. of Equity Shares of the Company) \* Investor's Proposed Shareholding – Existing Equity Shares held by the Investor) / (1 – Investor's Proposed Shareholding)

No. of Series A CCPS held by the Investor

Where,

Investor's Proposed Shareholding = (Investment Amount \* 2) / Fair Market Value of the Company as on the date of conversion

(h) The Conversion Ratio of the Series A CCPS may be adjusted at the option of the holder of the Series A CCPS upon the occurrence of a Liquidation Event in accordance with the formula and illustration thereof as stated below:

Conversion Ratio =

((Total no. of Equity Shares of the Company) \* Investor's Proposed Shareholding – Existing Equity Shares held by the Investor) / (1 – Investor's Proposed Shareholding)

No. of Series A CCPS held by the Investor

Where,

Investor's Proposed Shareholding = (Investment Amount) / Total liquidation equity valuation of the Company as on the date of conversion

- (i) The adjustments under this Article 17.8 shall not be mutually exclusive.
- 17.9. <u>Conversion Cost</u>. The Company shall bear all expenses arising from the conversion of the Series A CCPS as set out in this Article 17, including *inter alia*, any stamp duty applicable on the issuance of share certificates subsequent to conversion of the Series A CCPS.
- 17.10. Other Distributions. In the event the Company declares a distribution payable in securities of other Persons, evidences of indebtedness issued by the Company or other Persons, assets (excluding cash dividends) or options or rights not referred to in Article 17.8(a) above, then, in each such case for the purpose of this Article 17.10, the Preference Holders shall be entitled to a proportionate share of any such distribution as though they were the holders of the number Equity Shares into which their Series A CCPS are convertible as of the record date fixed for the determination of the holders of Equity Shares entitled to receive such distribution.
- 17.11. Reservation of Equity Shares Issuable Upon Conversion. The Company shall at all times reserve and keep available out of its authorized but unissued Equity Shares, solely for the purpose of effecting the conversion of the Series A CCPS, such number of its Equity Shares as shall from time to time be sufficient to effect the conversion of all outstanding Series A CCPS; and if at any time the number of authorized but unissued Equity Shares shall not be sufficient to effect the conversion of all then outstanding Series A CCPS, in addition to such other remedies as shall be available to the Preference Holder, the Company shall take such corporate action as may be necessary to increase its authorized but unissued Equity Shares to such number of shares as shall be sufficient for such purposes, including, without limitation, to obtain the requisite shareholder approval of any necessary amendment to the Articles.
- 17.12. Waiver of Adjustment to Conversion Price. Notwithstanding anything herein to the contrary, any downward adjustment of the Series A Conversion Price may be waived, either prospectively or retroactively and either generally or in a particular instance, by the written consent or vote of the Investor. Any such waiver shall bind all future holders of Series A CCPS.

#### XXVI - TERMS OF SERIES A1 CCPS

## 18. **Status of CCPS**

Unless specifically agreed to by the A1 Preference Holder in writing, the Series A1 CCPS shall rank senior to the Equity Shares and other preference shares at all times and in all events. It is hereby clarified that Series A1 CCPS shall rank *pari passu* with Series A CCPS at all times and in all events.

#### 19. **Voting Rights**

The voting rights in respect of the Series A1 CCPS shall be governed by Article 37.7, as if each time the term "Series A CCPS" has been used in such Article 37.7 is replaced by the term "Series A1 CCPS".

# 20. **Term**

In the event that, for any reason whatsoever, the conversion of the Series A1 CCPS does not occur in the manner set forth herein, each Series A1 CCPS shall have a term of twenty (20) years.

# 21. **Dividends**

- 21.1. Subject to applicable Law, each holder of Series A1 CCPS (a "A1 Preference Holder") shall be entitled to receive a dividend at the rate of 0.001% of the face value per annum on each Series A1 CCPS held by such holder, payable when, as and if declared by the Board of the Company. Except to any other Equity Security issued in future and which ranks higher in priority than the Series A1 CCPS, no dividend or distribution may be paid to, or set aside for any other Shareholder unless dividend hereunder is paid to the holders of Series A1 CCPS.
- 21.2. Further, subject to applicable Law, after the payment of the dividend specified above, any additional dividends shall be distributed to all Shareholders in proportion to the number of Equity Shares held by each Shareholder on a Fully Diluted Basis (which for the avoidance of doubt, includes the Investor assuming conversion of the Series A1 CCPS).
- 21.3. No dividend or distribution shall be paid on any share of any class or series of the Company if and to the extent that as a consequence of such dividend or distribution holders of Series A1 CCPS would be entitled to a dividend hereunder greater than the maximum amount permitted to be paid in respect of preference shares of an Indian company held by a non-resident under applicable Laws (including without limitation, the Foreign Exchange Management (Transfer or Issue of Security by a Person Resident Outside India), Regulations, 2000).

# 22. Conversion of the Series A1 CCPS.

- 22.1. Conversion Right. Each A1 Preference Holder shall have the right to require the Company to convert all or a part of such Series A1 CCPS held by them into such number of fully paid Equity Shares equal to the Series A1 Original Issue Price divided by the Series A1 Conversion Price (as defined below) then in effect (the conversion ratio for the Series A1 CCPS is referred to herein as the "A1 Conversion Ratio"), in accordance with the terms of the Addendum Agreement and Articles (the "A1 Conversion Right") at any time before the conversion of the Series A1 CCPS is carried out under Article 22.2 below. The A1 Conversion Ratio shall initially be 1:1, until adjusted in accordance with Article 22.8 and Article 33.2 (Broad Based Anti-Dilution Protection).
- 22.2. <u>Mandatory Conversion</u>. The Company shall mandatorily convert each Series A1 CCPS into Equity Shares at the A1 Conversion Ratio then in effect upon the earlier of (i) immediately prior to the filing of a red herring prospectus in relation to any IPO (approved by the Investor) or any proposed QIPO, or (ii) the issuance of an A1

Conversion Notice in terms of Article 22.3 below or (iii) transfer of Offered Securities by the Investor to a Competitor in terms of Article 30.5 below. Notwithstanding the foregoing, each Series A1 CCPS shall mandatorily convert into Equity Shares at the A1 Conversion Ratio then in effect one day prior to the expiry of the twentieth anniversary of the date on which the Series A1 CCPS were first issued by the Company.

- 22.3. Exercise of A1 Conversion Right and procedure for conversion. An A1 Preference Holder may exercise the A1 Conversion Right by (a) delivering a written notice (a "A1 Conversion Notice") to the Company of its intention to so convert Series A1 CCPS held by such A1 Preference Holder into Equity Shares and (b) surrendering the relevant share certificates representing such Series A1 CCPS at the office of the Company together with the A1 Conversion Notice. If the Equity Shares are to be issued in dematerialized form, the A1 Preference Holder shall also provide standard information required to allow the Company to issue such shares in dematerialized form. The A1 Conversion Notice shall specify the number of Series A1 CCPS that such A1 Preference Holder elects to convert (such Series A1 CCPS referred to as the "Relevant A1 CCPS") and state therein the name or names of any nominee for such A1 Preference Holder in which the certificate or certificates for Equity Shares are to be issued and whether or not such Equity Shares are to be issued in physical form or dematerialized form.
- 22.4. As soon as reasonably practicable, but in no event later than 7 (seven) Business Days from the date of the A1 Conversion Notice, the Company shall take all necessary corporate actions and obtain all necessary Consents and issue the appropriate number of Equity Shares into which the Relevant A1 CCPS are convertible at the A1 Conversion Ratio then in effect. Not later than the 7<sup>th</sup> (seventh) Business Day from the date of the A1 Conversion Notice, the Company shall deliver to such A1 Preference Holder:
  - (a) duly stamped and executed share certificates with respect to the A1 Conversion Shares issued on conversion of the Relevant A1 CCPS;
  - (b) certified true copies of all filings necessary to effect and validate the issue of the A1 Conversion Shares, including Form PAS-3;
  - (c) certified true copy of the register of members of the Company showing the A1 Preference Holder as the registered owner of the A1 Conversion Shares;
  - (d) duly acknowledged copies of all intimations and filings made by the Company with the RBI in respect of the conversion of the Series A1 CCPS and issue of Equity Shares in lieu thereof; and
  - (e) in the event that the A1 Preference Holder has requested in the A1 Conversion Notice or otherwise in writing to the Company and has provided all required information to the Company to hold the converted Equity Shares in a dematerialized form, evidence that such Equity Shares have been deposited in the account of such A1 Preference Holder or to the nominee or nominees of such A1 Preference Holder previously identified to the Company in writing

accompanied with a beneficiary position statement issued by the concerned depository.

- 22.5. <u>Procedure for Mandatory Conversion</u>. In the case of a mandatory conversion of Series A1 CCPS pursuant to Article 22.2, the Company shall take all necessary corporate and other actions and obtain all Consents on or prior to the date of conversion, and shall provide the documents/ information listed in Articles 22.4(a) to 22.4(e), to the A1 Preference Holder on the date of conversion of the Series A1 CCPS.
- 22.6. <u>No Fractional Shares</u>. No fractional A1 Conversion Shares shall be issued upon conversion of Series A1 CCPS. If the computation of the number of A1 Conversion Shares to be issued, results in a fraction, then the number of A1 Conversion Shares shall be rounded down to the nearest whole number.
- 22.7. Conversion Price. The "Series A1 Conversion Price" for the Series A1 CCPS shall initially be equal to the Series A1 Original Issue Price and, subject to applicable Law, shall be adjusted in accordance with Article 22.8 below and Article 33.2 (*Broad Based Anti-Dilution Protection*). It is clarified that any adjustment of the Series A1 Conversion Price and A1 Conversion Ratio shall not automatically result in conversion of the Series A1 CCPS. In the event of any adjustment to the Series A1 Conversion Price and the A1 Conversion Ratio of the Series A1 CCPS, the Company, at its expense, shall promptly compute such adjustment and inform the holders of Series A1 CCPS of the details of such adjustment in writing. The Company shall upon the written request at any time of any A1 Preference Holder furnish or cause to be furnished to such holder a certificate setting forth (A) such adjustment, (B) the Series A1 Conversion Price and A1 Conversion Ratio at the time in effect, and (C) the number of Equity Shares and the amount, if any, of other property that at the time would be received upon the conversion of Series A1 CCPS.
- 22.8. Conversion Price and Conversion Ratio Adjustments for Certain Dilutive Issuances, Splits and Combinations. The Series A1 Conversion Price and A1 Conversion Ratio shall be subject to adjustment from time to time as follows:
  - Adjustments for Dilutive Issuances. If at any time after the Tranche II Closing (a) Date, the Company issues to any Person any Equity Securities (other than pursuant to an Exempted Issuance), at a price per security that is lower than the Series A1 Conversion Price in effect immediately prior to such issuance (a "A1 Dilutive Issuance"), then the holders of Series A1 CCPS shall be entitled to a broad-based weighted average anti-dilution protection in accordance with Article 33.2. In such an event, the Company and the other Shareholders shall be bound to cooperate with the holders of Series A1 CCPS and the Company such that, the Company forthwith takes all necessary steps to adjust the Series A1 Conversion Price and A1 Conversion Ratio in accordance with Article33.2. Notwithstanding the foregoing, if the adjustment set forth in this Article 22.8(a) is not permitted to be made, in whole or in part, under applicable Law, the Parties agree to take all necessary acts to put the A1 Preference Holder in the position that they would have been if such adjustment to the Series A1 Conversion Price had been made, including potentially the issuance of new

Equity Shares to the A1 Preference Holder, or an Affiliate or designated nominee of a A1 Preference Holder, whereby the A1 Preference Holder, or such Affiliate or designated nominee thereof, are not required to pay any additional amounts for the issuance of such new Equity Shares, if permitted by applicable Law.

- In the event the Company should at any time or from time to time after the (b) Tranche II Closing Date fix a record date for the effectuation of a split or subdivision of the outstanding Equity Shares or the determination of holders of Equity Shares entitled to receive a dividend or other distribution payable in additional Equity Shares or other securities or rights convertible into, or entitling the holder thereof to receive directly or indirectly, additional Equity Shares (hereinafter referred to as "A1 Equity Share Equivalents") without payment of any consideration by such holder for the additional Equity Shares or the A1 Equity Share Equivalents (including the additional Equity Shares issuable upon conversion or exercise thereof), then, as of such record date (or the date of such dividend distribution, split or subdivision if no record date is fixed), the Series A1 Conversion Price and A1 Conversion Ratio shall be appropriately decreased so that the number of Equity Shares issuable on conversion of each Series A1 CCPS shall be increased in proportion to such increase of the aggregate of Equity Shares outstanding and those issuable with respect to such A1 Equity Share Equivalents.
- (c) If the number of Equity Shares outstanding at any time after the Tranche II Closing Date is decreased by a combination or consolidation (reverse stock split) of the outstanding Equity Shares, then, following the record date of such combination or consolidation (reverse stock split), the Series A1 Conversion Price and A1 Conversion Ratio shall be appropriately increased so that the number of Equity Shares issuable on conversion of each Series A1 CCPS shall be decreased in proportion to such decrease in outstanding Equity Shares.
- (d) Subject to the provisions of clause 12.3 of the SSA, the A1 Conversion Ratio of the Series A1 CCPS may be adjusted at the option of the Indemnified Party, on the basis of the Fair Market Value, such that the number of A1 Conversion Shares increases to make good the Indemnity Amount (as defined in clause 12.2 of the SSA) payable to the Indemnified Party ("A1 Indemnity Conversion Ratio"). Provided however that, in the event the Indemnifying Party pays the Indemnity Amount or part thereof, to the Indemnified Party, the A1 Indemnity Conversion Ratio shall be readjusted such that the number of A1 Conversion Shares is reduced to the extent of the Indemnity Amount (or part thereof), paid to the Indemnified Party. Provided further that, at the time of exit of the Investor in accordance with the terms of the Agreement and the Articles, if the value received by the Investor as a consequence of the exit ("A1 Exit Value") and in respect of the A1 Conversion Shares (reckoned as a result of the adjustment of the A1 Conversion Ratio) is greater than the Indemnity Amount, the Investor shall transfer to the Promoters, such number of Investor Shares representing the difference between the A1 Exit Value of the A1

Conversion Shares and the Indemnity Amount. Any Tax liability associated with any such adjustment shall be borne by the Investor.

- (e) The A1 Conversion Ratio of the Series A1 CCPS shall also be adjusted at the time of allocation / issuance of any Equity Shares pursuant to the Agreed ESOP by the Company, the intention being that upon allocation / issuance of Equity Securities under the Agreed ESOP, the Investor shall not be diluted and the A1 Conversion Shares to be issued at the time of conversion are increased to the extent of Equity Securities allocated / issued under the Agreed ESOP.
- (f) The Conversion Ratio of the Series A1 CCPS shall be adjusted in respect of amounts receivable by CES-FZE, in accordance with the provisions of clause 3.5.8 (f) of the Agreement.
- (g) The A1 Conversion Ratio of the Series A1 CCPS may be adjusted at the option of the holder of the Series A1 CCPS upon the occurrence of a Liquidity Event in accordance with the formula and illustration thereof as stated below:

A1 Conversion Ratio =

[(Total no. of Equity Shares of the Company) \* Investor's Proposed Shareholding – Existing Equity Shares held by the Investor] / (1 – Investor's Proposed Shareholding)

No. of Series A1 CCPS held by the Investor

Where,

Investor's Proposed Shareholding = (Investment Amount \* 2) / Fair Market Value of the Company as on the date of conversion

(h) The A1 Conversion Ratio of the Series A1 CCPS may be adjusted at the option of the holder of the Series A1 CCPS upon the occurrence of a Liquidation Event in accordance with the formula and illustration thereof as stated below:

A1 Conversion Ratio =

[(Total no. of Equity Shares of the Company) \* Investor's Proposed Shareholding – Existing Equity Shares held by the Investor] / (1 – Investor's Proposed Shareholding)

No. of Series A1 CCPS held by the Investor

Where,

Investor's Proposed Shareholding = (Investment Amount) / Total liquidation equity valuation of the Company as on the date of conversion

(i) The adjustments under this Article 22.8 shall not be mutually exclusive.

- 22.9. <u>Conversion Cost</u>. The Company shall bear all expenses arising from the conversion of the Series A1 CCPS as set out in this Article 22, including *inter alia*, any stamp duty applicable on the issuance of share certificates subsequent to conversion of the Series A1 CCPS.
- 22.10. Other Distributions. In the event the Company declares a distribution payable in securities of other Persons, evidences of indebtedness issued by the Company or other Persons, assets (excluding cash dividends) or options or rights not referred to in Article 22.8(a) above, then, in each such case for the purpose of this Article 22.10, the A1 Preference Holders shall be entitled to a proportionate share of any such distribution as though they were the holders of the number Equity Shares into which their Series A1 CCPS are convertible as of the record date fixed for the determination of the holders of Equity Shares entitled to receive such distribution.
- 22.11. Reservation of Equity Shares Issuable Upon Conversion. The Company shall at all times reserve and keep available out of its authorized but unissued Equity Shares, solely for the purpose of effecting the conversion of the Series A1 CCPS, such number of its Equity Shares as shall from time to time be sufficient to effect the conversion of all outstanding Series A1 CCPS; and if at any time the number of authorized but unissued Equity Shares shall not be sufficient to effect the conversion of all then outstanding Series A1 CCPS, in addition to such other remedies as shall be available to the A1 Preference Holder, the Company shall take such corporate action as may be necessary to increase its authorized but unissued Equity Shares to such number of shares as shall be sufficient for such purposes, including, without limitation, to obtain the requisite shareholder approval of any necessary amendment to the Articles.
- 22.12. <u>Waiver of Adjustment to Conversion Price</u>. Notwithstanding anything herein to the contrary, any downward adjustment of the Series A1 Conversion Price may be waived, either prospectively or retroactively and either generally or in a particular instance, by the written consent or vote of the Investor. Any such waiver shall bind all future holders of Series A1 CCPS.

#### XXVII - TERMS OF SERIES A2 CCPS

# 23. Status of CCPS

Unless specifically agreed to by the A2 Preference Holder in writing, the Series A2 CCPS shall rank senior to the Equity Shares and other preference shares at all times and in all events. It is hereby clarified that Series A2 CCPS shall rank *pari passu* with Series A CCPS and the Series A1 CCPS at all times and in all events.

# 22.1 <u>Voting Rights</u>

The voting rights in respect of the Series A2 CCPS shall be governed by Article XXXVIII – 37.7, as if each time the term "Series A CCPS" has been used in such Article XXXVIII – 37.7 shall mean reference to the term "Series A2 CCPS" as well.

# 22.2 <u>Term</u>

In the event that, for any reason whatsoever, the conversion of the Series A2 CCPS does not occur in the manner set forth herein, each Series A2 CCPS shall have a term of twenty (20) years.

# 22.3 Dividends

- 22.3.1 Subject to applicable Law, each holder of Series A2 CCPS (a "A2 Preference Holder") shall be entitled to receive a dividend at the rate of 0.001% of the face value per annum on each Series A2 CCPS held by such holder, payable when, as and if declared by the Board of the Company. Except to any other Equity Security issued in future and which ranks higher in priority than the Series A2 CCPS, no dividend or distribution may be paid to, or set aside for any other Shareholder unless dividend hereunder is paid to the holders of Series A2 CCPS.
- 22.3.2 Further, subject to applicable Law, after the payment of the dividend specified above, any additional dividends shall be distributed to all Shareholders in proportion to the number of Equity Shares held by each Shareholder on a Fully Diluted Basis (which for the avoidance of doubt, includes the Investor assuming conversion of the Series A2 CCPS).
- 22.3.3 No dividend or distribution shall be paid on any share of any class or series of the Company if and to the extent that as a consequence of such dividend or distribution holders of Series A2 CCPS would be entitled to a dividend hereunder greater than the maximum amount permitted to be paid in respect of preference shares of an Indian company held by a non-resident under applicable Laws (including without limitation, the Foreign Exchange Management (Transfer or Issue of Security by a Person Resident Outside India), Regulations, 2000).

# 22.4 Conversion of the Series A2 CCPS.

- 22.4.1 Conversion Right. Each A2 Preference Holder shall have the right to require the Company to convert all or a part of such Series A2 CCPS held by them into such number of fully paid Equity Shares equal to the Series A2 Original Issue Price divided by the Series A2 Conversion Price (as defined below) then in effect (the conversion ratio for the Series A2 CCPS is referred to herein as the "A2 Conversion Ratio"), in accordance with the terms of these Articles (the "A2 Conversion Right") at any time before the conversion of the Series A2 CCPS is carried out under Article 22.4.2 below. The A2 Conversion Ratio shall initially be 1:1, until adjusted in accordance with Article 22.4.8 below and Article 33.
- 22.4.2 <u>Mandatory Conversion</u>. The Company shall mandatorily convert each Series A2 CCPS into Equity Shares at the A2 Conversion Ratio then in effect upon the earlier of (i) immediately prior to the filing of a red herring prospectus in relation to any IPO (approved by the Investor) or any proposed QIPO, or (ii) the

issuance of an A2 Conversion Notice in terms of Article 23.4.3 below or (iii) transfer of Offered Securities by the Investor to a Competitor in terms of Article 30.5 below. Notwithstanding the foregoing, each Series A2 CCPS shall mandatorily convert into Equity Shares at the A2 Conversion Ratio then in effect one day prior to the expiry of the twentieth anniversary of the date on which the Series A2 CCPS were first issued by the Company.

- Preference Holder may exercise the A2 Conversion Right by (a) delivering a written notice (a "A2 Conversion Notice") to the Company of its intention to so convert Series A2 CCPS held by such A2 Preference Holder into Equity Shares and (b) surrendering the relevant share certificates representing such Series A2 CCPS at the office of the Company together with the A2 Conversion Notice. If the Equity Shares are to be issued in dematerialized form, the A2 Preference Holder shall also provide standard information required to allow the Company to issue such shares in dematerialized form. The A2 Conversion Notice shall specify the number of Series A2 CCPS that such A2 Preference Holder elects to convert (such Series A2 CCPS referred to as the "Relevant A2 CCPS") and state therein the name or names of any nominee for such A2 Preference Holder in which the certificate or certificates for Equity Shares are to be issued and whether or not such Equity Shares are to be issued in physical form or dematerialized form.
- 22.4.4 As soon as reasonably practicable, but in no event later than 7 (seven) Business Days from the date of the A2 Conversion Notice, the Company shall take all necessary corporate actions and obtain all necessary Consents and issue the appropriate number of Equity Shares into which the Relevant A2 CCPS are convertible at the A2 Conversion Ratio then in effect. Not later than the 7<sup>th</sup> (seventh) Business Day from the date of the A2 Conversion Notice, the Company shall deliver to such A2 Preference Holder:
  - (a) duly stamped and executed share certificates with respect to the A2 Conversion Shares issued on conversion of the Relevant A2 CCPS:
  - (b) certified true copies of all filings necessary to effect and validate the issue of the A2 Conversion Shares, including Form PAS-3;
  - (c) certified true copy of the register of members of the Company showing the A2 Preference Holder as the registered owner of the A2 Conversion Shares;
  - (d) duly acknowledged copies of all intimations and filings made by the Company with the RBI in respect of the conversion of the Series A2 CCPS and issue of Equity Shares in lieu thereof; and
    - (e) in the event that the A2 Preference Holder has requested in the A2 Conversion Notice or otherwise in writing to the Company and has provided all required information to the Company to hold the converted Equity Shares in a dematerialized form, evidence that such Equity Shares have been deposited in the account of such A2

Preference Holder or to the nominee or nominees of such A2 Preference Holder previously identified to the Company in writing accompanied with a beneficiary position statement issued by the concerned depository.

- 22.4.5 <u>Procedure for Mandatory Conversion</u>. In the case of a mandatory conversion of Series A2 CCPS pursuant to Article 22.4.2 hereinabove, the Company shall take all necessary corporate and other actions and obtain all Consents on or prior to the date of conversion, and shall provide the documents/ information listed in sub-Articles (a), (b), (c), (d) and (e) of Article 22.4.5 to the A2 Preference Holder on the date of conversion of the Series A2 CCPS.
- 22.4.6 No Fractional Shares. No fractional A2 Conversion Shares shall be issued upon conversion of Series A2 CCPS. If the computation of the number of A2 Conversion Shares to be issued, results in a fraction, then the number of A2 Conversion Shares shall be rounded down to the nearest whole number.
- 22.4.7 Conversion Price. The "Series A2 Conversion Price" for the Series A2 CCPS shall initially be equal to the Series A2 Original Issue Price and, subject to applicable Law, shall be adjusted in accordance with Article 22.4.8 below and Article 33. It is clarified that any adjustment of the Series A2 Conversion Price and A2 Conversion Ratio shall not automatically result in conversion of the Series A2 CCPS. In the event of any adjustment to the Series A2 Conversion Price and the A2 Conversion Ratio of the Series A2 CCPS, the Company, at its expense, shall promptly compute such adjustment and inform the holders of Series A2 CCPS of the details of such adjustment in writing. The Company shall upon the written request at any time of any A2 Preference Holder furnish or cause to be furnished to such holder a certificate setting forth (A) such adjustment, (B) the Series A2 Conversion Price and A2 Conversion Ratio at the time in effect, and (C) the number of Equity Shares and the amount, if any, of other property that at the time would be received upon the conversion of Series A2 CCPS.
- 22.4.8 <u>Conversion Price and Conversion Ratio Adjustments for Certain Dilutive</u>
  <u>Issuances, Splits and Combinations</u>. The Series A2 Conversion Price and A2
  Conversion Ratio shall be subject to adjustment from time to time as follows:
  - (a) Adjustments for Dilutive Issuances. If at any time after the Tranche III Closing Date, the Company issues to any Person any Equity Securities (other than pursuant to an Exempted Issuance), at a price per security that is lower than the Series A2 Conversion Price in effect immediately prior to such issuance (a "A2 Dilutive Issuance"), then the holders of Series A2 CCPS shall be entitled to a broad-based weighted average anti-dilution protection in accordance with Article 33.2. In such an event, the Company and the other Shareholders shall be bound to cooperate with the holders of Series A2 CCPS and the Company such that, the Company forthwith takes all necessary steps to adjust the Series A2 Conversion Price and A2 Conversion Ratio in accordance with Article 33.2. Notwithstanding the foregoing, if the adjustment set forth in this Article 22.4.8 is not permitted

to be made, in whole or in part, under applicable Law, the Parties shall take all necessary acts to put the A2 Preference Holder in the position that they would have been if such adjustment to the Series A2 Conversion Price had been made, including potentially the issuance of new Equity Shares to the A2 Preference Holder, or an Affiliate or designated nominee of a A2 Preference Holder, whereby the A2 Preference Holder, or such Affiliate or designated nominee thereof, are not required to pay any additional amounts for the issuance of such new Equity Shares, if permitted by applicable Law.

- (b) In the event the Company should at any time or from time to time after the Tranche III Closing Date fix a record date for the effectuation of a split or subdivision of the outstanding Equity Shares or the determination of holders of Equity Shares entitled to receive a dividend or other distribution payable in additional Equity Shares or other securities or rights convertible into, or entitling the holder thereof to receive directly or indirectly, additional Equity Shares (hereinafter referred to as "A2 Equity Share Equivalents") without payment of any consideration by such holder for the additional Equity Shares or the A2 Equity Share Equivalents (including the additional Equity Shares issuable upon conversion or exercise thereof), then, as of such record date (or the date of such dividend distribution, split or subdivision if no record date is fixed), the Series A2 Conversion Price and A2 Conversion Ratio shall be appropriately decreased so that the number of Equity Shares issuable on conversion of each Series A2 CCPS shall be increased in proportion to such increase of the aggregate of Equity Shares outstanding and those issuable with respect to such A2 Equity Share Equivalents.
- (c) If the number of Equity Shares outstanding at any time after the Tranche III Closing Date is decreased by a combination or consolidation (reverse stock split) of the outstanding Equity Shares, then, following the record date of such combination or consolidation (reverse stock split), the Series A2 Conversion Price and A2 Conversion Ratio shall be appropriately increased so that the number of Equity Shares issuable on conversion of each Series A2 CCPS shall be decreased in proportion to such decrease in outstanding Equity Shares.
- (d) Subject to the provisions of Clause 12.3 of the SSA, the A2 Conversion Ratio of the Series A2 CCPS may be adjusted at the option of the Indemnified Party, on the basis of the Fair Market Value, such that the number of A2 Conversion Shares increases to make good the Indemnity Amount (as defined in Clause 12.2 of the SSA) payable to the Indemnified Party ("A2 Indemnity Conversion Ratio"). Provided however that, in the event the Indemnifying Party pays the Indemnity Amount or part thereof, to the Indemnified Party, the A2 Indemnity Conversion Ratio shall be readjusted such that the number of A2 Conversion Shares is reduced to the extent of the Indemnity Amount (or part thereof), paid to the Indemnified Party. Provided further that, at the time of exit of the Investor in accordance with the terms of the Agreement

and the Articles, if the value received by the Investor as a consequence of the exit ("A2 Exit Value") and in respect of the A2 Conversion Shares (reckoned as a result of the adjustment of the A2 Conversion Ratio) is greater than the Indemnity Amount, the Investor shall transfer to the Promoters, such number of Investor Shares representing the difference between the A2 Exit Value of the A2 Conversion Shares and the Indemnity Amount. Any Tax liability associated with any such adjustment shall be borne by the Investor.

- (e) The A2 Conversion Ratio of the Series A2 CCPS shall also be adjusted at the time of allocation / issuance of any Equity Shares pursuant to the Agreed ESOP by the Company, the intention being that upon allocation / issuance of Equity Securities under the Agreed ESOP, the Investor shall not be diluted and the A2 Conversion Shares to be issued at the time of conversion are increased to the extent of Equity Securities allocated / issued under the Agreed ESOP.
- (f) The Conversion Ratio of the Series A2 CCPS shall be adjusted in respect of amounts receivable by CES-FZE, in accordance with the provisions of the Agreement.
- (g) The A2 Conversion Ratio of the Series A2 CCPS may be adjusted at the option of the holder of the Series A2 CCPS upon the occurrence of a Liquidity Event in accordance with the formula provided in Clause 3C (a) of the Agreement and Article 22A (c) of these Articles.
- (h) The A2 Conversion Ratio of the Series A2 CCPS may be adjusted at the option of the holder of the Series A2 CCPS upon the occurrence of a Liquidation Event in accordance with the formula provided in Clause 3C (b) of the Agreement and Article 22A (b) of these Articles.
- (i) The adjustments under this Article 22.4.8 shall not be mutually exclusive.
- 22.4.9 <u>Conversion Cost</u>. The Company shall bear all expenses arising from the conversion of the Series A2 CCPS as set out in this Article 22, including *inter alia*, any stamp duty applicable on the issuance of share certificates subsequent to conversion of the Series A2 CCPS.
- 22.4.10 Other Distributions. In the event the Company declares a distribution payable in securities of other Persons, evidences of indebtedness issued by the Company or other Persons, assets (excluding cash dividends) or options or rights not referred to in Article 22.4.8 (a) above, then, in each such case for the purpose of this Article 22.4.10, the A2 Preference Holders shall be entitled to a proportionate share of any such distribution as though they were the holders of the number Equity Shares into which their Series A2 CCPS are convertible as

of the record date fixed for the determination of the holders of Equity Shares entitled to receive such distribution.

- 22.4.11 Reservation of Equity Shares issuable Upon Conversion. The Company shall at all times reserve and keep available out of its authorized but unissued Equity Shares, solely for the purpose of effecting the conversion of the Series A2 CCPS, such number of its Equity Shares as shall from time to time be sufficient to effect the conversion of all outstanding Series A2 CCPS; and if at any time the number of authorized but unissued Equity Shares shall not be sufficient to effect the conversion of all then outstanding Series A2 CCPS, in addition to such other remedies as shall be available to the A2 Preference Holder, the Company shall take such corporate action as may be necessary to increase its authorized but unissued Equity Shares to such number of shares as shall be sufficient for such purposes, including, without limitation, to obtain the requisite shareholder approval of any necessary amendment to the Articles.
- 22.4.12 <u>Waiver of Adjustment to Conversion Price</u>. Notwithstanding anything herein to the contrary, any downward adjustment of the Series A2 Conversion Price may be waived, either prospectively or retroactively and either generally or in a particular instance, by the written consent or vote of the Investor. Any such waiver shall bind all future holders of Series A2 CCPS.
- 22A. Notwithstanding anything contained in these Articles:
  - (a). The Conversion Ratio of the combined Series A CCPS, Series A1 CCPS and Series A2 CCPS may be adjusted at the option of the holder of the CCPS upon the occurrence of a Liquidity Event in accordance with the formula and illustration thereof as stated below:

Conversion Ratio =

[(Total no. of Equity Shares of the Company) \* Investor's Proposed Shareholding – Existing Equity Shares held by the Investor] / (1 – Investor's Proposed Shareholding)

\_\_\_\_\_

No. of CCPS held by the Investor

Where,

Investor's Proposed Shareholding = (Investment Amount \* 2) / Fair Market Value of the Company as on the date of conversion

(b). The Conversion Ratio of the combined Series A CCPS, Series A1 CCPS and Series A2 CCPS may be adjusted at the option of the holder of the CCPS upon the occurrence of a Liquidation Event in accordance with the formula and illustration thereof as stated below:

Conversion Ratio =

[(Total no. of Equity Shares of the Company) \* Investor's Proposed Shareholding – Existing Equity Shares held by the Investor] / (1 – Investor's Proposed Shareholding)

\_\_\_\_\_

# No. of CCPS held by the Investor

#### Where.

Investor's Proposed Shareholding = (Investment Amount) / Total liquidation equity valuation of the Company as on the date of conversion

(c). The adjustments under this Article shall not be mutually exclusive.

#### XXVIII - PRE-EMPTIVE RIGHT OF THE INVESTOR

# 24. **Pre-emptive Right of the Investor**

- 24.1. The Shareholders shall have a pre-emptive right of subscription ("**Pre-Emptive Right**") in the event that the Company proposes to undertake any issuance of Equity Securities other than the Exempted Issuances (an "**Issuance**"). A Shareholder shall be entitled to apportion the Pre-Emptive Right hereby granted among itself and its Affiliates in such proportions as it deems appropriate, in which case the Affiliate of the Promoters shall be deemed to be a Promoter for the purposes of the Agreement and the Articles.
- 24.2. If undertaking an Issuance, the Company shall issue a written notice to the Shareholders ("Issuance Notice") setting forth in detail (a) the terms of the proposed issuance, including the proposed issuance price ("Issuance Price"), (b) the date of closing of the proposed Issuance (which shall not be less than thirty (30) days from the date of receipt of the Issuance Notice) and (c) the number of Equity Securities proposed to be issued (the "New Securities").
- 24.3. If a Shareholder wishes to exercise its Pre-emptive Right, then within 30 (thirty) Business Days from the delivery of the Issuance Notice, it/ he shall deliver a written notice to the Company, communicating its/ his intention to subscribe, at the price and on the terms specified in the Issuance Notice, up to that portion of such New Securities that equals the proportion that the number of Equity Securities held by such Shareholder (calculated on a Fully-Diluted Basis) bears to the total number of Equity Securities then outstanding (calculated on a Fully-Diluted Basis). In the event that a Shareholder (a "Non-Participating Shareholder") is unable to, or does not, for any reason whatsoever, (i) subscribe to its entitlement of the Issuance; and / or (ii) respond within 30 (thirty) Business Days from the delivery of Issuance Notice, then the other Shareholders (each a "Participating Shareholder") shall be entitled, by issue of a notice to the Company and the Non-Participating Shareholder, to subscribe either directly or through any of their Affiliates to the unsubscribed Equity Securities of the Issuance in proportion to their shareholding in the Company. In such case, the

- shareholding of the Non-Participating Shareholder in the Company shall stand diluted to the extent mentioned above.
- 24.4. If all New Securities that the Shareholder is entitled to obtain pursuant to Article 24.3 above are not elected to be obtained as provided in Article 24.3, then the Board of the Company may, subject to the prior written consent of the Investor, during the ninety (90) day period following the expiration of the period provided in Article 24.3 hereof, offer the remaining unsubscribed portion of such New Securities to any Person or Persons at a price not less than that, and upon terms no more favorable to the offeree than those, specified in the Issuance Notice. If the Company does not enter into an agreement for the issuance of the New Securities within such period, or if such agreement is not consummated within sixty (60) days of the execution thereof, the right provided hereunder shall be deemed to be revived and such New Securities shall not be offered unless first reoffered to the Investor in accordance with this Article 24.

#### XXIX - CALLS ON SHARES / DEBENTURES

#### 25. Calls on Shares/Debentures

- 25.1. The Board of Director may subject to the provisions of Section 49 of the Act from time to time to make such calls as they think fit upon the members in respect of all moneys unpaid on the shares held by them respectively and not by the conditions of allotment thereof made payable at fixed times and each members shall pay the amount of every call so made on him to the person and at the time and place appointed by the Directors. A call may be made payable by installments.
- 25.2. A call shall be deemed to have been made at the time when the resolution of the Board authorizing such call was passed.
- 25.3. Joint-holder of shares shall be jointly as well as severally liable for calls made on shares held by them.
- 25.4. No call shall exceed one fourth of the nominal amount of a share and be made payable within one month after the last preceding call was payable.
- 25.5. Not less than fourteen days' notice on any call shall be given specifying the time and place of payment and to whom such call be paid.
- 25.6. The Board may, if they think fit, receive from any member willing to advance the same, all or any part of the monies uncalled upon any shares held by him even if no part of the amount is called up, and upon all or any of the monies so advanced may (until the same would, but for such advance, become presently payable) pay interest at such rate (not exceeding 12% per annum without the sanction of the Company in the general meeting) as may be agreed upon between the member paying the sum in advance and the Board. Moneys paid in advance of calls shall not, in respect thereof, confer a right to vote or to dividend or to participate in profits. The Directors may at any time repay the amount so advanced upon giving to such member 3 (three) Months' notice in

writing. Provided that option or right to call of Shares shall not be given to any person or persons without the sanction of the Company in the General Meeting.

# 26. Payment in anticipation of call may carry interest

- (d) The Board may, if it thinks fit (subject to the provisions of Section 50 of the Act) agree to and receive from any Shareholder willing to advance the same, the whole or any part of the money due upon the shares held by him beyond the sums actually called up, and upon the amount so paid or satisfied in advance or so much thereof as from time to time and at any time thereafter as exceeds the amount of the calls then made upon and due in respect of the shares in respect of which such advance has been made, the Company may pay interest, as the Shareholder paying such sum in advance and the Board agree upon in accordance with the provisions of the Act, provided that the money paid in advance of calls shall not confer a right to participate in profits or dividend. The Directors may at any time repay the amount so advanced.
- (e) No Shareholder shall be entitled to voting rights in respect of the money (ies) so paid by him until the same would but for such payment, become presently payable.
- (f) The provisions of these Articles shall mutatis mutandis apply to the calls on Debentures of the Company.

#### XXX - SECURITIES CERTIFICATE

#### 27. Securities Certificate

- 27.1. The Company shall issue its securities in dematerialized form.
- 27.2. Unless the conditions of issue of any shares or debentures provide otherwise, the Company shall, within (i) 2 (two) Months after the allotment of any of its shares; (ii) within 1 (one) Month from the date of receipt of transfer request from the Depository Participant or, as the case may be, of the intimation of transmission, credit the shares to the holders demat accounts as the case may be in accordance with the provisions of the applicable laws for the time being in force.
- 27.3. In case the law permits issue physical issue of shares, every certificate shall be under the Seal of the Company and shall specify the share to which it relates and amount paid thereon, and shall bear signature of at least two directors of the Company, and the secretary of the company or any person authorized by the Board of Directors in this regard.
- (d) If any certificate be worn out, defaced, mutilated or torn or if there be no further space on the back thereof for endorsement of transfer, then upon production and surrender thereof to the Company, a new certificate may be issued in lieu thereof, and if any certificate is lost or destroyed then upon proof thereof to the satisfaction of the Company and on execution of such indemnity as the Company deems adequate, being given, a new certificate in lieu thereof shall be given to the party entitled to such lost or destroyed certificate, within a period of 30 days from the receipt of such lodgement. Every certificate under the Articles shall be issued without payment of fees if the Directors so decide, or on payment of such fees (not exceeding Rupees two for each certificate) as the Directors shall prescribe. Provided that, no fee shall be charged for

issue of a new certificate in replacement of those which are old, defaced or worn out or where there is no further space on the back thereof for endorsement of transfer.

Provided that notwithstanding what is stated above the Directors shall comply with such rules or regulation or requirements of any Stock Exchange or the rules made under the Act or rules made under Securities Contracts (Regulation) Act, 1956 or any other Act, or rules applicable thereof In this behalf.

The provision of this Article shall mutatis mutandis apply to debentures of the company.

- 27.4. In respect of share or shares held jointly by several persons, the Company shall not be bound to issue more than one certificate and delivery and certificate for all securities to one of several joint shall be sufficient delivery to all such security holders.
- 27.5. The Company shall observe rules and conditions as may be prescribed under the Act and the Companies (Shares and Debentures) Rules, 2014 for renewal of security certificates or issue of duplicate security certificates.

#### XXXI - FORFEITURE OF SHARES

#### 28. Forfeiture of Shares

- 28.1. If a member fails to pay any call within the time allowed by the Board of Directors at the time of making a call or subsequently, the Board may issue final notice naming a further date (not being earlier than the expiry of fourteen days from the date of service of the notice) on or before which the payment required by the notice is to be made and if the calls remain unpaid till then, the shares may be subject to forfeiture without further notice. Such forfeiture shall include all dividends declared in respect of the forfeited shares and not actually paid before the forfeiture.
- 28.2. Forfeited shares may be restored to the holder thereof on payment of the call money in full, before the forfeited shares are sold to other member of members of the Company or otherwise disposed of.
- 28.3. Any shares so forfeited shall be deemed to be the property of the Company and may be sold, re-allotted or otherwise disposed of either to existing members or to any other person upon such terms and conditions and in such manner as the Board shall think fit.
- 28.4. A person whose shares have been forfeited shall cease to be a member in respect of the forfeited shares, but shall notwithstanding the forfeiture, be liable to pay to the Company all moneys, which at the time of forfeiture were presently payable by him to the Company in respect of the share.
- 28.5. The Company may receive the consideration, if any, given for the shares on any sale or disposal thereof and may execute a transfer of the share in favour of the person to

whom the share is sold or disposed of. The transfer thereupon shall be registered as the holder share shares.

28.6. Neither a judgment nor a decree in favour of the Company for calls or other moneys due in respect of any Shares nor any part payment or satisfaction thereof nor the receipt by the Company of a portion of any money which shall from time to time be due from any Member in respect of any Shares either by way of principal or interest nor any indulgence granted by the Company in respect of payment of any such money shall preclude the forfeiture of such Shares as herein provided. There shall be no forfeiture of unclaimed dividends before the claim becomes barred by applicable Law.

#### XXXII - TRANSFER OF SHARES

29. **General**. Transfer of any Equity Securities by any Shareholder must comply with the provisions of Article 29 and Article 30 and the Company shall not record or register any Transfer that does not satisfy the provisions of Article 29 and Article 30. Any attempt by any Shareholder to Transfer its Equity Securities in contravention of the provisions contained herein shall be considered void and invalid. A common form of transfer should be used.

#### 30. **Conditions on Transfers**

#### 30.1. Lock-in of Promoter Shares.

Without limiting any other provision of these Articles, till the consummation of the exit of Investor in accordance with Article 43 of these Articles, the Promoters will not Transfer, or cause to be Transferred, any Equity Securities in the Company, or otherwise achieve or cause to be achieved any liquidity with respect to any Equity Securities of the Company held directly or indirectly by them in the Company and its Subsidiaries, without the prior written consent of the Investor. Should the Investor consent to such Transfer, the same shall be in accordance with and subject to the provisions of these Articles and the Agreement. Provided however that, in the event the Promoters desire to conduct an estate planning of their respective assets, they shall be permitted to Transfer up to 15% (fifteen percent) of their aggregate shareholding in the Company cumulatively to a trust for the benefit of the legal heirs of the Promoters, subject to the trustee of such trust executing a Deed of Adherence. Provided further that the Promoters shall also be permitted to transfer their shareholding in the Company during the lock-in period in the following circumstances:

- (a) Encumbrance for the purpose of availing any borrowings by the Company as part of the approved Business Plan and approved in accordance with the terms of the Agreement and the Articles;
- (b) a re-organization, consolidation, spin-off, merger, acquisition or other similar transaction by the Company and/or Subsidiaries approved in accordance with the terms of the Agreement and the Articles; and
- (c) any transmission of shares by operation of Law.

#### 30.2. **Right of First Offer**.

- (a) Subject to Article 30.1 above (Lock-in of Promoter Shares) and Article 30.5 (Right of First Refusal) below, if any Shareholder ("Transferring Party") proposes to Transfer within 60 (sixty) months from the Closing Date, any or all of the Equity Securities held by such Transferring Party in the Company to a Third Party (other than a Competitor), then the Transferring Party shall first offer all but not less than all of the Equity Securities held by it ("ROFO Securities") to the other Shareholders ("Non-Transferring Party") (such a right being referred to as "Right of First Offer"). Such Right of First Offer shall be exercisable in the manner set out below. Any right/ action under this Article 30.2 to be exercised by the Promoters as a Non-Transferring Party shall be exercised by the Promoters as a block and not by each Promoter separately or independently.
- (b) The Transferring Party shall issue a written notice ("**ROFO Notice**") to the Non Transferring Party stating therein: (a) the name of the Transferring Party, (b) its/ his intention to sell the ROFO Securities, and (c) the number of Equity Securities held by the Transferring Party in the Company as on the date of such ROFO Notice.
- (c) Within a period of 60 (sixty) calendar days from the date of receipt of the ROFO Notice, the Non Transferring Party shall be entitled to respond to the ROFO Notice in writing (such notice, an "Offer Response Notice") indicating its acceptance, the price offered for such ROFO Securities ("Offer Price") and reasonable documentary evidence to display availability of, or accessibility to, requisite funds with the Non-Transferring Party for the purchase of ROFO Securities.
- (d) Within 15 (Fifteen) Business Days after receipt of the Offer Response Notice, the Transferring Party shall provide a written notice to the Non-Transferring Party, either accepting or rejecting the Offer Price ("ROFO Acceptance Notice"). If the ROFO Acceptance Notice is not provided within aforementioned time then the ROFO Notice shall be deemed to have been rejected. If the Offer Price is accepted, the Transferring Party shall consequently Transfer the ROFO Securities to the Non-Transferring Party within 90 (ninety) calendar days from the date of the ROFO Acceptance Notice. If the Non-Transferring Party has rejected the offer set forth in the ROFO Notice or if the Offer Price is rejected by the Transferring Party, the Transferring Party shall be entitled to Transfer the ROFO Securities to a Third Party within a period of 90 (ninety) calendar days from the date of Offer Response Notice of ROFO Acceptance Notice as the case may be ("Offer **Period**"), subject to the provisions of Article 30.3 (Tag-Along Right) and provided (i) the Transferring Party shall not offer the ROFO Securities to a Third Party at a price lesser than the Offer Price and/or on terms which are more favourable to the Third Party as compared to the terms offered by the Non-Transferring Party; and (ii) such Third Party buyer executes a Deed of Adherence agreeing to abide by the terms and conditions of the Agreement.

For the avoidance of doubt, the Third Party referred to in this Article 30.2 shall not include a Competitor.

- (e) The Offer Period shall be extended to the extent required to obtain approval(s), required if any, under the applicable Law. The Transfer of the ROFO Securities as per applicable Law (including delivery of share certificates, etc.) and the payment of the Offer Price for the ROFO Securities shall occur simultaneously. The Transferring Party shall provide customary representations and warranties with respect to title of the ROFO Securities (including that the title is free from all Encumbrances) and authority to sell.
- (f) Each Party shall bear its own costs and expenses incurred in connection with the exercise of Right of First Offer provided to such Party under this Article 30.2. All stamp duty and other transfer Taxes in relation to the Transfer of the ROFO Securities to the Non Transferring Party (other than those in nature of capital gains tax) shall be borne by the Company.
- (g) In the event that such Transfer is not consummated within the Offer Period, the Transferring Party shall not be permitted to sell the ROFO Securities pursuant to this Article 30.2 without again complying with each of the requirements of this Article 30.2.
- (h) The Non Transferring Party shall be entitled to nominate any of its Affiliates, at its sole discretion, to acquire/ hold the ROFO Securities pursuant to the provisions of this Article 30.2.

## 30.3. Tag Along Rights.

- (a) If the Non-Transferring Party is the Investor and the Investor does not intend to exercise the Right of First Offer under Article 30.2 above ("Non ROFO Exercising Party") or does not respond to the ROFO Notice within a period of 60 (sixty) calendar days from the date of receipt of the ROFO Notice, the Transferring Party being the Promoters in this case, shall deliver a written notice to the Non ROFO Exercising Party ("Sale Notice") stating: (a) that it/ they intend to Transfer the ROFO Securities ("Sale Securities") to a Third Party (the "Proposed Buyer"); (b) the name of the Proposed Buyer; (c) the price and the key terms and conditions on which the Sale Securities are proposed to be Transferred to the Proposed Buyer; and (d) the number of Sale Securities proposed to be Transferred to the Proposed Buyer. Any right/action under this Article 30.3 to be exercised by the Promoters shall be exercised by the Promoters as a block and not by each Promoter separately / independently.
- (b) Subject to the provisions of this Article 30.3, within a period of 15 (Fifteen) calendar days from the date of receipt of the Sale Notice ("Tag Exercise Period"), the Non ROFO Exercising Party ("Tag Along Party"), shall have the right (but not the obligation), ("Tag Along Right") to offer all or part of the Equity Securities held by the Tag Along Party to the Proposed Buyer ("Tag Securities") and at a price and on terms and conditions no less favourable than

those offered to the Transferring Party by the Proposed Buyer by intimating the Transferring Party in writing ("Tag Along Notice"). The Parties agree that the Tag Along Right under this Article 30.3 is only a right of the Investor and shall not be available to the Promoters. The number of Tag Securities shall not exceed the product of the total number of Equity Securities held by the Tag Along Party on a Fully Diluted Basis and the fraction, the numerator of which shall be number of Sale Securities proposed to be transferred by the Transferring Party and the denominator of which shall be the total number of Equity Securities outstanding in the Company on a Fully Diluted Basis. Provided however that, where the Transferring Party is the Promoters (acting as block) and such Transfer results in a change of control, the Investor shall be entitled to Transfer all of their Equity Securities to the Proposed Buyer.

- (c) If the Tag Along Party has exercised its Tag Along Right by sending a Tag Along Notice to the Transferring Party, within the Tag Exercise Period, the Transferring Party shall intimate this to the Proposed Buyer. If the Proposed Buyer is willing to purchase all the Tag Securities, then the Transferring Party, the Tag Along Party and the Proposed Buyer shall complete the transfer of all the Sale Securities and Tag Securities, at the price and on the terms and conditions as mentioned in the Sale Notice. If, however, the Proposed Buyer cannot or is not willing to purchase all of the Tag Securities, then the Transferring Party shall not be entitled to transfer the Sale Securities to the Proposed Buyer and such Transferring Party shall be obligated to follow the procedure set out in Article 30.2 (Right of First Offer) and this Article 30.3 for any subsequent sale of any securities held by the Transferring Party.
- (d) In the event the Non ROFO Exercising Party does not exercise its Tag Along Right within the Tag Exercise Period, then the Transferring Party shall be entitled to Transfer all but not less than all of the Sale Securities to the Proposed Buyer on the terms set out in the Sale Notice, within a period of 90 (ninety) calendar days from the date of expiry of the Tag Exercise Period ("Sale Period"). The Sale Period shall be extended to the extent required to obtain approval(s), required if any, under the applicable Law.
- (e) In the event that the Transferring Party is unable to Transfer the Sale Securities within the Sale Period (including any period extended in accordance with Article 30.3(d)), then such Transferring Party shall be obligated to follow the procedure set out in Articles 30.2 and this Article 30.3 for any subsequent sale of any securities held by the Transferring Party.
- 30.4. <u>Limitations to Rights of First Offer and Tag-Along Right.</u> Notwithstanding the provisions of Articles 30.2 and 30.3, the Right of First Offer and the Tag-Along Right of the Investor shall not apply to any sale of Equity Securities to the public pursuant to an IPO (including a QIPO).

# 30.5. Right of First Refusal

- (a) In the event the Investor proposes to Transfer any Equity Securities of the Company to a Competitor within 60 (sixty) months from the Closing Date (in each case "ROFR Sale Period"), then the Investor shall give to the Promoters, a written notice of the Investor's intention to make the Transfer accompanied by the term sheet or letter of intent provided by the Competitor recording the proposed terms of such Transfer (the "ROFR Notice"). The ROFR Notice shall include (i) a description of the Equity Securities to be transferred (the "Offered Securities"), (ii) the name(s) and address(es) of the Competitor, and (iii) the purchase price proposed to be paid for the Offered Securities.
- (b) If the Promoters intend to purchase the Offered Securities, the Promoters may respond to the ROFR Notice indicating their acceptance to purchase the Offered Securities by themselves, along with reasonable documentary evidence to display availability of, or accessibility to, requisite funds with the Promoters for the purchase of Offered Securities ("Exercise Notice") within 30 (Thirty) calendar days of the receipt of the ROFR Notice ("ROFR Exercise Period").
- (c) Within 30 (thirty) calendar days from the date of dispatch of the Exercise Notice, ("**ROFR Confirmation Period**"), the Promoters shall have the right to provide to the Investor, confirmation of the Promoters' willingness to purchase the Offered Securities ("**ROFR Confirmation Notice**").
- (d) Within 60 (Sixty) calendar days of dispatch of the ROFR Confirmation Notice ("Purchase Period"), the Promoters shall be obligated to purchase the Offered Securities from the Investor and the Investor shall be obligated to Transfer the Offered Securities to the Promoters. Any transfer of the Offered Securities to the Promoters shall be completed within the Purchase Period. In case of failure of the Promoters to purchase the Offered Securities within the Purchase Period, the Investor shall have the right to either sell the Offered Securities in accordance with Article 30.5(e) hereinbelow, or trigger an Event of Default whereupon the Drag Right as provided under the provisions of Article 45.2 shall apply. If and only if the Offered Securities constitute all of the Equity Securities held by the Investor, then the Promoters shall have the right to purchase such Offered Securities through its nominee.
- (e) If (i) the Promoters do not deliver the Exercise Notice within the ROFR Exercise Period; or (ii) the Promoters do not deliver the ROFR Confirmation Notice within the ROFR Confirmation Period; or (iii) the Promoters deliver the ROFR Confirmation Notice, but fail to purchase the Offered Securities within the Purchase Period (either through themselves or through any Third Party or nominee), then the Investor shall have 90 (ninety) calendar days from the expiry of the ROFR Exercise Period or the Purchase Period, as the case may be ("Free Sale Period"), within which it can transfer the Offered Securities on the terms set out to the relevant transferee stated in the ROFR Notice.

- (f) Notwithstanding the above, if the Offered Securities are Transferred to a Competitor prior to expiry of 60 (sixty) months from the Closing Date, such Competitor shall not be entitled to exercise the following rights under the Agreement:
  - (i) Pre-emptive right (Article 24);
  - (ii) Reserved Matter in respect of any amendment to the Articles that may adversely affect the rights of such Competitor.
- 30.6. <u>Transfer to Competitor.</u> In case of any Transfer of Equity Securities by the Investor to a Competitor after the expiry of 60 (sixty) months from the Closing Date, the Competitor shall be entitled to all the rights (except rights relating to exit under Article 43) of the Investor under the Agreement.
- 30.7. Prior to the transfer of Offered Securities by the Investor to the Competitor in accordance with the provisions of Article 30.5, all the Offered Securities shall first automatically and mandatorily convert into Equity Shares of the Company in accordance with provisions of Article 17.2 and Article 17.5 and the Equity Shares into which the Offered Shares are converted shall then be transferred to the Competitor.
- 30.8. <u>Transfer to Affiliates</u>. The Investor shall be permitted to Transfer their Equity Securities to any of its Affiliates without complying with provisions of this Article 30, provided that such Affiliate executes the Deed of Adherence.
- 30.9. <u>Consents</u>. The purchase or sale/Transfer under this Article 30 shall be subject to the necessary Consents being obtained. The Company and the Transferring Party, shall each use their best endeavors to obtain the necessary Consents. Time periods specified in this Article 30 shall be extended by the time taken to obtain necessary Consents.
- 30.10. **Director's powers to refuse to register a transfer.** Subject to the provisions of Sections 58 and 59 of the Act, these Articles and other applicable provisions of the Act or any other Law for the time being in force, the Board may, refuse to register the transfer of, or the transmission by operation of law of the right to, any securities or interest of a Shareholder in the Company. The Company shall, within 30 (thirty) days from the date on which the instrument of transfer, or the intimation of such transmission, as the case may be, was delivered to the Company, send a notice of refusal to the transferee and transferor or to the Person giving notice of such transmission, as the case may be, giving reasons for such refusal.

Provided that, registration of a transfer shall not be refused on the ground of the transferor being either alone or jointly with any other Person or Persons indebted to the Company on any account whatsoever except when the Company has a lien on the shares.

Subject to the applicable provisions of the Act and these Articles, the Directors shall have the absolute and uncontrolled discretion to refuse to register a Person entitled by transmission to any shares or his nominee as if he were the transferee named in any ordinary transfer presented for registration, and shall not be bound to give any reason

for such refusal and in particular may also decline in respect of shares upon which the Company has a lien.

#### 30.11. Instrument of Transfer

(b) The instrument of transfer of any Share shall be in writing and all the provisions of the Act shall be duly complied with in respect of all transfer of Shares and registration thereof. The Company shall use the form of transfer, as prescribed under the Act, in all cases. In case of transfer of Shares, where the Company has not issued any certificates and where the Shares are held in dematerialised form, the provisions of the Depositories Act shall apply.

In accordance with Section 56 of the Act, the Rules and such other conditions as may be prescribed under Law, every instrument of transfer of shares held in physical form shall be in writing. In case of transfer of shares where the Company has not issued any certificates and where the shares are held in dematerialized form, the provisions of the Depositories Act shall apply. The Company shall cause to be kept a register and index of members / beneficial owners in accordance with all applicable provisions of the Companies Act, 2013 and the Depositories Act, 1996 with details of shares held in physical and dematerialised forms in any medium as may be permitted by law including in any form of electronic medium. The Company shall be entitled to keep in any State or Country outside India a branch register of beneficial owners / Register of Members, resident in that State or Country. The Register and Index of Beneficial Owners maintained by a Depository under the Depositories Act, shall be deemed to be the Register and Index (if applicable) of Shareholders and Securityholders for the purposes of these Articles.

- (e) (i) An application for the registration of a transfer of the shares in the Company may be made either by the transferor or the transferee within the time frame prescribed under the Act.
  - (ii) Where the application is made by the transferor and relates to partly paid shares, the transfer shall not be registered unless the Company gives notice of the application to the transferee in a prescribed manner and the transferee communicates no objection to the transfer within 2 (two) weeks from the receipt of the notice.
- (f) No fee shall be charged for registration of transfer, transmission, probate, succession certificate and letters of administration, certificate of death or marriage, power of attorney or similar other document.

#### XXXIII - TRANSMISSION OF SHARES/ DEBENTURE

# 31. Transmission of Shares/Debentures

- 31.1. On the death of a member, the survivor or survivors where the member was a joint holder, and his nominee or nominees or the heir of the deceased or his legal representative shall be the persons recognized by the Company as having any title or interest in the shares/ debentures.
- 31.2. Nothing in Article 17.1 above, shall release the estate of a deceased joint holder from any liability in respect of any share/debenture which has been jointly held by him with other persons.

- 31.3. Any heir or successor being owner of the shares in consequence of the death of the member, upon such evidence being produced as may properly be required by the Board elect either (i) to register his or their names as the holders of the shares or (ii) to make arrangement to transfer of shares in the way as the deceased member could have made in accordance with the transfer clause of these Articles.
- 31.4. If the person or persons aforesaid shall elect to transfer the shares, he shall testify his election by executing a transfer of the share, in conformity with the transfer clause of these Articles.
- 31.5. A person becoming holder of share on such transmission, shall become member of the Company entitled to the same dividends and advantages as his predecessors would have otherwise enjoyed, but if he be a minor, lunatic or insolvent, he shall have no voting right.
- 31.6. The person so becoming entitled and willing to elect to be registered holder of the share shall deliver or send to the Company a notice in writing stating that he so elects.
- 31.7. The Board shall register his name in place of the deceased member.
- 31.8. If the persons aforesaid fails to elect either to be registered himself to transfer the share, the Board on having confirmation of the event of the death of the member shall serve a notice to person asking him to comply with the requirement within three months of the issue of the notice failing which Board may thereafter withhold payment of all dividends, bonuses or other money payable in respect of the share.
- 31.9. If nothing is done within three months, the Company shall issue a notice in the local newspaper asking the legal heirs or the legal representative of the deceased member whether to be registered as shareholders or to transfer the said shares in conformity with these Article three months thereafter, and if nothing is done within the said period, the shares shall be liable to forfeiture and the forfeiture clause of these Articles shall prevail. The forfeited shares may be re-issued to the member at the called up value of the shares and forfeited amount be transferred to the Capital Reserve Account or dealt with as the Board of Directors shall think fit.
- 31.10. The Company shall have a first and paramount lien on (i) every share, not being a fully paid share, for all moneys, whether presently payable or not, called or payable in respect of the share and (ii) on all shares, not being fully paid shares, standing in the name of any person, for all moneys, presently payable by him, provided that the Board of Directors may at any time declare any share to be wholly or in part exempt from the provisions of the clause.
- 31.11. The Company's lien, if any, on a share shall extend to all dividends payable thereon.
- 31.12. For the purpose of enforcing such lien as aforesaid the Directors may sell the Shares subject thereto in such manner as they shall think fit, but no sale shall be made:
  - (a) Unless a sum in respect of which the lien exists is presently payable; and

(b) Until the expiration of seven days after a notice in writing, stating and demanding payment of such part of the amount in respect of which the lien exists and as is presently payable, has been given to the registered holder for the time being of the Share or to the person entitled thereto by transmission, and default shall have been made by him in payment of the sum payable as aforesaid for seven days after such notice.

The residue, if any, shall be paid to the person entitled by transmission to the shares so sold at the date of the sale.

31.13. The fully paid up Shares shall be free from all lien and in the case of partly paid up Shares the Company's lien shall be restricted to moneys called or payable at a fixed time in respect of such Shares.

#### 32. COMPANY'S LIEN:

#### iii. On shares:

- (f) The Company shall have a first and paramount lien:
  - (ii) on every share (not being a fully paid share), for all money (whether presently payable or not) called, or payable at a fixed time, in respect of that share;

Provided that the Board may, at any time, declare any shares wholly or in part to be exempt from the provisions of this Article.

- (g) Company's lien, if any, on such partly paid shares, shall extend to all Dividends payable and bonuses declared from time to time in respect of such shares.
- (h) Unless otherwise agreed, the registration of a transfer of shares shall operate as a waiver of the Company's lien, if any, on such shares. The fully Paid up shares shall be free from all lien and that in case of partly paid shares, the Company's lien shall be restricted to money called or payable at a fixed time in respect of such shares.
- (i) For the purpose of enforcing such lien, the Board may sell the shares, subject thereto in such manner as they shall think fit, and for that purpose may cause to be issued a duplicate certificate in respect of such shares and may authorise one of their Shareholders to execute and register the transfer thereof on behalf of and in the name of any purchaser. The purchaser shall not be bound to see to the application of the purchase money, nor shall his title to the shares be affected by any irregularity or invalidity in the proceedings in reference to the sale.

#### Provided that no sale shall be made:

- (iii) unless a sum in respect of which the lien exists is presently payable; or
- (iv) until the expiration of 14 days after a notice in writing stating and demanding payment of such part of the amount in respect of which the lien exists as is presently payable, has been given to the registered holder for the time being of the share or the person entitled thereto by reason of his death or insolvency.

The net proceeds of any such sale shall be received by the Company and applied in payment of such part of the amount in respect of which the lien exists as is presently payable. The residue,

if any, shall (subject to a like lien for sums not presently payable as existed upon the shares before the sale) be paid to the Person entitled to the shares at the date of the sale.

(j) No Shareholder shall exercise any voting right in respect of any shares registered in his name on which any calls or other sums presently payable by him have not been paid, or in regard to which the Company has exercised any right of lien.

#### iv. On Debentures:

- (f) The Company shall have a first and paramount lien:
  - (ii) on every Debenture (not being a fully paid Debenture), for all money (whether presently payable or not) called, or payable at a fixed time, in respect of that Debenture;

Provided that the Board may, at any time, declare any Debentures wholly or in part to be exempt from the provisions of this Article.

- (g) Company's lien, if any, on the Debentures, shall extend to all interest and premium payable in respect of such Debentures.
- (h) Unless otherwise agreed, the registration of a transfer of Debentures shall operate as a waiver of the Company's lien, if any, on such Debentures. The fully paid up Debentures shall be free from all lien and that in case of partly paid Debentures, the Company's lien shall be restricted to money called or payable at a fixed price in respect of such Debentures.
- (i) For the purpose of enforcing such lien, the Board may sell the Debentures, subject thereto in such manner as they shall think fit, and for that purpose may cause to be issued a duplicate certificate in respect of such Debentures and may authorize the debenture trustee acting as trustee for the holders of Debentures or one of the holder of Debentures to execute and register the transfer thereof on behalf of and in the name of any purchaser. The purchaser shall not be bound to see to the application of the purchase money, nor shall his title to the Debentures be affected by any irregularity or invalidity in the proceedings in reference to the sale.

Provided that no sale shall be made:

- (iii) unless a sum in respect of which the lien exists is presently payable; or
- (iv) until the expiration of 14 days after a notice in writing stating and demanding payment of such part of the amount in respect of which the lien exists as is presently payable, has been given to the registered holder for the time being of the Debenture or the Person entitled thereto by reason of his death or insolvency.

The net proceeds of any such sale shall be received by the Company and applied in payment of such part of the amount in respect of which the lien exists as is presently payable. The residue, if any, shall (subject to a like lien for sums not presently payable as existed upon the Debentures before the sale) be paid to the Person entitled to the Debentures at the date of the sale.

(j) No holder of Debentures shall exercise any voting right in respect of any Debentures registered in his name on which any calls or other sums presently payable by him have not been paid, or in regard to which the Company has exercised any right of lien.

#### 33. **Anti-dilution**

33.1. Subject to Article 38 (Reserved Matters), if at any time after the Closing Date, the Company issues to any Third Party, any Equity Securities, at a price per share that is lower than the price at which the Investor subscribed to the Series A CCPS (such an event being referred to as a "**Dilution Event**"), then the Investor shall be entitled to a broad-based weighted average anti-dilution protection in accordance with this Article 33.2 below simultaneously with such Dilution Event or at such other time as may be indicated by the Investor at the Investor's sole option.

#### 33.2. **Broad-Based Anti-Dilution Protection**

- (a) Relevant Calculations:
  - (i) Determine Anti-dilution Price (i.e., Broad Based Weighted Average Share Price) in the following manner:

# (OS immediately prior to the Dilution Event x Investor Price Per $\frac{Share) + AC}{OS \ immediately \ following \ issuance}$

Where:

"OS" means the number of Equity Shares issued and outstanding on a Fully Diluted Basis,

"Investor Price Per Share" means the fraction, the numerator of which is the Investment Amount and the denominator of which is the total number of Investor Shares, and

- "AC" means the aggregate consideration to be received by the Company in connection with the new issuance.
- (ii) Determine number of Equity Shares that Investor would have received if the Investor had paid the Anti-dilution Price for the Equity Shares or Equity Securities convertible into Equity Shares subscribed by it, by dividing the aggregate consideration paid by Investor pursuant to the Share Subscription Agreement by the Anti-dilution Price.
- (iii) The number of additional Equity Shares or instruments/ Equity Securities convertible into Equity Shares to be issued to the Investor (i.e., Additional Securities) shall equal the number of Equity Shares or instruments/ Equity Securities convertible into Equity Shares that Investor would have received as determined pursuant to II above minus the number of Equity Shares or instruments/ Equity Securities convertible into Equity Shares actually held by the Investor.

- (b) In performing the foregoing calculations, the following provisions shall be applicable:
  - (i) In a Dilution Event, in the case of the new issuance being for cash, the aggregate consideration shall be deemed to be the amount of cash paid therefor before deducting therefrom any discounts, commissions or placement fees payable by the Company to any underwriter or placement agent in connection with the issuance and sale thereof.
  - (ii) In a Dilution Event, in the case of the new issuance being for a consideration, in whole or in part other than cash, the consideration other than cash shall be deemed to be the Fair Market Value thereof.
  - (iii) In the case of the Dilution Event involving the issuance of options to purchase or rights to subscribe to Equity Shares, securities by their terms convertible into or exchangeable for Equity Shares, or options to purchase or rights to subscribe for such convertible or exchangeable securities (other than Equity Shares, options or other securities issued under any ESOP or Equity Shares, options or other securities issued upon the exercise thereof):
    - (A) the aggregate maximum number of Equity Shares deliverable upon exercise of such options to purchase, exercise of rights to subscribe for Equity Shares or conversion of or in exchange for any such convertible exchangeable securities, shall be deemed to have been issued at the time such options or rights were issued and for a consideration equal to the consideration (determined in the manner provided above), if any, received by the Company upon the issuance of such options or rights plus the exercise price provided in such options or rights for the Equity Shares covered thereby;
    - (B) on any increase in the number of shares or decrease in exercise price of Equity Shares deliverable upon exercise of any such options or rights or conversions of or exchanges for such securities, other than a change resulting from the anti-dilution provisions thereof, the Weighted Average Share Price shall be readjusted retroactively to give effect to such increase or decrease and additional Equity Shares shall be issued to the Investor;
    - (C) on any decrease in the number of shares or increase in exercise price of Equity Shares deliverable upon exercise of any such options or rights or conversions of or exchanges for such securities, the Weighted Average Share Price shall be readjusted retroactively to give effect to such decrease or increase; and

- (D) no further adjustment shall be made as a result of the actual issuance of Equity Shares on the exercise of any such rights or options or any conversion or exchange of any such securities.
- (iv) All references to Equity Shares shall be on a Fully Diluted Basis.
- (v) All calculations of the Anti-dilution Price shall be made to the nearest one one-hundredth of a Rupee.
- (vi) In the event that of the Additional Securities, determined in the manner provided above, being a fraction, such number shall be round up to the nearest whole share as follows:
  - (A) in case the fraction is up to 0.49, then the number of shares to be issued shall be rounded off to the next lower number; and
  - (B) in case the fraction is 0.5 or more, then the number of shares to be issued rounded off to the next higher number.
- 33.3. In such a Dilution Event, the Company shall forthwith take all necessary steps, at the sole option of the Investor, but subject to applicable Law, to (i) adjust the Conversion Ratio such that the number of Conversion Shares increases in accordance with calculation made pursuant to the broad-based weighted average anti-dilution protection in accordance with Article 33.2 above (*Broad Based Anti-Dilution Protection*); or (ii) issue additional Equity Shares or instruments/ Equity Securities convertible into Equity Shares to the Investor, either directly and/or to their respective nominees (as may be opted by the Investor), at such time as indicated by the Investor and for no additional consideration or consideration other than cash and if not permissible under applicable Law, at the lowest permissible consideration under applicable Law such that the Investor price per share is reduced to the price arrived at pursuant to the calculations made in accordance with Article 33.2 above.
- 33.4. In case of a Corporate Action Event, the Company shall forthwith take all necessary steps, at the sole option of the Investor, to (i) adjust the Conversion Ratio such that the number of Conversion Shares increases in accordance with calculation made pursuant to the broad-based weighted average anti-dilution protection in accordance with Article 33.2 above; and/or (ii) issue additional Equity Shares or instruments/ Equity Securities convertible into Equity Shares to the Investor, either directly and/or to their respective nominees (as may opted by the Investor), at such time as indicated by the Investor and for no additional consideration or consideration other than cash and if not permissible under applicable Law, at the lowest permissible consideration under applicable Law, the effect of which shall be such that there is no Dilution Event on account of such Corporate Action Event.
- 33.5. For the purposes of this Article 33, the Parties shall be bound to co-operate with each other. Further, if the adjustment as contemplated in this Article 33 cannot be undertaken due to applicable Law, then the Parties shall mutually discuss and agree on an

alternative to achieve the adjustment as aforesaid including without limitation through the issuance of rights shares, bonus shares, etc. to the Investor or its nominee as the case may be.

- 33.6. Nothing contained in this Article 33 shall apply to the following categories of issuance of Equity Securities by the Company:
  - (a) Equity Securities issued pursuant to any ESOP or options granted under such ESOP:
  - (b) Equity Securities issued to the Investor and/or its nominees pursuant to this Article 33:
  - (c) The issuance of Equity Securities in an IPO or QIPO;
  - (d) Equity Shares issued upon conversion of the Series A CCPS into Equity Shares; and
  - (e) Equity Securities issued to banks and/or financial institutions pursuant to any credit facilities availed after the date of the Agreement with the written consent of the Investor.

#### XXXV - AFFILIATES

34. Notwithstanding any other provision of these Articles, but subject to execution of the Deed of Adherence, the Investor and/or its Affiliate/s may, at any time and from time to time during the subsistence of the Agreement, acquire any new Shares or other Equity Securities offered to them by the Company, the Promoters and/or the other Shareholders under the provisions of the Agreement and/or transfer any existing Shares or other Equity Securities of the Company held by them to one or more of their Affiliates.

# XXXVI - INFORMATION, REPORTING, INSPECTION AND ACCOUNTING STANDARDS

- 35. Information, Reporting, Inspection And Accounting Standards
  - 35.1. The Company shall provide to the Investor:
    - (a) within 5 (five) Business Days from the last day of each month, copies of Management Reports;
    - (b) within 150 (one hundred and fifty) calendar days following the closure of the preceding Financial Year, annual (audited) Financial Statements prepared in accordance with Indian GAAP;
    - (c) within 30 (thirty) days after the end of each financial quarter, standalone unaudited Financial Statements for the Company and the Subsidiaries prepared in accordance with Indian GAAP;

- (d) within 45 (forty five) days after the end of each financial quarter, consolidated unaudited Financial Statements for the Company and the Subsidiaries prepared in accordance with Indian GAAP;
- (e) within 30 (thirty) days after the end of each financial quarter, internal quarterly income and cash flow statements for the Company and the Subsidiaries;
- (f) within 25 (twenty five) days after the end of each month, internal monthly MIS and marketing reports of the Company and the Subsidiaries in a format mutually agreed between the Promoters and the Investor prior to Closing.
- (g) any other material information including correspondence with the Company's and the Subsidiaries' auditors, material litigation, any relevant filings are made with regulatory agencies etc.; and
- (h) such other information and periodical reports as the Investor reasonably requires.
- 35.2. Upon the Investor giving the Company and/ or relevant Subsidiary at least 3 (three) days prior notice and at the cost of the Investor unless a specific material breach or a probable material breach is proposed to be investigated by the Investor, the Investor and its representatives may, during normal business days and hours, (i) inspect and examine and take copies of the Books and Records and Accounts kept by the Company, (ii) access/inspect all properties of the Company, including, but not limited to, the office premises of the Company; and (iii) discuss, consult and/or interview the Business, action plans, budgets and finances and other aspects related to the Company and the Subsidiaries with the Directors and/or the Key Management Team of the Company and the Subsidiaries. All information provided by the Company and/or its Subsidiaries under this Article 35 shall be deemed to be Confidential Information and shall be subject to confidentiality obligations under clause 28.1 of the Agreement.
- 35.3. The Company shall maintain a system of accounting established and administered in accordance with Indian GAAP, as appropriate and as applicable, consistently applied, and will set aside on its books all such proper reserves as shall be required by applicable Law.

#### XXXVII - REGISTRATION RIGHTS

# 36. **Registration Rights**

36.1. If any Equity Shares or other Equity Securities of the Company are listed or proposed to be listed on one or more stock exchanges overseas, then upon the request of the Investor, the Company shall take all such steps, do all such things, execute all such writings and make all regulatory applications and filings as may be required by Law for permitting or facilitating the unrestricted sale and distribution of the Equity Securities held by the Investor on such exchanges to the extent permissible by applicable Law, such that the Equity Securities held by the Investor are freely transferable on such stock exchanges ("Registration Rights").

- 36.2. The Investor shall be entitled to demand that (i) all or part of the Equity Securities held by the Investor be converted into American Depository Receipts or Global Depository Receipts as permissible under applicable Law; and (ii) the Company register the Equity Securities of the Company held by the Investor with appropriate and necessary regulatory authorities required in connection with such offering. Such registration shall be at the expense of the Company, to the extent permissible under Law. Such offerings will be subject to limitations recommended by an independent qualified advisor.
- 36.3. The Investor will be entitled to piggyback rights (to make an Offer for Sale simultaneously) in all primary offerings and all other secondary offerings of the Company in connection with the Registration Rights, and will, subject to applicable Law, pay such expenses incurred in all piggyback registrations and expenses toward any such offering pro rata to its participation.

#### XXXVIII - GENERAL MEETING

# 37. **General Meeting**

- 37.1. An annual general meeting of the Shareholders of the Company shall be held within six (6) months of the end of each financial year of the Company subject to any extension provided by the registrar of companies. Subject to the foregoing, the Board or the Parties may convene an extraordinary general meeting of the Shareholders of the Company whenever they deem appropriate.
- 37.2. At least Twenty-one (21) days prior written notice of every general meeting of members shall be given to all Shareholders whose names appear on the register of members of the Company. A general meeting may be called after giving shorter notice than that specified above if consent, in writing or by electronic mode, is accorded thereto—
  - (i) in the case of an annual general meeting, by not less than ninty-five per cent. of the members entitled to vote thereat; and
- (ii) in the case of any other general meeting, by members of the company majority in number of members entitled to vote and who represent not less than ninety-five per cent. of such part of the paid-up share capital of the company as gives a right to vote at the meeting; or
- 37.3. The notice of each general meeting of Shareholders shall include an agenda setting out the place, date and time and business proposed to be transacted at the meeting, together with copies of all relevant papers connected therewith and/or proposed to be placed before or tabled at the general meeting and not business shall be transacted at such meeting unless the same has been stated in the notice convening the meeting.
- 37.4. The Chairman of a general meeting of the Company shall not have any second or casting vote. The Chairman of the Board shall be the Chairman for all general meetings,

- unless the meeting is called by the Investor, in which case the Investor will appoint the Chairman for that meeting.
- 37.5. Any shareholder of the Company may appoint another Person as his proxy (and in case of a corporate shareholder, an authorized representative) to attend a meeting and vote thereat on such shareholder's behalf, provided that the power given to such proxy must be in writing.
- 37.6. Quorum. The quorum for a general Meeting of the Shareholders, shall be as provided under the Act and must include at least 1 (one) authorized representative of the Investor (unless waived by the Investor in writing) and an authorized representative of the Promoter, in order to constitute a valid quorum for the meeting. If on the date of the general meeting, a valid quorum is not present, the meeting shall automatically stand adjourned to the same day and time and at the same venue in the following week; provided that (A) no business or items not being part of the agenda of the original meeting shall be dealt with in such adjourned meeting; and (B) no business concerning any of the Reserved Matters shall be discussed or approved at such adjourned meeting unless such Reserved Matter has been approved in writing by the Investor and the Company has received such written approval prior to the meeting.

## 37.7. Voting Rights available to the holders of Series A CCPS.

- (a) With respect to voting rights exercised at any meeting of the Shareholders of the Company, the holders of Series A CCPS shall enjoy such voting rights available to the extent permissible pursuant to the Act, carry voting rights as if Series A CCPS have been fully converted into Equity Shares.
- (b) Each Series A CCPS shall entitle the holder to the number of votes equal to the number of Equity Shares into which such Series A CCPS could then be converted. To this effect, each Shareholder holding Shares with voting rights agrees that, if applicable Law does not permit any holder of Series A CCPS to exercise voting rights on all or any Shareholder matters submitted to the vote of the Shareholders of the Company (including the holders of Equity Shares) (the "Non-Voting Preference Shares"), then until the conversion of all such Non-Voting Preference Shares into Equity Shares, each Shareholder shall vote in accordance with the instructions of the holders of such Non-Voting Preference Shares at a general meeting of the Shareholders or provide proxies without instructions to the holders of the Non-Voting Preference Shares for the purposes of a general meeting of the Shareholders, in respect of such number of Equity Shares held by each of them such that a relevant percentage (the "Relevant Percentage") of the Equity Shares of the Company are voted in the manner required by the holders of the Non-Voting Preference Shares. For the purposes of this Article 37.7(b), the Relevant Percentage in relation to a holder of Non-Voting Preference Shares shall be equal to the percentage of Equity Shares in the Company that such Non-Voting Preference Shareholder would hold if such Non-Voting Preference Shareholder was to elect to convert its Series A CCPS into Equity Shares based on the then applicable Conversion Ratio for each Series A CCPS held by such holder. The obligation of the

Shareholders to vote their Shares as aforesaid shall be pro-rated in accordance with their inter se shareholding in the Company.

(c) Provisions of section 47 of the Act shall not apply to the Company and members shall vote in accordance with the provisions in these Articles.

## XXXIX - ITEMS OF BUSINESS REQUIRING CONSENT OF THE INVESTOR

#### 38. **Reserved Matters**

38.1. The Company shall, and each of the Shareholders shall, exercise or refrain from exercising all rights and powers available to it to procure that from the Closing Date, no resolution (whether of the Shareholders or the Directors, including committees of Board of the Company) or action constituting any of the matters set out in Article 38.6 (the "Reserved Matters") shall be passed or undertaken or occur with respect to the Company or any of the Subsidiaries (each a "Group Company") unless so approved in accordance with the provisions of this Article 38. Without prejudice to the foregoing, the Company shall, and each of the Shareholders shall, procure that resolutions passed in breach of this Article 38 (i.e. on any Reserved Matter on which the Investor Nominee Directors or any of the Investors have, pursuant to Articles 38.2, 38.3 and 38.4, not given their affirmative vote), shall not be effective. For the avoidance of doubt, it is clarified that the Promoters positive vote on all matters required to be undertaken by the Company and/or Subsidiaries shall be mandatory.

## 38.2. In respect of the Company:

- (a) a Reserved Matter which requires Board approval (under applicable Law, the Articles or any constitutive documents) shall not occur unless it has first been approved by the Board of the Company which shall include the affirmative vote of an Investor Nominee Director; and
- (b) a Reserved Matter which requires shareholder approval (under applicable Law, the Articles or any constitutive documents) shall not occur unless it has first been approved by the Shareholders which shall include the affirmative vote of the Investor.

## 38.3. In respect of a Group Company (other than the Company):

- (a) a Reserved Matter which requires Board approval (under applicable Law, the articles of association or any constitutive documents of such Group Company) shall not occur unless:
  - (i) in the case of a Group Company (in respect of which, the Investor has exercised its right to appoint directors), it has first been approved by the Board of such Group Company which shall include the affirmative vote of the director appointed by the Investor to the Board of such Group Company; and

- (ii) in the case of a Group Company (in respect of which the Investor has not exercised its right to appoint directors), it has first been approved by the Board of such Group Company in accordance with the following procedure:
  - (A) such Reserved Matter shall not occur unless it has first been approved by the Board of such Group Company which shall include the affirmative vote of the director appointed by the Company to the Board of such Group Company;
  - (B) for the purposes of this Article 38.3(a)(ii), the director appointed by the Company to the Board of such Group Company shall, subject to applicable Law, the articles of association or any constitutive documents of such Group Company, be instructed to vote in accordance with the prior decision taken by the Board of the Company in respect of a relevant Reserved Matter which shall, for the avoidance of doubt, include the affirmative vote of an Investor Nominee Director; and
  - (C) if, due to any reason whatsoever, the director appointed by the Company to the Board of such Group Company is unable to vote at the Board meeting of such Group Company in respect of the Reserved Matter in accordance with such instructions, such director shall insist (subject to applicable Law, the articles of association or any constitutive documents of such Group Company), that the Reserved Matter be discussed at a general meeting of such Group Company, in which case the procedure set out in Article 38.3(b) shall be adhered to.
- (b) a Reserved Matter which requires shareholder approval (under applicable Law, the articles of association or any constitutive documents of the relevant Group Company) shall not occur unless it has first been approved by the shareholders of such Group Company (whether or not the Investor has appointed directors to the Board of such Group Company) which shall include the affirmative vote of the Company. For the purposes of this Article 38.3(b), the Company shall vote in accordance with the prior decision taken by the Board of the Company in respect of the relevant Reserved Matter which shall, for the avoidance of doubt, include the affirmative vote of an Investor Nominee Director.
- 38.4. For the purposes of this Article 38, a decision to be taken by the Board of the Company may be taken by way of a circular written resolution of the Board in lieu of a physical Board meeting, such resolution to include the affirmative vote of Investor Nominee Directors.
- 38.5. If any other provision of these Articles conflicts with the provisions of this Article 38, the provisions of this Article 38 shall prevail and be given effect.

- 38.6. The following matters shall be the Reserved Matters in relation to the Company:
  - (a) Any change to the name and/ or registered office of the Company and/ or the Subsidiaries;
  - (b) Changes in the Articles and Memorandum of the Company and/ or the Subsidiaries;
  - (c) Any substantial diversification, modernization, modification or expansion of the Business either through the Company and its Subsidiaries and/ or any new entity set up for such purpose;
  - (d) Change in the shareholding pattern or capitalization structure of the Company and/ or the Subsidiaries, except to the extent permitted under the Transaction Documents;
  - (e) Issuance of new Equity Securities by the Company and/ or the Subsidiaries;
  - (f) Any change in terms of any Equity Securities by the Company and/ or the Subsidiaries that adversely affects the rights of the Investor;
  - (g) Issuance of any debt securities by the Company and/ or the Subsidiaries;
  - (h) Redemption, buyback (except as contemplated in the Agreement), reduction of capital or cancellation or reduction of any of the Equity Securities of the Company and/ or the Subsidiaries;
  - (i) Formulation, adoption of, or change of, ESOP including with respect to Company and/ or the Subsidiaries except to the extent specifically contemplated under the Transaction Documents;
  - (j) Creation of any Encumbrance on the securities of the Company and/ or the Subsidiaries or providing lenders with an option to acquire/ subscribe to Securities in the Company and/ or the Subsidiaries;
  - (k) Distribution or payment of dividends other than in compliance with the dividend policy of the Company or any changes in the dividend policy of the Company;
  - (l) Amending or terminating or agreeing to amend or terminate any material contracts or any arrangements, having a value in excess of INR 2,00,00,000 (Rupees Two Crores only);
  - (m) Any Related Party transactions by the Company and/ or the Subsidiaries except for those undertaken in compliance with provisions of Article 41.1 and for the avoidance of doubt, including but not limited to loans, borrowings and investments in any form whatsoever, but specifically excluding any business

- transactions inter-se the Company, Reva, RSS, CES-FZE, BWTT and CES Mexico:
- (n) Any transactions with Affiliates of Promoters and/ or the Key Management Team, excluding payments of salaries under the respective employment agreements;
- (o) Making of a public offering of the shares of the Company, except as contemplated by the Agreement, including the size and timing of the issue, price of the shares and appointment of the investment bankers, legal advisors and any and all other intermediaries in connection with such a public offering;
- (p) Any borrowing, indebtedness, issuance, incurrence or guarantee by the Company or its Subsidiaries of any indebtedness or creation of Encumbrance in relation to assets of the Company, other than as contemplated in the Business Plan and in excess of an aggregate of INR 2,00,00,000 (Rupees two crores only), in any Financial Year;
- (q) Prepayment or repayment of any loans or borrowings not in accordance with the ordinary repayment schedule applicable with respect to such loans or borrowings;
- (r) Formation of subsidiaries or entering into joint ventures or any investments/ divestments by the Company and/ or any Subsidiary;
- (s) Formulation and approval of the Business Plan of the Company and/ or the Subsidiaries;
- (t) Any change to the Business Plan in excess of 5% of the relevant parameter or line item in the Business Plan;
- (u) Entering into contracts or any other arrangements which result in exclusivity or non-compete restriction or that restrict the ability of the Company to carry on Business in any part of the world;
- (v) Unless provided for in the Business Plan, capital expenditures in excess of INR 1,00,00,000 (Rupees One Crore only) per Financial Year;
- (w) Unless provided for in the Business Plan, disposal of Assets of the Company and/ or the Subsidiaries whose aggregate value exceeds 5% (five percent) of Company's / relevant Subsidiary's audited net asset value;
- (x) Any change in the number of directors/ members of the Boards of the Company and/ or the Subsidiaries other than as specifically contemplated in the Agreement;
- (y) Removal or appointment of independent Directors of the Company and/ or the Subsidiaries, if any;

- (z) Any change in the terms of appointment of any member of the Key Management Team or change in remuneration of any member of the Key Management Team by more than 15% (Fifteen percent) of his/ her prevailing cost to company;
- (aa) Appointment and dismissal of an employee (including a member of the Key Management Team) of the Company and/ or the Subsidiaries having CTC of more than INR 40,00,000 (Rupees forty lakhs only). "CTC", for the purposes of the preceding sub-clauses, shall mean the cost to company and include salary and any monetary or non-monetary remuneration/consideration (non-monetary remuneration being determined on the basis of the fair market value) received by the employee of the Company or the Subsidiaries;
- (bb) Change in the accounting policies employed by the Company except when so required by applicable Law, including, but not limited to, any change in the Financial Year of the Company;
- (cc) Appointment and removal of the internal auditor and statutory auditor of the Company and/ or the Subsidiaries except for reappointment of the existing internal auditor and/ or statutory auditor;
- (dd) Initiation or determination of any individual litigation or arbitration of a value in excess of INR 50,00,000 (Rupees fifty lakhs only) and cumulative value in excess of INR 1,00,00,000 (Rupees One Crore only);
- (ee) Any filings, actions or resolutions for winding up, liquidation, bankruptcy or insolvency of any of the Company and/ or the Subsidiaries;
- (ff) Any reorganization, consolidation, spin-off, merger, acquisition, Liquidity Event or other similar transaction by the Company and/ or the Subsidiaries;
- (gg) Surrendering any tax benefits that may be available to the Company;
- (hh) Selling, Encumbering, Transferring, licensing, alienating or otherwise disposing of any of the Company's/ Subsidiaries' Intellectual Property, or interests therein or contracting to do so, other than in the ordinary course of business or as contemplated by the Business Plan;
- (ii) Any change (including for the avoidance of doubt, any alteration, modification, deletion and/or substitution) to the visual representation and format of the existing Intellectual Property owned and/ or licensed by the Company, other than in the ordinary course of business or as contemplated by the Business Plan;
- (jj) Creation of any Encumbrance on any of the property or assets of the Company and/ or the Subsidiaries (except in the ordinary course of business or as

contemplated by the Business Plan and, in any event, not exceeding in the aggregate, INR 2,00,00,000 (Rupees two crores only) during a Financial Year);

(kk) Entering into any binding agreement to take any of the foregoing actions.

#### XL - DIRECTORS

#### 39. **Directors**

- 39.1. The Investor shall have the right to nominate and appoint (a) 2 (two) nominee Directors on the Board of Directors of the Company, provided however that such right shall be exercisable so long as the shareholding of the Investor is above or equal to, but does not fall below 10.00% (ten per cent) of the total paid-up share capital of the Company, on a fully diluted basis, and (b) 1 (one) nominee Director on the Board of Directors of the Company, provided however that such right shall be exercisable so long as the shareholding of the Investor is less 10% (ten per cent), but does not fall below 5.00% (five per cent) of the total paid-up share capital of the Company, on a fully diluted basis (such nominee directors hereinafter referred to as "Investor Nominee Director(s)").
- 39.2. Notwithstanding anything to the contrary that may be contained in these Articles, it is hereby clarified that the Investor's right to nominate the Investor Nominee Directors shall terminate on and from Listing Date.
- 39.3. The Promoters shall have the right to nominate and appoint (a) 3 (three) nominee Directors on the Board of Directors of the Company, provided however that such right shall be exercisable so long as the aggregate shareholding of the Promoters is above or equal to, but does not fall below 10.00% (ten per cent) of the total paid-up share capital of the Company, on a fully diluted basis, and (b) 1 (one) nominee Director on the Board of Directors of the Company, provided however that such right shall be exercisable so long as the aggregate shareholding of the Promoters is less 10% (ten per cent), but does not fall below 5.00% (five per cent) of the total paid-up share capital of the Company, on a fully diluted basis (such nominee directors hereinafter referred to as "Promoter Nominee Director(s)").
- 39.4. Notwithstanding anything to the contrary that may be contained in these Articles, it is hereby clarified that the Promoters' right to nominate the Promoter Nominee Directors shall terminate on and from the Listing Date.

The Board shall also contain such number of independent directors as may be required under the regulations prescribed by SEBI and other applicable law."

- 39.5. Qualification Shares. A Director shall not be required to hold any qualification shares.
- 39.6. <u>Alternate Director</u>. Any Party having the right to nominate a Director shall have the right to nominate an alternate Director to the nominee Director in accordance with the

provisions of the Act. The Company and the Shareholders shall take all steps necessary to secure the appointment of the alternate Director. The alternate directors so appointed shall be entitled to attend the meetings of the Boards of the Company and vote in the event the nominee Director is unable to attend any meeting of the Boards of the Company. Any such alternate director shall be considered for the constitution of the quorum and shall be entitled to attend and vote at such meetings in place of the original director and generally to perform all functions of the original director in his absence.

#### 39.7. Board Committees.

- (a) The Boards of the Company and/ or the Subsidiaries shall resolve to establish committees, which will have delegated responsibility for dealing with specific functions otherwise carried out by such Boards, including a compensation committee and audit committee.
- (b) The Investor shall have the right to pro-rata representation, as the Board(s) of the Company, on any committee formed by such Board.
- (c) <u>Audit Committee.</u> The audit committee shall be constituted in accordance with the provisions of the Act and shall carry out such functions as directed by the Board of the Company.
- (d) <u>Compensation Committee.</u> The compensation committee shall be constituted in accordance with the provisions of the Act and shall bear responsibility for reviewing and approving the compensation of the Key Management Team and other senior employees of the Company.
- 39.8. The nominee Directors shall be entitled to receive all notices, agenda (and all information and documents circulated to the Board(s) of the Company and the Shareholders in connection with meetings of the Board(s) of the Company and to attend all such meetings, shareholders meetings and meetings of any committees of the Board(s) and the shareholders of the Company.
- 39.9. <u>Vacancies.</u> If any Director resigns, vacates or is removed from office before his term expires, the resulting casual vacancy may be filled by a nominee of the Shareholder who originally nominated the Director vacating office, but any person so nominated, shall retain his office only so long as the vacating Director would have retained the same, if no vacancy had occurred.

# 39.10. Liability of Investor Nominee Director.

- (a) <u>Non-Executive Director</u>. The Investor Nominee Director will be a non-executive Director in accordance with the Act.
- (b) Not to be responsible for day to day management. The Promoters and the Company expressly agree and undertake that the Investor Nominee Director, so long as he is acting as non-executive director, shall not be in charge of, or responsible for the day-to-day management of the Company and shall not be

deemed to be an "officer in default" as the term is defined in the Companies Act, 2013, and shall accordingly not be liable for any default or failure of the Company and/ or the Subsidiaries in complying with the provisions of any applicable law, subject to applicable law, including the Companies Act, 2013 and any regulations as may be notified by the Securities and Exchange Board of India, from time to time. The Promoters and Company expressly agree that the Investor Nominee Director shall not be identified as an occupier of any premises used by the Company or an employer of the employees of the Company

- 39.11. Subject to provisions of section 149 of the Companies Act, 2013 and until otherwise determined by the Company in General meeting, the number of directors shall be not less than two and not more than fifteen.
- 39.12. A director shall not be disqualified by reason of his holding any other office or place of profit under the company other than those mentioned in Section 188 of the Companies Act, 2013 in conjunction with his office of director, except that of auditor and may be appointed thereto upon such terms as to remuneration, tenure of office, as the case may, with the previous consent of the Company, accorded by a special resolution.
- 39.13. Subject to the provisions of Section 188 of the Act and Reserved Matters, the Directors shall not be disqualified from contracting with the Company either as vendors, purchasers or otherwise, nor shall any such contract or arrangement entered into by or on behalf of the Company with any company or partnership in which any director so contracting being member or so interested, be liable to account to the Company for any profit realized by any such contract or arrangement by reason of such director holding that office or of the fiduciary relation thereby established but the nature of his interest must be disclosed by him at the meeting of the Board of Directors at which the contract or arrangement is determined on, if the interest then exists, or in any other case at the first meeting of the Board of directors after the acquisition of the interest. Provided, nevertheless that such Director shall be entitled to present at the meeting and to vote as Director in respect of the contract or arrangement in which he is so interested as aforesaid and if he does so, vote shall be counted. No general notice and no renewal thereof shall be of effect unless either it is given at a meeting of a Board or the director concerned takes reasonable steps to secure that it is brought up and read at the first meeting of the Board after it is given.
- 39.14. The payment of sitting fees shall be governed by the policy of the Company in this regard as may be in force from time to time.
- 39.15. The Board of directors may fix such remuneration to be paid to a Director of the Company as they think fit, subject to section 188 of the Act, if applicable.

- 39.16. In addition to the remuneration payable to them in pursuance of these Articles, the Directors may be paid all traveling, hotel and other expenses property incurred by them connection of the business of the Company as the Board may determine.
- 39.17. No director shall retire at any general meeting of the Company and subject to the provisions of the Act and these Articles, a Director shall continue to act as such until death, resignation and/or removal, as the case may be, unless tenure of employment is fixed by the Board in writing.
- 39.18. The Board of directors shall have powers to appoint the Directors nominated by any lending financial institution. Such director shall not be required to retire by rotation until the arrangement with the said financial institution continues.
- 39.19. The Board shall also have power to fill a casual vacancy in the Board. Such casual vacancy shall be filled up by the Board of Directors at a meeting of the Board. Any person so appointed shall hold office only upto the date upto which the Director in whose place he is appointed would have held office if it had not been vacated as aforesaid, but he shall then be eligible for election.
- 39.20. **Managing Director.** Subject to Article 38 (Reserved Matters) and the Act, the Board may subject to provisions of the Act appoint any person to be the Managing Director of the Company and may fix the remuneration to be paid to him and to determine to power exercisable and duties to be performed by such Managing Director. An individual may be appointed as the Managing Director and Chairman of the Company at the same time.
- 39.21. **Company Secretary.** Subject to Article 38 (Reserved Matters) a company secretary may be appointed upon such terms and conditions as may be fixed by the Board.
- 39.22. **Additional Director**. Subject to Article 38 (Reserved Matters) an additional director may be appointed by the Board, upon such terms and conditions as may be fixed by the Board.

## XLI - PROCEEDINGS OF BOARD

## 40. **Proceedings of Board**

- 40.1. <u>Number of Board Meetings</u>. The Board shall meet at least four (4) times in every calendar year *provided that*, not more than 120 (one hundred and twenty) days shall intervene between 2 (two) consecutive Board meetings. Subject to applicable Law, a Board meeting may also be held by video conferencing and/or any other permitted means remote participation.
- 40.2. <u>Convening meetings of the Board</u>. Any Director may, and the secretary of Company, if so appointed, shall on the requisition of a Director, summon a meeting of the Board, in accordance with the notice and other requirements set out below in Article 21.3 (Notice for Board Meetings).

40.3. <u>Notice for Board Meetings</u>. A Board meeting may be called by a Director and 14 (fourteen) days' written notice of each meeting of the Board or a committee thereof shall be given to each of the Directors at the address notified from time to time by each of them, in writing to the Company, whether in India or abroad, *provided that* a meeting may be convened by a shorter notice with written consent (which may be signified by letter, facsimile or e-mail with receipt acknowledged) of all the Directors..

### 40.4. Contents of the Notice.

- (a) Every notice of a Board meeting of the Company and/ or the Subsidiaries shall set forth in full and sufficient detail each item of the business to be transacted thereat, and no item or business shall be transacted at such Board meeting, unless the same has been stated in full and in sufficient detail in the notice convening the meeting, except as otherwise consented to by all the Directors, or their respective alternate directors.
- (b) The draft resolutions and other documents for all matters to be considered at the Board meetings of the Company and/ or the Subsidiaries must be furnished to all the Directors along with the notice for such Board meeting.
- Quorum for Board meeting. Apart from the requirements of the Act, the presence of at 40.5. least 1 (one) Investor Nominee Director, if any, and 1 (one) Promoter Nominee Director, personally or by way of video conference throughout the meetings of the Boards of the Company shall be required to constitute valid quorum. If the Investor Nominee Director, if any, or the Promoter Nominee Director, is not present at any meeting of the Board of the Company ("Initial Meeting") within 30 (thirty) minutes of the scheduled time, the meeting shall be adjourned to same day and time of the subsequent week, unless otherwise agreed by all the Directors, and if that day is not a Business Day to the immediately succeeding Business Day ("First Adjourned Meeting"). If an Investor Nominee Director or the Promoter Nominee Director is not present at such First Adjourned Meeting, within 30 (thirty) minutes of the scheduled time, then the meeting shall again be adjourned ("Second Adjourned Meeting") to the same day and time of the subsequent week, unless otherwise agreed by all the Directors, and if that day is not a Business Day to the immediately succeeding Business Day. If an Investor Nominee Director or the Promoter Nominee Director is not present at such Second Adjourned Meeting, the Directors present shall constitute valid quorum, provided that at least 2 (two) Directors are present. The agenda for the Initial Meeting shall be the agenda for the Adjourned Meeting, and material matters which are not specifically defined and stated in the agenda for the Initial Meeting shall in no event be taken up for discussion or approved at the Adjourned Meeting, it being understood and agreed by the Parties that any matters in respect of Reserved Matters shall in no event be discussed or approved by the Board of the Company and/ or the Subsidiaries without the presence of an Investor Nominee Director (and the approvals required by Article 38 (Reserved Matters)).
- 40.6. <u>Decisions of the Board.</u> Subject to Article 38 (*Reserved Matters*), a decision shall be said to have been made and/or a resolution passed at a meeting of the Board of the Company only if at a validly constituted meeting, such decisions are approved of by a

majority of the Directors, present and voting at such Board meeting. In the event there is a vacancy on the Board and an individual has been designated to fill such vacancy, the first order of business shall be to fill such vacancy.

- 40.7. <u>Chairman.</u> Each meeting of the Board shall be chaired by the Chairman nominated by the Board. The Chairman shall not have a casting vote, to resolve a deadlock in voting. In the absence of the Chairman at any meeting, the Board shall elect one of their members to chair the meeting in question.
- 40.8. Resolution by circulation or written consent. Subject to Article 38 (Reserved Matters) and except for those actions required by the Act to be determined at a meeting of the Board, all decisions of the Board may be taken by circular resolution. No resolution shall be deemed to have been duly passed by a Board of the Company and/ or the Subsidiaries, or a committee thereof by circulation or written consent, unless the resolution has been circulated in draft, together with the information required to make a fully-informed good faith decision with respect to such resolution, to all Directors, or to all members of the relevant committee, at their usual address (whether in India or abroad) or through electronic means, and has been approved (subject to the Article 38 (Reserved Matters)), by majority of the Directors.
- 40.9. <u>Removal/Resignation of Directors</u>. The Company and the Shareholders shall not remove any nominee Director. Each Party entitled to nominate a Director shall have the right to require the removal of such nominee at any time and shall be entitled, to nominate another Person as the nominee Director in place of the Person removed. In the event of the resignation, retirement or vacation of office of the Director nominated by any Party, such Party shall be entitled to nominate another Person as Director in place of such Director and the Shareholders shall exercise their rights in such manner so as to cause the appointment of such Person as nominee Director to the Board of the Company.

### XLII - COVENANTS BY THE PROMOTERS AND THE COMPANY

### 41. Covenants by the Promoters and the Company

41.1. <u>Related Party Transactions</u>. All the transactions, contracts, arrangements or understandings entered into or proposed to be entered into between the Company and its Related Parties shall be (i) on commercially justifiable terms and at an arms-length arrangement; and (ii) subject to the approval of the Investor as set forth in Article 38 (*Reserved Matters*).

### 41.2. **No Solicitation**.

On and from the Closing Date and until the Investor holds any Equity Securities in the Company ("**Limited Period**"), Promoters shall not and shall procure that their Affiliates shall not, whether directly or indirectly, by themselves or together with or through any Person, in any manner whatsoever, (whether in their own capacity or in conjunction with or on behalf of any Person) do or undertake or attempt to do or undertake any of the following activities:

- (a) propose to, canvass, solicit, entice away or attempt to canvass, solicit or entice away from the Company and/ or the Subsidiaries any of their customers, clients, franchisees, vendors, lessor, representative, agent, franchisees, business associates and/or employees ("Restricted Persons"), whether or not such Person would commit a breach of contract by reason of such act and or assist, influence, encourage or induce any of the foregoing action in any manner whatsoever;
- (b) provide any know-how or technical assistance to any Person (other than the Company and the Subsidiaries) in relation to the Business.
- 41.3. <u>Non-compete</u>. On and from the Closing Date and during the Limited Period, Promoters shall not and shall procure that their Affiliates shall not, whether directly or indirectly, by themselves or together with or through any Person, in any manner whatsoever (whether in their own capacity or in conjunction with or on behalf of any Person), do or undertake or attempt to do or undertake any of the following activities:
  - (a) commence, establish, promote, finance, engage in, carry on, join in, participate in, manage, advise, operate, control, conduct, own, invest in or have an interest in any business, venture or Person which is competing or in competition with the Business of the Company and/ or the Subsidiaries in any manner whatsoever, including initiating any new activities or expansion related to the Business or any other business undertaken by the Company and/ or the Subsidiaries, through any Person, including any Person in which they have any interest. Nothing contained in this Article 41.3 shall apply to any investment made by the Promoters and/or their Affiliates in any listed business or venture or company provided that: (i) such investment is purely a financial investment and in no manner whatsoever can be deemed to be a strategic investment or one seeking or providing control over the business, venture or company; and (ii) such investment does not result in the Promoters and/or their Affiliates holding in aggregate, more than 5% (five percent) beneficial interest in such business competing with the whole or any part of the Business of the Company and/ or the Subsidiaries:
  - (b) provide financial assistance, or technical, managerial or any other services (whether for consideration or free of charge), in any manner to, or be Affiliates with, any Competitor;
  - (c) through Rochem Green Energy Private Limited ("RGE"), commence any other business that competes with the Business (including any new activities or expansion related to the Business), expand its waste-to-energy concession business and/ or waste management concession business or devote more than 3 (three) Business Days in a calendar month towards the business of RGE;
  - (d) enter into any business, which is similar or identical to, or that competes with, the Business of the Company and/ or the Subsidiaries, or any business that the

Company or its Subsidiaries may undertake, in any manner, anywhere in the world; and

- (e) for its own account or as agent, consultant, or shareholder of, or as owner of any equity or economic interest in, any other Person, engage or attempt to engage or assist any Competitor or other Person to engage in any business activity that is competitive with, or similar or identical to, the Business of the Company and/ or the Subsidiaries, or any business that the Company or its Subsidiaries may undertake, anywhere in the world.
- 41.4. <u>Focus on Business and Activities of the Company</u>. The Company and the Subsidiaries shall undertake the Business and such other businesses as may be approved by the Investor in accordance with Article 38 (Reserved Matters).
- 41.5. <u>Management.</u> The Promoters and the Key Management Team shall be in charge of the day-to-day management of the Company and the Subsidiaries.
- 41.6. Employees and Resources of the Company. Employees and resources of the Company and the Subsidiaries shall at all times be employed/ utilized only towards the Business and activities of the Company and/ or the Subsidiaries and shall at no instance be deployed or diverted towards carrying out any work/ activities for any Affiliate of the Promoters. The Company shall further enter into appropriate employment agreements with all the present and future members of its Key Management Team covering confidentiality, non-compete, non-solicitation, exclusivity and IP assignment obligations.
- 41.7. Anti-corruption. The Company and the Subsidiaries shall not violate the United States Foreign Corrupt Practices Act, 1977, the U.K. Bribery Act, 2010, Prevention of Corruption Act, 1988, Prevention of Money Laundering Act, 2002 or any other applicable anti-bribery or anti- corruption law, or promise, authorize or make any payment to, or otherwise contribute any item of value to, directly or indirectly, any officer, employee or any other person acting in an official capacity for any Government Entity, to any political party or official thereof or to any candidate for political office (individually and collectively, a "Government Official") or to any person under circumstances where such Company and/ or the Subsidiary knows or is aware of a high probability that all or a portion of such money or thing of value would be offered, given or promised, directly or indirectly, to any Government Official, for the purpose of:
  - (a) influencing any act or decision of such Government Official in his official capacity; (ii) inducing such Government Official to do or omit to do any act in relation to his lawful duty; (iii) securing any improper advantage; or (iv) inducing such Government Official to influence or affect any act or decision of any Government Entity.
  - (b) assisting the Company and/ or the Subsidiaries in obtaining or retaining business for or with, or directing business to the Company or the Subsidiaries.

The Company and the Subsidiaries shall maintain systems of internal accounting controls to ensure compliance with applicable anti-bribery or anti-corruption Laws. For the purpose of this Article, "Government Entity" means any government or any department, agency or instrumentality thereof, including, but not limited to, any entity or enterprise owned or controlled by a government, or a public international organization.

- 41.8. Action Plan. The Company and the Subsidiaries shall implement and comply with the Action Plan and undertake the Business of the Company and/ or the Subsidiaries in compliance with the Applicable S&E Law. The compliance with the Action Plan and the Applicable S&E Law shall be reviewed by a Third Party service provider appointed by the Investor on an annual basis. Based on the findings of such Third Party service provider, the Action Plan, as presently set out in Schedule 5 of the Agreement, shall be revised/ modified mutually by the Investor, the Company and the Promoters, if deemed necessary by such Third Party service provider, and the Company shall implement and comply with such revised/modified Action Plan, as the case may be, from time to time.
- 41.9. Transactions with Prohibited Persons. The Company and the Subsidiaries shall not:
  - (a) enter into any transaction or engage in any activity prohibited by any resolution of the United Nations Security Council under Chapter VII of the United Nations Charter.
  - (b) conduct business or enter into any transaction with, or transmit any funds through, a Shell Bank.

If any of the Companies, Subsidiaries and/or the Promoters become aware of any violation of Article 41.7 or this Article 41.9, such Person shall promptly within 7 (seven) Business Days from the date of becoming aware of such violation, notify the Investor in writing, and the Company and such other party shall cooperate in good faith with the Investor and its representatives in determining whether such a violation has occurred, and shall respond promptly and in reasonable detail to any notice from the Investor, and shall furnish documentary support for such response upon the Investor's request.

- 41.10. <u>Insurance</u>. The Company and the Subsidiaries shall obtain and maintain at all times (i) insurance policies customarily held by companies of the similar value/ size or similar business as the Company and the Subsidiaries (as applicable); (ii) where applicable, insurance policies which are in accordance with any insurance coverage requirements set out in a contract to which it is a party (or by which it is bound), including the directors' and officers' liability insurance as required to be obtained under the Share Subscription Agreement.
- 41.11. <u>Business Plan</u>. For each Financial Year, the Promoters along with the Key Management Team shall prepare and provide to the Investor at least 30 (thirty) days prior to the end of the previous Financial Year (i) an annual business plan, including, the profit and loss statement, balance sheet and cash flow statement, outlining expectations on the Company's as well as the Subsidiaries' financial performance and specifying, amongst

other things, operational milestones, revenues originating from various existing and new segments, associated costs, capital expenditures, required financing, corporate overheads, labor costs, general and administrative expenses, production particulars, etc. ("**Draft Business Plan**") in a form satisfactory to the Investor and the Promoters and present it to the Board of the Company for approval. The form of the Draft Business Plan is attached in **SCHEDULE 6**. Upon the Board of the Company approving the Draft Business Plan for a Financial Year ("**Business Plan**"), a copy of the same shall be provided to the Investor and acknowledged by the Investor, and the Company and the Promoters shall adhere to the Business Plan.

### 41.12. OFAC.

- The Company shall not directly or indirectly use the Subscription Amount or (a) lend, contribute or otherwise make available the Subscription Amount to any subsidiary, joint venture partner or other person for the purpose of funding or facilitating any activities or business of or with any person towards any sales or operations in Cuba, Iran, Libya, Syria, Sudan, the Democratic People's Republic of Korea, Myanmar or any other country sanctioned by the Office of Foreign Assets Control of the U.S. Department of Treasury ("OFAC") or for the purpose of funding any operations or financing any investments in, or make any payments to, any person targeted by or subject to any sanctions administered by the OFAC, or by the U.S. Department of State, or any sanctions imposed by the European Union (including under Council Regulation (EC) No. 194/2008), the United Nations Security Council, Her Majesty's Treasury or any other relevant governmental entity and any activities sanctionable under the Comprehensive Iran Sanctions, Accountability, and Divestment Act of 2010, as amended or the Iran Sanctions Act, as amended (collectively, the "Sanctions").
- (b) The Company will use the Subscription Amount in accordance with the provisions of clause 10 of the SSA. The use of Subscription Amount will be in compliance with and will not result in the breach by the Company, the Subsidiaries, the Promoters, any officer, employee, director, agent, affiliate or person acting on behalf of the Company and/ or the Subsidiaries of the Sanctions; and the Company further covenants not to engage, directly or indirectly, in any other activities that would result in a violation of Sanctions by any person, including any person participating in the transaction.
- 41.13. <u>Greenhouse Gas Audit.</u> The Investor may require the Company and/ or the Subsidiaries to undergo a greenhouse gas audit annually, by a Third Party agency acceptable to the Investor. The Company shall, and the Promoters shall ensure that the Company and the Subsidiaries shall extend full cooperation and provide all necessary information and documents required for the conduct of such audit. All costs and expenses in relation to the conduct of such annual greenhouse gas audit shall be borne by the Company.
- 41.14. The Company shall, and the Promoters shall ensure that the Company and the Subsidiaries duly pay the requisite stamp duty on all agreements entered into by the

Company and the Subsidiaries, if required and other than in the ordinary course of business.

#### XLIII - ESOP

#### 42. **ESOP**

The Investor and Promoters shall as soon as reasonably practicable after the Closing Date, jointly formulate an appropriate ESOP plan for the employees of the Company, which shall not result in the dilution of the Investor's shareholding, up to 3% (three percent) of the pre-money share capital of the Tranche I Investment ("Agreed ESOP"). The Company shall and the Promoters shall undertake all procedures, including but not limited to approving the necessary Board and Shareholder resolutions of the Company, creating a trust and transferring/ issuing the requisite number of Equity Shares to such Agreed ESOP trust to give effect to this covenant. Any issuance of Equity Securities to such Agreed ESOP trust beyond 3% (three percent) of the pre-money share capital of the Tranche I Investment will result in the dilution of all Shareholders.

### **XLIV - EXIT**

#### 43. **Exit**

43.1. Within a period of 54 (fifty four) months from the Closing Date, (the "QIPO Period"), the Company and the Promoters shall utilize their best endeavours to provide the Investor an exit through the undertaking of a QIPO, in accordance with Article 43.2 below. If the Company is not able to successfully complete a QIPO or IPO within the QIPO Period, then the Company shall be obligated to provide alternative exit options to the Investor after the expiry of the QIPO Period, in the manner provided in this Article 43.

#### 43.2. **QIPO**.

- (a) The Company shall determine the following matters in connection with the QIPO, which decision shall not be undertaken without obtaining the prior written consent of the Investor:
  - (i) whether the public offering shall be by a fresh issue of Equity Shares by the Company and/ or an offering of Equity Shares for sale by the Shareholders;
  - (ii) the quantum of Equity Shares comprised in the issue
  - (iii) the appointment of lead managers, merchant bankers, bankers, registrars, financial advisors, issue managers and other intermediaries.
- (b) In the event of an Offer for Sale, the Investor will be entitled to offer some or all of its shareholding for sale in the QIPO, subject to applicable law.

- (c) In any QIPO undertaken by the Company, the Promoter shall offer such number of Equity Securities as may be required to fulfill the lock-in requirements under applicable Law.
- (d) Except as otherwise directed by SEBI, for the purposes of a QIPO and subject to applicable Law, the Investor shall not be considered as a "promoter" of the Company and therefore the Equity Securities held by the Investor (including any Equity Shares held by it pursuant to conversion prior to a QIPO or IPO) shall not be subject to any lock-in conditions applicable to promoters, for and after the IPO.
- (e) The Parties jointly and severally undertake to do the following, in connection with the QIPO:
  - (i) The Promoters and the Investor undertake to exercise their voting rights (at the meetings of the Board of the Company and the Shareholders), and to cause the Board of the Company to take all steps necessary for the Company to undertake a QIPO, including but not limited to, preparing and signing the relevant offer documents, providing all necessary information and documents necessary for preparing the offer document, obtaining all Consents and doing such further reasonable acts or deeds as may be necessary for undertaking a QIPO.
  - (ii) Ensure that the total offer of Equity Shares to the public shall constitute not less than such percentage (as prescribed under the applicable Law) of the total post issue paid-up share capital of the Company to comply with the listing requirements of the concerned Stock Exchanges and the concerned regulatory authority.
  - (iii) Provide all material information that is necessary to file the prospectus and other documents in relation to the QIPO, and ensure compliance with all Law including the Act, the SEBI Regulations, the listing agreement, etc.
  - (iv) All cost, fees and expenses in respect of the IPO will be borne by the Company and/ or the Selling Shareholders in the IPO, in the manner agreed in the offer agreement which will be executed in relation thereto and in accordance with applicable law.
- (f) The provisions of this Article 43.2 shall apply to any IPO of the Company.

## 43.3. **Buy-Back and Put Option**.

(a) If the Company has been unable to complete a QIPO within the QIPO Period, then at any time within the Exit Period, the Investor shall have, at its own discretion, the right to require the Company to and/ or the Promoters to exercise any or a combination of the following rights, by issuing a notice to the

Company and the Promoters, stating the right which it intends to exercise ("**Option Notice**") and the details of the Equity Securities with respect to which it intends to exercise its rights under this Article 43.3.

## (b) **Put Option**.

- (i) The Investor shall have the right to require the Promoters to purchase all the Investor Shares by delivering the Option Notice ("Put Option"). The Company and the Promoters shall undertake all procedures and obtain all Consents necessary to effect the purchase by the Promoters of all the Investor Shares subject to the Option Notice, including obtaining the necessary Board and Shareholder resolutions of the Company. The price to be paid by the Promoters to the Investor shall be, subject to applicable Law, the higher of (i) the Fair Market Value of the Investor Shares; or (ii) 2 (two) times the aggregate of the Investment Amount including any cash/ accrual payments of interest or coupon already paid, on an INR basis ("Put Option Price"), provided that any Tax payable by the Investor on receipt of any consideration shall be borne entirely by the Investor.
- (ii) The purchase of the Investor Shares by the Promoters shall be completed within 30 (thirty) Business Days from the date of delivery of the Put Notice.

## 43.4. **Drag Right**.

- (a) At any time during the Exit Period or thereafter, the Investor shall, by delivering the Option Notice, have the right to require the Company to appoint a reputed investment banker, as may be acceptable to the Investor, to facilitate the sale of the Investor Shares to a Third Party (including a Competitor) at a price and terms acceptable to the Investor. The fees and other expenses in relation to the appointment of the investment banker and the transaction shall be borne by the Company, provided that any Tax payable by the Investor for the sale of Investor Sale Securities (defined herein below) on receipt of any consideration shall be borne entirely by the Investor.
- (b) The Investor shall have the right to Transfer (i) all or any of the Investor Shares in case of a Transfer only by the Investor; and (ii) all (and not less than all) Investor Shares in case of exercise of rights under Article 43.4(c) below (in each case the "Investor Sale Securities") to such Third Party. The Company shall, and the Promoters shall cause the Company to undertake all procedures and obtain all Consents necessary to effect the sale and purchase by the Third Party of the Investor Sale Securities.
- (c) If the Third Party requires any additional shareholding to be Transferred to consummate the transaction, the Investor shall have the right to require the Promoters to offer such number of Equity Securities from their shareholding,

at a price and on terms and conditions no less favourable than those offered to the Investor ("Drag Securities") as may be required to give effect to the Drag Right of the Investor (as defined below) and the Promoter shall, upon request by the Investor, be obligated to Transfer the Drag Securities to the Third Party, including a Competitor (the "Drag Purchaser"), along with the Investor Sale Securities. Provided however that if as a result of the Drag Right, the aggregate shareholding of the Promoters falls below 20% (twenty percent) on a Fully Diluted Basis, the Promoters shall have a right to require the Investor to ensure that the Drag Purchaser acquires all and not less than all Equity Securities held by the Promoters along with the Investor Sale Securities. This right of the Investor to require the Promoters to sell all or some of their Equity Securities to the Drag Purchaser shall be referred to as the "Drag Right" and shall be exercised in the manner set forth hereinafter.

- In the event that the price offered by the Drag Purchaser for the Investor Shares is less than two (2) times the aggregate of the Investment Amount including any cash/accrual payments of interest or coupon already paid, on an INR basis, then the Promoters shall have the right to purchase the Investor Shares at such price offered by the Drag Purchaser. Within 75 (seventy-five) calendar days from the date of price offered by the Drag Purchaser, the Promoters shall issue a written notice to the Investor of them exercising their right to <u>purchase</u> the Investor Shares at the price offered by the Drag Purchaser along with reasonable documentary evidence to display availability of, or accessibility to, requisite funds with the Promoters for the purchase of Investor Shares. The Promoters shall purchase the Investor Shares at such price offered by the Drag Purchaser within 45 (forty-five) calendar days from the date of issuance of the written notice by the Promoters to the Investor as aforementioned.
- (e) If the Investor exercises the Drag Right, it shall issue a written notice ("Drag Notice") to the Promoters ("Dragged Sellers") calling upon them to Transfer the Drag Securities on date specified therein within 60 (sixty) days from the date of the Drag Notice (the "Drag Completion Date"). Subject to Article 43.4(c) above, the Dragged Sellers shall be bound and obligated to Transfer the Drag Securities specified in the Drag Notice to the Drag Purchaser simultaneously with a Transfer of the Investor Sale Securities on the Drag Completion Date. The distribution of the proceeds from the sale pursuant to the Drag Right shall be subject to the provisions of Article 44 below.
- (f) The Parties hereby covenant to take all steps necessary to give effect to the provisions of this Article 43.4 including the passing of all necessary resolutions and obtaining all necessary Consents.
- 43.5. The Parties agree and undertake that the obligation of the Company and the Promoters to provide an exit to the Investor in accordance with Article 43.4 above shall not be limited to the Exit Period.

# **XLV - LIQUIDATION PREFERENCE**

## 44. Liquidation Preference

- 44.1. In case of a Liquidity Event or a Liquidation Event, the mechanism for distribution shall, subject to applicable Law, be in the manner as provided in this Article 44.
- 44.2. In case of a Liquidity Event, the proceeds shall be distributed in the following order of priority:
  - (a) Firstly, to the Investor, until the Investor receives at least 2 (two) times of the Investment Amount, inclusive of any cash/accrual payments of interest or coupon already paid on INR basis;
  - (b) Secondly, to the Promoters and other shareholders (if any), until the Promoters and other shareholders receive, on a per Equity Security basis and for the total number of Equity Securities held, an amount equivalent to what the Investor has received under (a) above on a per Equity Security basis;
  - (c) Thirdly, amongst all Shareholders pro-rata to their shareholding in the Company.
- 44.3. In case of a Liquidation Event, the Investor shall be entitled to a liquidation preference for an amount which would be equal to the Investment Amount, inclusive of any cash/accrual payments of interest or coupon already paid, on an INR basis.

## XLVI - EVENT OF DEFUALT

#### 45. Event of Default

- 45.1. (i) Any failure or breach by the Company and/ or Promoters to comply with any of the Identified Terms, or (ii) any fraud by the Promoters with respect to the Company, the Subsidiaries or the Investor shall, if capable of remedy, be remedied by the Company and/ or Promoters within 60 (Sixty) days from the date of receipt of a written notice (the 'written notice') from the Investor requesting the Company and/ or the Promoters to remedy such failure or breach. In the event such failure or breach is not remedied by the Company and/ or the Promoters within such aforesaid stipulated time, after receipt of the 'written notice', such failure or breach shall be considered as an event of default ("Event of Default").
- 45.2. On or any time after the occurrence of an Event of Default, the Investor may (in its sole and absolute discretion) (i) serve a notice in writing (a "**Default Notice**") on the Company and the Promoters, informing the Company and the Promoters that an Event of Default has occurred; and (ii) shall be entitled to, subject to applicable Law, exercise the Drag Right in accordance with Article 43.4, either through the Default Notice or under a separate written notice addressed to the Company and the Promoters, provided however that in the event that the Drag Right is not exercised within a period of 6 (Six) months from the date of establishment of the Event of Default, this right of the Investor shall lapse without prejudice to the rights of the Investor in terms of this Article 45, upon the occurrence of any subsequent Event of Default.

45.3. Notwithstanding anything to the contrary as may be contained in the Articles, upon occurrence of Event of Default in accordance with Article 30.5, there shall not be any remedy period available to the Promoters as stated in Article 45.1 and the right of the Promoters under Article 36.4(d) shall not be available to the Promoters in case of exercise of Drag Right by the Investor in accordance with Article 43.4.

#### XLVII - FALL-OFF OF RIGHTS

46. Notwithstanding anything set out in these Articles, in the event the Investor's (calculated at all times, together with Affiliates) shareholding in the Company falls below 8% (eight per cent) on a Fully Diluted Basis, then all the rights available to the Investor (and its Affiliates) under the Agreement, except the information rights under Article 35 and Tag Along Right under Article 30.3, shall fall off and the Investor shall not have the right to exercise any other rights available to it under the Agreement. For the avoidance of doubt, the obligations of the Investor under the Agreement and the Articles shall survive even after the Investor's shareholding in the Company falls below 8% (eight per cent) on a Fully Diluted Basis. Further, subject to applicable law, the restrictions, undertakings and covenants under these Articles will be applicable unless otherwise consented to, or waived, as applicable, in writing, between the relevant parties.

### **XLVIII - BORROWING POWERS**

- 47. Subject to section 179 of the Companies Act, 2013 and Article 38, Board may from time to time at their discretion raise and borrow and may themselves lend and secure the payment of any sum or sums of money for the purpose of the Company.
- 48. The Board may raise or secure the repayment of such sum or sums in the manner and upon such terms and conditions in all respect as they deem fit and particular by creation of any mortgage or charge on the undertaking of the whole or any part of the property, or future, or uncalled capital of the Company or by the issue of bonds, redeemable debentures or denture-stock of the Company charged upon all or any part of the property of the Company both present and future including its uncalled capital for the time being.
- 49. All monies hundies, bills of exchange, notes belonging to the Company shall be deposited with one or more banks as the Board of Directors may decide, into an account to be opened in the name of the Company. Cheques on the banks shall be signed by one or more of the managing directors or working directors or by the members of the staff of the Company to whom the Board assign power of attorney or empower him/them by the Board resolution.
- 50. The Board of Directors are authorized to open accounts with in any bank in the name of the Company and operate upon those accounts and to invest the funds of the Company in Government or other securities of any kind whatsoever and to collect interest on the same from time to time and/or endorse, transfer or otherwise deal with securities of the company.
- 51. The Board of Director by proper resolutions shall have the power to do all or any of the acts mentioned in the Memorandum of Association subject to section 179 of the Act.

- 52. The Company shall comply with the provisions of the Act, or any statutory modification thereof for the time being in force and in particular in regard to the registration of particulars of mortgages and charges affecting the property of the Company or created by it, and to keeping a register of the Directors and to send to the Registrar of Companies an annual return, statutory returns and a summary of particulars relating thereto and notice of any increase of share capital, and copies of special resolutions.
- 53. Debentures, Debenture-Stock, bonds, or other securities may be made assignable free from any equities between the Company and the person to whom the same may be issued.
- 54. Subject to applicable Law and these Articles, any debenture, debenture stock, bonds, or other securities may be issued at discount, premium or otherwise and may be issued on condition that they shall be convertible into shares of any denomination and with any privileges and conditions as to redemption, surrender, drawing, allotment of shares, attending (but not voting) at the General Meeting, appointment of Directors and otherwise Debentures with the right to conversion into or allotment of shares shall be issued only with the consent of the company in the General Meeting by a Special Resolution.

## XLIX - DISQUALIFICATIONS OF DIRECTORS

- 55. The office of a director shall become vacant:
  - 55.1. If by notice in writing to the Company a Director resigns the office of director,
  - 55.2. If a Director ceases to be a director by virtue of the Act or becomes disqualified in accordance with provisions of Section 164 of the Companies Act, 2013,
  - 55.3. If a Director adjudged insolvent, found lunatic or of unsound mind or involved in such offence that tell upon him as committing an act moral turpitude in the eyes of law or does such act during discharge of his duties as director as tantamount to criminal breach of trust.

## L - ACCOUNTS AND AUDIT

- 56. The Board of Director shall cause true account to be kept of the sums of money received and expended by the Company, all sales and purchases of goods by the Company and the matters in respect of which such receipt and expenditure take place and of the assets, credit and liabilities of the Company.
- 57. The books of accounts shall be kept the registered office of the Company or at such other place or places as the Board of the directors of the Company in writing decide and shall be open to inspection by the directors during business hours.
- 58. The Company may keep such books of account or other relevant papers in electronic mode in the manner as prescribed under the Companies (Accounts) Rules, 2014, if the Board so decides.
- 59. The Board shall from time to time determine whether and to what extent and at what time and places and under what conditions or regulation the accounts, books and documents of the

Company and any of them shall be open to the inspection by the members not being the Directors and no member (not being a Director) shall have any right of inspecting any account or books or document of the Company except as conferred by statute or authorized by the Board or by the Company in general meeting.

- 60. Once at least in every year the accounts of the Company shall be examined and the correctness of the Balance Sheet and Profit and Loss Account ascertained by one or more statutory auditor or auditors.
- 61. The Company shall, at the annual general meeting, appoint an individual or a firm as an auditor who shall hold office for such period as determined by the Board and the manner and procedure of selection of auditors shall be as prescribed under Section 139 of the Companies Act, 2013 and other applicable provisions of the Act and the Companies (Audit and Auditors) Rules, 2014, provided that the Company shall place the matter relating to such appointment for ratification by members at every annual general meetings. Eligibility, qualifications, disqualification, powers, duties and removal of auditors shall be as prescribed under the Act and Companies (Audit and Auditors) Rules, 2014 as applicable to a public company.
- 62. The Company may at a general meeting remove any such auditor or all or any of such auditors and appoint in his or their places any other person or persons who have been nominated for appointment by any member of the Company and whose nomination has been given to the members of the Company not earlier than three months but at least fourteen days before the date of the meeting.

## LI - DIVIDENDS AND RESERVES

- 63. Subject to applicable Law, the Board may from time to time recommend and pay to the member such interim divided as will appear it to be justified by the profit of the Company and the final divided including interim dividend have to be approved by the general body meeting held thereafter, which shall be payable to such members whose names appear in the members register on the date of declaration at the general body meeting.
- 64. The general body can't declare dividend more than recommended by the Board.
- 65. The Board recommend any dividend, but before that set aside out of the profit of the Company such sum as it think proper as a reserves which shall at the discretion of the Board be applicable for any purpose to which the profit of the Company may be properly applied, including provision for meeting contingencies or for equalizing dividends and pending such application, may, at the discretion, either be employed in the business of the company or to be invested in such investments (other than shares of the Company) as the Board may from time to time, think fit.
- 66. The Board may also carry toward any profit which it may think prudent not to divide, with or without setting aside any profit as reserve, but such decision must be supported by cogent reasons recorded in the Boards resolution and must be approved by the members in a following general meeting.

- 67. Subject to the rights of the persons, if any, entitled to shares with special rights as to dividends, all dividends shall be declared and paid in cash according to and in proportion to the amounts paid or credited as paid in respect whereof the dividend is to be paid but if and so long as there be any arrears in any of the calls of the shares in the Company, dividends may be declared according to the amount of the shares already paid and the amounts of dividends may be adjusted against arrears of such calls on the shares.
- 68. Except as otherwise provided in these Articles and the SHA, all dividends shall be apportioned and paid proportionate to the amount paid or credited as paid on the shares as on the last day of the year but if any share is issued on terms providing that they shall rank for dividend as from a particular date such share shall rank for divided accordingly.
- 69. The Board may deduct from any divided payable to any member all sums of money, if any presently payable be him to the Company on account of calls or otherwise in relation to the share of the Company.
- 70. Every divided warrant may be sent by post to last registered address of the member entitled thereto and the receipt by the person whose name at the date of declaration of the dividend appears on the Register of Members as the owner of the share or in case of joint holder, the receipt by any one of such holders shall be a good discharge to the Company for all payments made in respect of such dividend.

# 71. Right to unpaid or unclaimed dividend

- (a) If the Company has declared a Dividend but which has not been paid or the Dividend warrant in respect thereof has not been posted or sent within 30 (thirty) days from the date of declaration, transfer the total amount of dividend, which remained unpaid or unclaimed within 7 (seven) days from the date of expiry of the said period of 30 (thirty) days to a special account to be opened by the Company in that behalf in any scheduled bank or private sector bank.
- (b) Any money so transferred to the unpaid Dividend account of the Company which remains unpaid or unclaimed for a period of 7 (seven) years from the date of such transfer, shall be transferred by the Company to the Fund established under subsection (1) of Section 125 of the Act, viz. "Investors Education and Protection Fund".
- (c) No unpaid or unclaimed Dividend shall be forfeited by the Board before the claim becomes barred by Law and such forfeiture, if effected, shall be annulled in appropriate cases.

#### LII - THE SEAL

- 72. The Directors shall provide for the safe custody of the seal.
- 73. The seal of the Company shall not be affixed to any instrument except by authority of a resolution of the Board or of a committee of the Board authorised by it in that behalf, and except in the presence of at least one Director and of the secretary or such other person as the Board may appoint for the purpose; and that one Director and the secretary or other person aforesaid shall sign every instrument to which the seal of the Company is so affixed in their presence.

#### LIII - INDEMNITY

- 74. The Company shall be bound by the indemnification provisions as provided under clause 19 of the Agreement.
- 75. The Directors, manager, auditor, secretary and other officers for the time being of the Company and any trustees for the time being acting in relation to any affairs of the Company and the heirs, executors and administrators respectively shall be indemnified for the losses which any of them shall or may incur or sustain by reason of any act done or committed in or about the execution of their duty in their respective offices of trust (except such ) (if any) as they shall incur or sustain by or through their own willful neglect or default respectively) and no such officer or trustee shall be answerable for the acts, receipts, neglects or defaults of any other officer or trustee or for joining in any receipt, for the sake of conformity or for the solvency or honesty of any bankers or other person with whom any money or effects belonging to the Company may be lodged or deposited for the safe custody or for any insufficiency or deficiency or any security upon which any money of the Company shall be invested or for any other loss or damages due to any such cause as aforesaid or which may happen in or about the execution of his office or trust unless the same shall happen through the willful neglect or default of such office or trustee. This Article is subject to section 197 of the Act.
- 76. Every officer or agent for the time being of the Company shall be indemnified out of the assets of the Company against any liability incurred by him in defending any proceedings whether civil or criminal, in which judgment is given favour or in which he is acquitted.

#### LIV - FAIR MARKET VALUE OF SHARES

- 77. The Fair Market Value of the Equity Securities for the purposes of these Articles and the Agreement shall be determined by a Valuer appointed based on mutual agreement of the Investor and the Promoters at the relevant time, it being understood that the Investor and Promoters shall use their best endeavours to ensure such determination within 30 (Thirty) days of appointment of the Valuer.
- 78. If the Investor and the Promoters are unable to agree on a Valuer, then within 30 (Thirty) days of such disagreement, the Investor and the Promoters shall each appoint a Valuer.
- 79. The Shareholders shall ensure that each Valuer has such information relating to the Company as it reasonably requires for determination of the Fair Market Value.
- 80. For arriving at the Fair Market Value, each Valuer shall take into account all factors it considers to be relevant.
- 81. If the difference in Fair Market Value determined by each of the Valuers is 15% (fifteen percent) or less, the average of the Fair Market Value determined by each of the Valuers shall be taken as the Fair Market Value.
- 82. If the difference in Fair Market Value determined by each of the Valuers is more than 15% (fifteen percent), the two Valuers shall jointly appoint a third Valuer ("**Third Valuer**"), who shall review the findings of the two Valuers and independently determine the Fair Market Value

it deems more accurate from amongst the two, taking into account the factors it considers to be relevant. Upon the Third Valuer's determination of the Fair Market Value, the Third Valuer shall set out the correct Fair Market Value in a certificate signed by him and issued to the Promoters and the Investor.

- 83. The Valuers and the Third Valuer (as the case may be) shall determine the Fair Market Value within 30 (thirty) days of their respective appointment and shall notify the Promoters and the Investor of their determination. Each party shall bear the fees of the respective Valuers appointed by them. The fees of the Third Valuer appointed under this Section XXXIV shall be paid by the Company.
- 84. Each of the Valuers and the Third Valuer shall act as an expert and not as an arbitrator.

### LV - SECRECY CLAUSE

- 85. Every Director, (except institutional/ex-officio Director) auditor, trustee, member of a committee, officer, servant, agent, accountant or other person employed in the business of the Company shall, if so required by the Directors before entering upon his duties, sign a declaration pledging himself to observe strict secrecy respecting all transactions and affairs of the Company and all matters relating thereto, and shall by such declaration pledge himself not to reveal any of the matters which may come to his knowledge in the discharge of his duties except when required to do so by the Directors or by law or by the person to whom such matters relate and except so far as may be necessary in order to comply with any of the provisions of these Articles.
- 86. No member shall be entitled to visit or inspect any works of the Company, without the permission of the Directors or to require discovery of or any information respecting any details of the Company's trading or any matter which is or may be in the nature of a trade secret, mystery of trade, secret process or any other matter which may relate to the conduct of the business of the Company and which, in opinion of the Director, it would be inexpedient in the interest of the Company to disclose.